FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Regulations made under ss. 58 and 81.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003

(LN. 2003/082)

7.8.2003

| Amending enactments | Relevant current provisions | Commencement date |
|---------------------|---------------------------------------|-------------------|
| LN. 2005/091 | rr. 4(1)(f), (g), 5(2)(k), (l), (m), | |
| | 8(1)(d), (e), (f), 9(1), 9A, 10A | 2.6.2005 |
| 2015/143 | rr. 2(1), (2), 4(1)(b), 5(2), Sch. 1, | |
| | 2 | 4.9.2015 |

Transposing:

Directive 90/394/EEC Directive 2004/37/EC Directive 2014/27/EU

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

ARRANGEMENT OF REGULATIONS

Regulation

- 1. Title.
- 2. Interpretation and scope
- 3. Assessment of health risks created by work involving carcinogens and mutagens.
- 4. Inspection of Assessment.
- 5. Prevention or control of exposure.
- 6. Foreseeable exposure.
- 7. Unforeseen exposure.
- 8. Information, instruction and training for persons who may be exposed to substances hazardous to health.
- 9. Health surveillance.
- 9A. Abnormality in employee.
- 10. Washing and changing facilities.
- 10A. Information for workers.
- 11. Revocation.

SCHEDULE 1

Part 1 - Carcinogens and mutagens

Part 2 - Annex I to Directive 67/548/EEC

Part 3 - Annex VI to Directive 67/548/EEC

Part 4 - Annex II to Directive 99/45/EEC

SCHEDULE 2

List of substances, mixtures and processes which also constitute carcinogens

SCHEDULE 3

Limit values for occupational exposure

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

In exercise of the powers conferred on him by sections 58 and 81 of the Factories Act, and all other enabling powers and for the purposes of transposing into the law of Gibraltar Council Directive 90/394/EEC of 28 June 1990 on the protection of workers from the risks related to exposure to carcinogens at work (Sixth individual Directive within the meaning of Article 16 (1) of Directive 89/391/EEC), Council Directive 97/42/EC of 27 June 1997 amending for the first time Directive 90/394/EEC and Council Directive 99/38/EC of 29 April 1999 amending for the second time Directive 90/394/EEC on the protection of workers from the risks related to exposure to carcinogens at work and extending it to mutagens, the Governor has made the following regulations—

Title.

1. These Regulations may be cited as the Factories (Control of Carcinogens and Mutagens at Work) Regulations 2003.

Interpretation and scope.

2.(1) In these Regulations, unless the context otherwise requires—

"carcinogen" means

- (a) a substance described in Schedule 1; and
- (b) a substance, mixture or process referred to in Schedule 2 as well as a substance or mixture released by a process referred to in Schedule 2;
- "inspector" means a person appointed under section 77 of the Factories Act to be an inspector;
- "limit value" means, unless otherwise specified, the limit of the time weighted average of the concentration for a carcinogen or mutagen in the air within the breathing zone of a worker in relation to a specified reference period as set out in Schedule 3;

"mutagen" means a substance described in Schedule 1.

(2) Where a provision in these Regulations is more favourable to health and safety than an equivalent provision in the Control of Asbestos Regulations 2007 then the provision contained in these Regulations shall take precedence.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Assessment of health risks created by work involving carcinogens or mutagens.

- 3.(1) An employer shall not carry out any work which is liable to expose any employees to any carcinogen or mutagen unless he has made a suitable and sufficient assessment of the risks created by that work to the health of those employees and of the steps that need to be taken to meet the requirements of these Regulations.
- (2) The assessment required by subregulation (1) shall be reviewed regularly and forthwith if—
 - (a) there is reason to suspect that the assessment is no longer valid; or
 - (b) there has been a significant change in the work to which the assessment relates, and, where as a result of the review, changes in the assessments are required, those changes shall be made.

Inspection of Assessment.

- 4.(1) An inspector may require an employer to disclose the information used in making an assessment under regulation 3 including—
 - (a) the activities or industrial process in which carcinogens and mutagens are used;
 - (b) the quantities of substances or mixtures manufactured or used which contain carcinogens or mutagens;
 - (c) the number of workers exposed;
 - (d) the preventive measures taken;
 - (e) the type of protective equipment used;
 - (f) the nature and degree of exposure,
 - (g) the cases of replacement.
- (2) It is an offence for an employer to fail to disclose information required by an inspector under subregulation (1) or to disclose information which he knows is false.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Prevention or control of exposure.

- 5.(1) Every employer shall ensure that—
 - (a) the exposure of his employees to carcinogens or mutagens is. either prevented, or,
 - (b) where it is not reasonably practicable to prevent the exposure of his employees to carcinogens or mutagens, the exposure is—
 - (i) adequately controlled; and
 - (ii) does not exceed the limit value.
- (2) Without prejudice to the generality of subregulation (1), where the assessment made under regulation 3 shows that it is not reasonably practicable to prevent exposure to a carcinogen or mutagen by using an alternative substance, mixture or process, the employer shall apply the following measures, namely—
 - (a) the total enclosure of the process of handling systems unless this is not reasonably practicable;
 - (b) the limitation of the quantities of a carcinogen or mutagen at the place of work;
 - (c) the keeping of the number of persons who might be exposed to a carcinogen or mutagen to a minimum;
 - (d) the use of plant, processes and systems of work which minimise the generation of, or suppress and contain, spills, leaks, dust, fumes and vapours of carcinogens or mutagens;
 - (e) the provision of hygiene measures including adequate washing facilities and regular cleaning of walls and surfaces;
 - (f) the designation of those areas and installations which may be contaminated by carcinogens or mutagens, and the use of suitable and sufficient warning signs;
 - (g) the prohibition of eating, drinking and smoking in areas that may be contaminated by carcinogens or mutagens;

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (h) the safe storage, handling and disposal of carcinogens and mutagens and use of closed and clearly labelled containers;
- (i) the means for safe storage, handling and transportation;
- (j) means for safe collection, storage and disposal of waste by workers;
- (k) the evacuation at source of carcinogens and mutagens is not effected in a manner which is or results in a threat to public health or the environment;
- (l) the drawing up of plans to deal with emergencies likely to result in abnormally high exposure; and
- (m) collective prevention measures and/or, where exposure cannot be avoided by other means, individual protection measures.

Foreseeable Exposure.

- 6.(1) Where an employer intends to undertake an activity whereby workers are likely to be exposed to abnormally high levels of carcinogens or mutagens the employer must, in consultation with the workers or their representatives, take steps necessary for reducing to a minimum the duration of the worker's exposure.
- (2) An employer undertaking an activity described in subregulation (1) must provide and require workers to wear protective clothing and respiratory protection equipment for the duration of the exposure.
- (3) The area in which an activity to which subregulation (1) applies is conducted must be clearly indicated or demarcated and access denied to unauthorised persons and unprotected workers.

Unforeseen exposure.

- 7.(1) In the event of the failure of a control measure which might result in the escape of carcinogens or mutagens into the workplace, the employer shall ensure that employees and other persons who may be affected are informed of the failure as quickly as possible.
- (2) Where subregulation (1) applies the following measures must be taken until such time as the cause of the abnormal exposure has been eliminated and the levels of carcinogens or mutagens restored to normal—

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- (a) only those workers who are essential to the carrying out of repairs and other necessary work are permitted to work in the affected area;
- (b) workers must be issued and must wear protective clothing and individual respiratory protection equipment;
- (c) unprotected workers must not be allowed to work in the affected area.

Information, instruction and training for persons who may be exposed to substances hazardous to health.

- 8.(1) An employer who undertakes work which may expose any of his employees to carcinogens and mutagens shall provide that employee with such information, instruction and training as is suitable and sufficient for him to know—
 - (a) the risks to health created by such exposure;
 - (b) the precautions which should be taken;
 - (c) hygiene requirements;
 - (d) the correct wearing and use of protective clothing and equipment;
 - (e) the labelling, symbols and hazard signs associated with containers, packages and installations containing carcinogens and mutagens; and
 - (f) the steps to be taken by workers (including rescue workers) in the case of incidents and to prevent incidents,

and such training shall be adapted where new risks are identified or the nature of the risk changes, and if necessary such training shall be repeated periodically.

(2) Without prejudice to the generality of subregulation (1), the information provided under that subregulation shall include information on the collective results of any health surveillance undertaken in accordance with regulation 9 in a form calculated to prevent it from being identified as relating to any particular person.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Health surveillance.

- 9.(1) Where an employee is or is liable to be exposed to carcinogens or mutagens, the employer shall, if appropriate, ensure that the employee undergoes health surveillance prior to exposure and at regular intervals thereafter.
- (2) The employer shall ensure that a health record, in respect of each of his employees to whom subregulation (1) relates is made and maintained and that that record or a copy thereof is kept in a suitable form for at least 40 years from the date of the last entry made in it.
- (3) Where an employer who holds records in accordance with subregulation (2) ceases to trade, he shall forthwith notify the Gibraltar Health Authority thereof in writing and offer those records to the Gibraltar Health Authority.
- (4) Where, for the purpose of carrying out his functions under these Regulations, an employment medical adviser or appointed doctor requires to inspect any workplace or any record kept for the purposes of these Regulations, the employer shall permit him to do so.
- (5) On reasonable notice being given, the employer shall allow any of his employees access to the health record which relates to him.

Abnormality in employee.

- 9A. Where as a result of the health surveillance carried out in accordance with regulation 9, an employee is found to be suffering from an abnormality that is suspected to be the result of exposure to carcinogens or mutagens—
 - (a) an inspector or a doctor may require other employees who have been similarly exposed to undergo health surveillance;
 - (b) a further assessment under regulation 3 shall be undertaken; and
 - (c) where cancer is identified as resulting from the occupational exposure to a carcinogen or mutagens, an inspector must be informed.

Washing and Changing Facilities.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- 10. Every employer must provide for any employee exposed to carcinogens or mutagens adequate and suitable—
 - (a) washing and changing facilities;
 - (b) where he is required to provide protective clothing, separate facilities for the storage of
 - (i) that protective clothing, and
 - (ii) personal clothing not worn during working hours;
 - (c) where he is required to provide respiratory protective equipment, facilities for the storage of that equipment.

Information for workers.

- 10A. An employee and/or his representative shall be entitled to request information from his employer so that he may verify that the provisions of these regulations are being complied with, and in particular—
 - (a) to be able to assess the consequences for his safety and health of the selection, wearing and use of protective clothing and equipment, without prejudice to the employer's responsibility for determining the effectiveness of protective clothing and equipment;
 - (b) the steps determined by the employer which are referred to in regulation 6(1), without prejudice to the employer's responsibility for determining such steps;
 - (c) in the event of an abnormal exposure employees and/or their representatives shall be informed as quickly as possible of the causes thereof and of the measures taken or to be taken to rectify the situation;
 - (d) an employer must keep an up-to-date list of employees engaged in the activities in respect of which the results of an assessment reveal a risk to their health or safety, indicating, if the information is available, the exposure to which they have been subjected;

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (e) an employer shall grant to the doctor and/or an inspector, as well as all other persons who have responsibility for health and safety at work, access to the list referred to in point (d);
- (f) an employee shall be granted access by his employer to the information on the list which relates to him personally;
- (g) an employee and/or his representative shall be granted access to anonymous collective information.

Revocation.

11. The Factories (Control of Carcinogens at Work) Regulations 1997 are revoked.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

SCHEDULE 1

regulation 2

PART 1

CARCINOGENS AND MUTAGENS

- 1. A carcinogen is-
 - (a) a substance or mixture which meets the criteria for classification as a category 1A or 1B carcinogen set out in Annex I to Regulation (EC) No 1272/2008 of the European Parliament and of the Council;
 - (b) a substance, mixture or process referred to in Schedule 2 as well as a substance or mixture released by a process referred to in Schedule 2.
- 2. A mutagen is—

a substance or mixture which meets the criteria for classification as a category 1A or 1B germ cell mutagen set out in Annex I to Regulation (EC) No 1272/2008 of the European Parliament and of the Council.

PART 2

ANNEX I TO DIRECTIVE 67/548/EEC

FOREWORD TO ANNEX I

Introduction.

Annex I is an index of dangerous substances for which harmonised classification and labelling have been agreed at Community level in accordance with the procedure laid down in Article 4(3) of this Directive.

Numbering of entries.

Entries in Annex I are listed according to the atomic number of the element most characteristic of the substance's properties. A list of the chemical elements, arranged according to atomic number is shown in Table A. Organic substances, because of their variety, have been placed in the usual classes, as shown in Table B.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

The Index number for each substance is in the form of a digit sequence of the type ABC-RST-VW-Y, where:

- ABC is either the atomic number of the most characteristic chemical element (preceded by one or two zeros to make up the sequence) or the usual class number for organic substances,
- RST is the consecutive number of the substance in the series ABC,
- VW denotes the form in which the substance is produced or placed on the market,
- Y is the check-digit calculated in accordance with the ISBN (International Standard Book Number) method.

As an example, the Index number for sodium chlorate is 017-005-00-9.

For dangerous substances in the European Inventory of Existing Commercial Chemical Substance (Einecs, OJ No. C 146A, 15.6.1990) the Einecs number is included. This number is a seven-digit system of the type XXX-XXX-X which starts at 200-001-8.

For dangerous substances notified under the provisions of this Directive, the number of the substance in the European List of Notified Substance (Elincs) is included. This number is a seven-digit system of the type XXX-XXX-X which starts at 400-010-9.

For dangerous substances in the list of "No-longer-polymers" (Document, Office for Official Publications of the European Communities, 1997. ISBN 92-827-8995-0) the "No-longer-polymer" number is included. This number is a seven-digit system of the type XXX-XXX-X which starts at 500-001-0.

The Chemical Abstracts Service (CAS) number is also included to assist identification of the entry. It should be noted that the Einecs number includes both anhydrous and hydrated forms of a substance, and there are frequently different CAS numbers for anhydrous and hydrated forms. The CAS number included is for the anhydrous form only, and therefore the CAS number shown does not always describe the entry as accurately as the Einecs number.

Einecs, Elincs, "No-longer-polymer" or CAS numbers are not usually included for entries which comprise more than four individual substances.

Nomenclature.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Wherever possible, dangerous substances are designated by their Einecs, Elincs or "No-longer-polymer" names. Other substances not listed in Einecs, Elincs or the list of "No-longer-polymers" are designated using an internationally recognised chemical name (e.g. ISO, IUPAC). An additional common name is included in some cases.

Impurities, additives and minor components are normally not mentioned unless they contribute significantly to the classification of the substance.

Some substances are described as a mixture of A and B. These entries refer to one specific mixture. In some cases where it is necessary to characterise the substance put on the market, the proportions of the main substances in the mixture are specified.

Some substances are described with a specific percentage purity. Substances containing a higher content of active material (e.g. an organic peroxide) are not included in the Annex I entry and may have other hazardous properties (e.g. explosive). Where specific concentration limits are shown, these apply to the substance or substances shown in the entry. In particular, in the case of entries which are mixtures of substances or substances described with a specific percentage purity, the limits apply to the substance as described in Annex I and not the pure substance.

Article 23(2)(a) requires that for substances appearing in Annex I, the name of the substance to be used on the label should be one of the designations given in the Annex. For certain substances, additional information has been added in square brackets in order to help identify the substance. This additional information need not be included on the label.

Certain entries contain a reference to impurities. An example is Index No. 607-190-00-X: methyl acrylamidomethoxyacetate (containing $\geq 0,1$ % acrylamide). In these cases the reference in brackets forms part of the name, and must be included on the label.

Certain entries refer to groups of substances. An example is Index No. 006-007-00-5 "hydrogen cyanide (salts of) with exception of_ complex cyanides such as ferrocyanides, ferricyanides and mercuric oxycyanide". For individual substances covered by these entries, the Einecs name or another internationally recognised name must be used.

Format of entries.

The following information is given for each substance in Annex I:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (a) the classification:
 - (i) the process of classification consists of placing a substance in one or more categories of danger (as defined in Article 2(2) of Directive 92/32/EEC) and assigning the qualifying risk phrase or phrases. The classification has consequences not only for labelling but also for other legislation and regulatory measures on dangerous substances;
 - (ii) the classification for each category of danger is normally presented in the form of an abbreviation representing the category of danger together with the appropriate risk phrase or phrases. However, in some cases (i.e. substances classified as flammable, sensitising and some substances classified as dangerous for the environment) the risk phrase alone is used;
 - (iii) the abbreviation for each of the categories of danger is shown below:
 - explosive: E
 - oxidising: O
 - extremely flammable: F+
 - highly Flammable: F
 - flammable: R10
 - very toxic: T+
 - toxic: T
 - harmful: Xn
 - corrosive: C
 - irritant: Xi
 - sensitising: R42 and/or R43
 - carcinogenic: Carc. Cat. (1)
 - mutagenic: Muta. Cat. (1)
 - toxic for reproduction: Repr. Cat⁽¹⁾
 - dangerous for the environment: N or/and R52, R53, R59;
 - (iv) Additional risk phrases which have been assigned to describe other properties (see sections 2.2.6 and 3.2.8 of

(1) The category of carcinogen, mutagen or toxic for reproduction (i.e. 1, 2 or 3) is shown as appropriate.

© Government of Gibraltar (www.gibraltarlaws.gov.gi)

_

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

the labelling guide) are shown although they are not formally part of the classification.

- (b) the label, including:
 - (i) the letter assigned to the substance in accordance with Annex II (see Article 23(2)(c)). This acts as an abbreviation for the symbol and for the indication of danger (if these are assigned);
 - (ii) the risk phrases, denoted as a series of numbers preceded by the letter R indicating the nature of the special risks, in accordance with Annex III (see Article 23(2)(d)). The numbers are separated by either:
 - a dash (-) to denote separate statements concerning special risks (R), or
 - an oblique stroke (/) to denote a combined statement, in a single sentence, of the special risks as set out in Annex III;
 - (iii) the safety phrases denoted as a series of numbers preceded by the letter S indicating the recommended safety precautions, in accordance with Annex IV (see Article 23(2)(e)). Again the numbers are separated by either a dash or an oblique stroke; the significance of recommended safety precautions are set out in Annex IV. The safety phrases shown apply only to substances; for preparations, phrases are selected according to the usual rules.

Note that for certain dangerous substances and preparations sold to the general public certain S-phrases are mandatory.

S1, S2 and S45 are obligatory for all very toxic, toxic and corrosive substances and preparations sold to the general public.

S2 and S46 are obligatory for all other dangerous substances and preparations sold to the general public other than those that have only been classified as dangerous for the environment.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Safety phrases S1 and S2 are shown in brackets in Annex I and can only be omitted from the label when the substance or preparation is sold for industrial use only.

(c) the concentration limits and associated classifications necessary to classify dangerous preparations containing the substance in accordance with Directive 1999/45/EC.

Unless otherwise shown, the concentration limits are a percentage by weight of the substance calculated with reference to the total weight of the preparation.

Where no concentration limits are given, the concentration limits to be used when applying the conventional method of assessing health hazards are those in Annex II, and when applying the conventional method of assessing environmental hazards are those in Annex III of Directive 1999/45/EC.

General Explanatory Notes.

Groups of substances

A number of group entries are included in Annex I. In these cases, the classification and labelling requirements will apply to all substances covered by the description if they are placed on the market, insofar as they are listed in Einecs or Elincs. Where a substance that is covered by a group entry occurs as an impurity in another substance, the classification and labelling requirements described in the group entry shall be taken into account in the labelling of the substance.

In some cases, there are classification and labelling requirements for specific substances that would be covered by the group entry. In such cases a specific Annex I entry will be present for the substance and the group entry will be annotated with the phrase "except those specified elsewhere in this Annex".

In some cases, individual substances may be covered by more than one group entry. Lead oxalate (Einecs No 212-413-5) is for instance covered by the entry for lead compounds (Index No 082-001-00-6) as well as for salts of oxalic acid (607-007-00-3). In these cases, the labelling of the substance reflects the labelling for each of the two group entries. In cases where different classifications for the same hazard are given, the classification leading to the more severe classification is used for the label of the particular substance (see section on Note A below).

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Entries in Annex I for salts (under any denomination) cover both anhydrous and hydrous forms unless specifically specified otherwise.

Substances with an Elincs number

In Annex I, substances with an Elincs number have been notified under the provisions of this Directive. A producer or importer who has not previously notified these substances must refer to the provisions of this Directive if he intends to place these substances on the market.

Explanation of the notes relating to the identification, classification and labelling of substances.

Note A:

The name of the substance must appear on the label in the form of one of the designations given in Annex I (see Article 23(2)(a)).

In Annex I, use is sometimes made of a general description such as "... compounds" or "... salts". In this case, the manufacturer or any other person who markets such a substance is required to state on the label the correct name, due account being taken of the chapter entitled "Nomenclature" of the Foreword:

Example: for BeCl2 (Einecs No 232-116-4): beryllium chloride.

The Directive also requires that the symbols, indications of danger, R- and S-phrases to be used for each substance shall be those shown in Annex I (Article 23(2)(c), (d) and (e)).

For substances belonging to one particular group of substances included in Annex I, the symbols, indications of danger, R - and S- phrases to be used for each substance shall be those shown in the appropriate entry in Annex I.

For substances belonging to more than one group of substances included in Annex I, the symbols, indications of danger, R- and S-phrases to be used for each substance shall be those shown in both the appropriate entries given in Annex I. In cases where two different classifications are given in the two entries for the same hazard, the classification reflecting the more severe hazard classification is used.

Example:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

for substance AB - no individual entry in Annex I:

Annex I group entry for compounds of A: Repr. Cat. 1; R61 Repr. Cat. 3; R62 Xn; R20/22 R33 N; R50-53

Annex I group entry for compounds of B: Carc. Cat.1; R45 T; R23/25 N; R51-53

Classification of substance AB thus becomes: Carc. Cat. 1; R45 Repr. Cat. 1; R61 Repr. Cat. 3; R62 T; R23/25 R33 N; R50-53

Note B

Some substances (acids, bases, etc.) are placed on the market in aqueous solutions at various concentrations and, therefore, these solutions require different labelling since the hazards vary at different concentrations.

In Annex I entries with Note B have a general designation of the following type: "nitric acid ...%".

In this case the manufacturer or any other person who markets such a substance in aqueous solution must state the percentage concentration of the solution on the label.

Example: nitric acid 45%.

Unless otherwise stated, it is assumed that the percentage concentration is calculated on a weight/weight basis.

The use of additional data (e.g. specific gravity, degrees Baumé) or descriptive phrases (e.g. fuming or glacial) is permissible.

Note C:

Some organic substances may be marketed either in a specific isomeric form or as a mixture of several isomers.

In Annex I, a general designation of the following type is sometimes used: "xylenol".

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

In this case the manufacturer or any other person who markets such a substance must state on the label whether the substance is a specific isomer (a) or a mixture of isomers (b).

Example:

- (a) 2,4-dimethylphenol
- (b) xylenol (mixture of isomers).

Note D:

Certain substances which are susceptible to spontaneous polymerisation or decomposition are generally placed on the market in a stabilised form. It is in this form that they are listed in Annex I to this Directive.

However, such substances are sometimes placed on the market in a non-stabilised form. In this case, the manufacturer or any person who places such a substance on the market must state on the label the name of the substance followed by the words "non-stabilised".

Example: methacrylic acid (non-stabilised).

Note E:

Substances with specific effects on human health (see Chapter 4 of Annex VI) that are classified as carcinogenic, mutagenic and/or toxic for reproduction in categories 1 or 2 are ascribed Note E if they are also classified as very toxic (T⁺), toxic (T) or harmful (Xn). For these substances, the risk phrases R20, R21, R22, R23, R24, R25, R26, R27, R28, R39, R68 (harmful), R48 and R65 and all combinations of these risk phrases shall be preceded by the word "Also".

Examples:

R45-23 "May cause cancer. Also toxic by inhalation"

R46-27/28 "May cause heritable genetic damage. Also very toxic in contact with skin and if swallowed".

Note F:

This substance may contain a stabiliser. If the stabiliser changes the dangerous properties of the substance, as indicated by the label in Annex I, a label should be provided in accordance with the rules for the labelling of dangerous preparations.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Note G:

This substance may be marketed in an explosive form in which case it must be evaluated using the appropriate test methods and a label should be provided reflecting its explosive property.

Note H:

The classification and label shown for this substance applies to the dangerous property(ies) indicated by the risk phrase(s) in combination with the category(ies) of danger shown. The requirements of Article 6 of this Directive on manufacturers, distributors and importers of this substance apply to all other aspects of classification and labelling. The final label shall follow the requirements of section 7 of Annex VI of this Directive.

This note applies to certain coal- and oil-derived substances and to certain entries for groups of substances in Annex I.

Note J:

The classification as a carcinogen need not apply if it can be shown that the substance contains less than 0.1 % w/w benzene (Einecs No 200-753-7). This note applies only to certain complex coal- and oil-derived substances in Annex I.

Note K:

The classification as a carcinogen need not apply if it can be shown that the substance contains less than 0.1 % w/w 1,3- butadiene (Einecs No 203-450-8). If the substance is not classified as a carcinogen, at least the S-phrases (2-)9-16 should apply. This note applies only to certain complex oil-derived substances in Annex I.

Note L:

The classification as a carcinogen need not apply if it can be shown that the substance contains less than 3 % DMSO extract as measured by IP 346. This note applies only to certain complex oil-derived substances in Annex I.

Note M:

The classification as a carcinogen need not apply if it can be shown that the substance contains less than 0.005 % w/w benzo[a]-pyrene (Einecs No 200-

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

028-5). This note applies only to certain complex coal-derived substances in Annex I.

Note N:

The classification as a carcinogen need not apply if the full refining history is known and it can be shown that the substance from which it is produced is not a carcinogen. This note applies only to certain complex oil-derived substances in Annex I.

Note P:

The classification as a carcinogen need not apply if it can be shown that the substance contains less than 0.1 % w/w benzene (Einecs No 200-753-7).

When the substance is classified as a carcinogen, Note E shall also apply. When the substance is not classified as a carcinogen at least the S-phrases (2-)23-24-62 shall apply.

This note applies only to certain complex oil-derived substances in Annex I.

Note Q:

The classification as a carcinogen need not apply if it can be shown that the substance fulfils one of the following conditions:

- a short term biopersistence test by inhalation has shown that the fibres longer than 20 μ m have a weighted half-life less than 10 days, or
- a short term biopersistence test by intratracheal instillation has shown that the fibres longer than 20 μm have a weighted half-life less than 40 days, or
- an appropriate intra-peritoneal test has shown no evidence of excess carcinogenicity, or
- absence of relevant pathogenicity or neoplastic changes in a suitable long term inhalation test.

Note R:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

The classification as a carcinogen need not apply to fibres with a length weighted geometric mean diameter less two standard geometric errors greater than $6\mu m$.

Note S:

This substance may not require a label according to Article 23 (see Section 8 of Annex VI).

Explanation of the notes relating to the labelling of preparations.

The significance of the notes that appear to the right of the concentration limits is as follows:

Note 1:

The concentration stated or, in the absence of such concentrations, the general concentrations of Directive 1999/45/EC are the percentages by weight of the metallic element calculated with reference to the total weight of the preparation.

Note 2:

The concentration of isocyanate stated is the percentage by weight of the free monomer calculated with reference to the total weight of the preparation.

Note 3:

The concentration stated is the percentage by weight of chromate ions dissolved in water calculated with reference to the total weight of the preparation.

Note 4:

Preparations containing these substances have to be classified as harmful with R65 if they meet the criteria in Section 3.2.3 in Annex VI.

Note 5:

The concentration limits for gaseous preparations are expressed as volume per volume percentage.

Note 6 :

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Preparations containing these substances have to be assigned R67 if they meet the criteria in Section 3.2.8 in Annex VI.

This note will no longer apply from the date on which the criteria for the use of R67 provided for in Directive 1999/45/EC enter into force.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

TABLA A - TABEL A - TABELLE A - ΠΝΑΚΑΣ A - TABLE A - TABLEAU A - TABELLA A - TABEL A - TABELLA A - TABELA A - TABELLA A - TABELA A - TABELA A - TABELA A - TABELA

Lista de los elementos químicos clasificados por su número atômico (Z)

Liste over grundstoffer, ordnet efter deres atomvægt (Z)

Liste der chemischen Elemente, geordnet nach der Ordnungszuhl (Z)

Κατάλογος χημικών στοιχείων ταξινομημένων σύμφωνα με τον ατομικά τους αριθμά (7)

List of chemical elements listed according to their atomic number (Z)

Liste des éléments chimiques classés selon leur numéro atomique (Z)

Elenco degli elementi chimici ordinuti secondo il loro numero atomico (Z)

Lijst van chemische elementen, gerangschikt naar atoomgewicht (Z)

Lista dos elementos químicos ordenados segundo o seu número atómico (Z)

Lista över grundämnen, ordnade efter deras atomnummer (Z) Alkunineiden luettelo, järjestysluvun mukaan (Z)

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| Z | Symb. | DE | DA | EI. | EN | ES |
|----|-------|-------------|---------------------|----------|------------|-----------|
| 1 | н | Wasserstoff | Hydrogen | Υδρογόνο | Hydrogen | Hidrógene |
| 2 | He | Helium | Helium | 10.60 | Helium | Helio |
| 3 | Li | Lethium | Lithium | λ%ο | Lithiyan | Litio |
| 4 | Be | Beryllium | Beryllium | Βηράλλιο | Beryllium | Berilio |
| 5 | В | Bor | Bor | Ворю | Horon | Boro |
| 6 | c | Kohlenstoff | Carbon (kulstof) | Άνθρακος | Carbon | Carbono |
| 2 | N | Stickstoff | Nitrogen | λίμοτο | Nitrogen | Nitrogeno |
| 8 | 0 | Sauerstoff | Oxygen (ilt) | Οξυγίνο | Охудеп | Oxigeno |
| 9 | F | Fluor | Fluor | Φθόριο | Fluorine | Pláor |
| 10 | No | Neon | Neon | Nãov | Neon | Neón |
| 11 | Na | Natrium | Natrium | Νάτριο | Sockum | Sodio |
| 12 | Mg | Magnesium | Magnesium | Маумута | Magnesium | Magnesic |
| 13 | Al | Aluminism | Aluminium | Αργθλιο | Aluminium | Aluminso |
| 14 | Si | Silicium | Silleium | Порітю | Silicon | Silicio |
| 15 | Р | Phosphor | Phosphor | Φωσφόρος | Phosphorus | Fósforo |
| 16 | s | Schwefel | SvavI | Octov | Sulphur | Azufre |
| 17 | CI | Chlor | Chlor | Χλώριο | Chlorine | Clore |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| Z | Symb. | FR | IT | NL. | PT | sv | п |
|----|-------|--------------------------|------------|------------|------------|-----------|-----------|
| 1 | Н | Hydrogène | idrogeno | Waterstof | Hidrogénio | Văte | Vety |
| 2 | He | Hélium | Elio | Helium | Hélio | Helium | Helium |
| 3 | i,i | Lithium | Litio | Lithium | Litio | Litium | Litium |
| 4 | Be | Béryllium (Glucinium) | Betillio | Beryllium | Berilio | Beryllium | Berylliam |
| 5 | В | Bore | Boro | Boor | Boro | Bor | Boori |
| ń | C | Carbone | Carbonio | Koolstof | Carbono | Kol | Hidi |
| 2 | N- | Azote | Azoto | Stikstof | Azoto | Kväve | Турре |
| 8 | 0 | Oxygène | Ossigeno | Zaretel | Oxigénio | Syre | Happi |
| 9 | F | Fluor | Fluero | Fluor | Flóor | Fluor | Floors |
| 10 | Ne | Néon | Neon | Neon | Néon | Neon | Neon |
| 11 | Na. | Sodivan | Sodio | Natrium | Sódio | Natrium | Natrium |
| 12 | Mg | Magnésium | Magnesio | Magnesium | Magnésio | Magnesium | Magnesiva |
| 13 | Al | Aluminium | Allveninio | Alteninium | Aluminio | Aluminium | Alumins |
| 14 | Si | Silicium | Silicio | Silicium | Silicio | Kisel | Pii |
| 15 | P | Phosphore | Fosforo | Fosfor | Fésfore | Fosfor | Fosfori |
| 16 | s | Soufre | Zolfa | Zwavel | Enxofre | Svevel | Rikki |
| 17 | CI | Chlore | Cloro | Chloor | Cloro | Klor | Kloon |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| Z | Symb. | DE | DA | EL | EN | ES |
|----|-------|-----------|-----------|-------------|-----------|-----------|
| 18 | Ar | Argon | Argon | Αργδ | Argon | Argón |
| 19 | К | Kalium | Kalium | Калю | Poussium | Potasio |
| 20 | Ca | Calcium | Calcium | Λσβέστιο | Calcium | Calcio |
| 21 | Sc | Scandium | Scandium | Σκάνδιο | Scandium | Escandio |
| 22 | n | Titaes | Titan | Τιτάνιο | Titanium | Tetanio |
| 23 | v | Vanadism | Variadium | Baváðia | Vanadium | Vanadio |
| 24 | Cr | Chrom | Chrom | Хровно | Chromium | Cromo |
| 25 | Mn | Mangan | Mangan | Μαγγάνιο | Manganese | Manganeso |
| 26 | Fe | Eisen | Jern | Σιδηρος | Iron. | Hierro |
| 27 | Co | Kobalt | Cobalt | Κήλώτιο | Cobult | Cobulto |
| 28 | NI | Nickel | Nikkel | Νικίλιο | Nickel | Niquel |
| 29 | Cii | Kupfer | Kobber | Χαλκής | Copper | Cobre |
| 30 | Zn | Zink | Zink | Ψευδάρχυρος | Zinc | Zinc |
| 31 | Gu | Gallism | Gallium | L95799 | Gallium | Galio |
| 12 | Ge | Germanium | Germanium | Γερμάνιο | Germanium | Germanio |
| 33 | As | Arsen | Arsen | Αρστυνικό | Arsenic | Arsènico |
| 34 | Se | Selen | Selen | Σελήνιο | Selenium | Selenio |
| 35 | Вс | Brom | Brom | Βρώμιο | Bromine | Bromo |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| Z. | Symb. | FR | II | NL | PT | sv | FI |
|----|-------|-----------|-----------|------------|----------|-----------|-----------|
| 18 | Ar | Argon | Argon | Argon | Árgon | Argon | Argon |
| 19 | K | Potassium | Potassio | Kalium | Potássio | Kaliun | Kalium |
| 20 | Ca | Calcium | Calcio | Calcium | Cálicio | Kaleium | Kalsium |
| 21 | Sc | Scandium | Scandio | Scandium | Escándio | Skandium | Skandium |
| 22 | n | Titane | Titanio | Titaan | Titlmio | Titan | Titaani |
| 23 | V | Vanadium | Vanadio | Vanadium | Vanadio | Vanadin | Vanadiini |
| 24 | Cr | Chrome | Cromo | Chroom | Crómio | Krom | Kromi |
| 25 | Mn | Manganèse | Manganese | Mangaan | Manganés | Margan | Mangauti |
| 26 | Fe | Fer | Ferro | ljzer | Ferro | Järn | Raina |
| 27 | Co | Cobalt | Cobalto | Kobult | Cobalto | Koboli | Kobolin |
| 28 | NI | Nickel | Nichel | Nikkel | Niquel | Nickel | Nikkeli |
| 29 | Cu | Cuivre | Asme | Koper | Cobre | Корраг | Kupari |
| 30 | Zn | Zinc | Zinco | Zink | Zinco | Zink | Sinkki |
| 11 | Ga | Gallium | Gallie | Gallium | Gálio | Gallium | Gallium |
| 32 | Ge | Germanium | Germanie | Germaniyan | Germânio | Germanium | Germaniun |
| 33 | As | Arsenio | Arsenico | Arsoen | Arsésio | Arsenik | Arseeus |
| 34 | Se | Sélénium | Selenio | Selenium | Selénio | Solen. | Seleens |
| 35 | Br | Brome | Bremo | Broom | Bromo | Brom | Bromi |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| Z | Symb, | DE | DA | EL. | EN | ES |
|-----|-------|------------|------------|-------------|------------|-----------|
| 36 | Kr | Krypton | Krypton | Критто | Кгурчов | Kryptón |
| 37 | ЖЪ | Rubidium | Rubidium | Ρουβήδιο | Rubidium | Rubidio |
| 38 | St | Strontium | Strontium | Στρόντιο | Strontium | Estroncio |
| 19 | Y | Yttrium | Yttrium | 'Үттрю | Yttrium | Itrio |
| 40 | Zz | Zirkon | Zirconium | Ζιρκόνιο | Zirconium | Circonio |
| 41 | Nb | Niob | Niobium | Νώβιο | Niobium | Niobio |
| 42 | Мо | Molybdán | Molybden | Mokufilétno | Molybdenum | Molibdene |
| 4.3 | Te | Technetium | Technesism | Τοχνήτω | Technetium | Tecnecia |
| 44 | Ru | Ruthenium | Ruthenium | Ρουθήνιο | Ruthenium | Rutenio |
| 45 | Rh | Rhodium | Rhodium | Póőio | Rhodium | Rodio |
| 46 | Pd | Palladium | Palladium | Παλλάδιο | Pelladium | Paladio |
| 47 | Ag | Silber | Solv | Χργυρος | Silver | Plata |
| 48 | Cd | Codmism | Cadmium | Κάδμιο | Cadmium | Cadmio |
| 49. | In | Indium | todium | ให้ข้อ | Indium | Indio |
| 50 | Sn | Zimn | Tin | Κωσσίτερος | Tin | Estaño |
| 51 | Sb | Antimon | Antimon | Αντιμόνιο | Antimony | Antimonic |
| 52 | Te | Tellur | Telur | Τελλοίριο | Tellurium | Telurio |
| 53 | 1 | Jod | Jod | Ιώδιο | Iodine | Yodo |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| Z | Symb. | FR | II | NL | PT | sv | FI |
|----|-------|------------|-----------|------------|------------|------------|-----------|
| 36 | Kr | Krypton | Krypton | Krypton | Kripton | Krypton | Krypton |
| 37 | Rb | Rubidium | Ruhidio | Rubidium | Rubidio | Rubidium | Rubidium |
| 38 | Sr | Strontium | Strenzio | Strontium | Estrôncio | Strontium | Strontium |
| 39 | Y | Ystrium | Ittrio | Ystrium | Itrio | Yttrium | Yttrium |
| 40 | Zr | Zirconium | Zirconio | Zirkonium | Zircônio | Zirkonium | Zirkenium |
| 41 | Nb | Niobium | Niobio | Niebium | Niébia | Niob | Niobium |
| 42 | Мо | Molyhdène | Molibdeno | Molybdeen | Molibdenio | Molybden | Molyhdeen |
| 43 | Te | Technetium | Tecnezio | Tochnetium | Tecnécio | Teknetium | Teknetium |
| 44 | Ru | Ruthdnium | Rutenio | Ruthernium | Ruténio | Rutenium | Rutenium |
| 45 | Rh | Rhodism | Rodin | Rodism | Rôdio | Rodium | Rodium |
| 46 | Pd | Palladium | Palladio | Palladium | Paládio | Palladisen | Palladium |
| 47 | Ag | Argent | Argento | Zilver | Pnita | Silver | Hopea |
| 48 | Cd | Cadmirum | Cadmio | Cadmium | Cádmio | Kadmium | Kadmisen |
| 49 | In | Indiven | Indio | Indium | Índio | Indium | Indium |
| 50 | Sn | Étain | Stagno | Tin | Estanbo | Tenn | Tina |
| 51 | Sb | Antimoine | Antimonio | Antimoon | Antimônio | Antimon | Antimoni |
| 52 | To | Tellure | Tellurio | Telluar | Telúrio | Tellur | Telluari |
| 53 | 1 | lode | Iodio | Jood | Indo | Jod | Jodi |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| Z | Symb. | DE | DA | EL. | EN | ES |
|----|-------|------------|------------|-------------|--------------|-------------|
| 34 | Xe | Xenon | Xenon | Eévo | Xenon | Xenón |
| 55 | Cs | Caesium | Caesium | Καίσιο | Caestum | Cesio |
| 56 | Ва | Barium | Barium | Βάριο | Bartum | Bario |
| 57 | La | Lenthan | Lanthan | Λανδάνιο | Lanthamum | Lantano |
| 58 | Ce | Cer | Cerium | Δημήτριο | Cerium | Cerio |
| 59 | Pr | Praseodym | Pruseodym | Πρωσοοδόμιο | Preseodymium | Praseodimio |
| 60 | Nd | Neodym | Neodym | Νεοδόμιο | Neodymium | Niodimio |
| 61 | Pm | Promothium | Promethium | Προμήθειο | Promethium | Prometio |
| 62 | Sen | Samarium | Samanum | Σαμάριο | Samarium | Samario |
| 63 | Eu | Europium | Europium | Eupómo | Europium | Europio |
| 64 | Gd | Gadolinium | Gadotinium | Γαδολίνιο | Gadolinium | Gadolinio |
| 65 | ть | Terbium | Terbium | Τέρβιο | Terbisan | Terbio |
| 66 | Dy | Dysprosium | Dysprosium | Δυσπρόσιο | Dysprosium | Disprosio |
| 67 | Ho | Holmium | Holmium | Одио | Holmium | Holmio |
| 68 | Er | Erbium | Erbium | Терво | Erbium | Erbio |
| 69 | Tm | Thuliam | Thulium | 906/10 | Thulium | Tulio |
| 70 | Yb | Ytterbium | Ysterbium | Υττέρβιο | Ytterbium | herbio |
| 71 | 1.u | Lutetium | Lutetium | Λουτήτιο | Lutetium | Lutecio |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| Z | Symb. | FR | rr | NL | PT | sv | FI |
|----|-------|------------|-------------|--------------|-------------|------------|------------|
| 54 | Xe | Xénon | Xenon | Xenon | Xènon | Xenon | Ksenon |
| 55 | Cs | Césium | Cesio | Cesium | Césio | Cesium | Cesium |
| 56 | Ва | Baryum | Bario | Barium | Bário | Bariun | Barium |
| 57 | La | Lanthane | Lantanio | Lanthaan | Lantânio | Lantan | Lantanni |
| 58 | Ce | Cérium | Cerio | Cerium | Cério | Cerium | Cérium |
| 59 | Pr | Praséodyme | Praseodinio | Praseodymium | Prasoodimio | Praseodym | Praseodyyn |
| 60 | Nd | Néodyme | Neodimio | Neodymium | Neodimio | Neodym | Neodyymi |
| 61 | Pm | Promethium | Promezio | Promethium | Promôcio | Prometium | Promenum |
| 62 | Sm | Samarium | Sunario | Samarium | Samano | Samarium | Samariam |
| 63 | Eu | Europium | Europie | Europium | Енгоріо | Europhan | fluropium |
| 64 | Gd | Gadelinium | Gadelinie | Gadolinium | Gadelinie | Gadelinium | Gadolinaun |
| 65 | Тъ | Techism | Terbio | Terbium | Térbio | Terbium | Terbium |
| 66 | Dy | Dysprosium | Dispresio | Dysprosium | Disprésso | Dysprosium | Dysprosiun |
| 67 | Но | Holmium | Olmio | Holmium | Hólmio | Holmium | Holmium |
| 68 | Er | Erbisano | Erbio | Erbium | Érbio | Erbium | Erbium |
| 69 | Tm | Thulium | Tulio | Thulium | Tillie | Tulium | Tuliam |
| 70 | Yb | Ytterbium | Itterbio | Ytterbium | Itérbie | Ytterbium | Ytterbium |
| 71 | Lu | Lutetrum | Lutezio | Luterium | Lutécio | Lutetium | Lutetium |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| z | Symb. | DE | DA | EL | EN | ES |
|-----|-------|-------------|----------|----------------------------|----------|----------|
| 72 | Hf | Hafnium | Haftsum | Хомо | Hafnium | Hafnio |
| 73 | Ta | Tantal | Tental | Τοντάλιο | Tantalum | Tántalo |
| 74 | w | Wolfram | Wolfram | Βολφράμιο (Τουγκοτένιο) | Tungsten | Volframi |
| 7.5 | Re | Rhenium | Rhesium | Pilyno | Rhetium | Renio |
| 76 | Os | Osmium | Osmium | Оеню | Osmium | Osmio |
| 77 | lr . | Iridium | Iridium | Ιρίδιο | Eridium | tridio |
| 78 | Pt | Platin | Platin | Λευκόχρυσος | Platimum | Platino |
| 79 | Au | Gold | Guld | Χρισός | Gold | Oro |
| 80 | Hg | Quecksilber | Kviksalv | Υδρά, γγυρος | Mercury | Mercurio |
| 81 | TI | Thalliven | Thalium | O6λλ.ιο | Thallium | Tabio |
| 82 | Pb | Blei | Bly | Μάλυβδος | Lead | Plomo |
| 83 | Bi | Wismuth | Bismuth | Вигрюбою | Bismith | Bismuto |
| 84 | Po | Polonium | Plotium | Flokémo | Polouium | Polonia |
| 85 | At | Astat | Assa | Αστάτιο | Astatine | Astato |
| 86 | Rn | Radon | Radon | Ρυδόνιο | Radon | Radón |
| 87 | Fr | Francium | Francium | Φράγκιο | Francium | Francio |
| 88 | Ra | Radium | Radium | Ράδιο | Radium | Radio |
| 89 | Ac | Actinium | Actinium | Ακτίνιο | Actinium | Actinio |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| Z | Symb. | FR | rr | NL | PT | sv | FI |
|----|-------|-----------|-----------|-----------|------------|-------------|-----------|
| 72 | Hf | Heftsum | Afnio | Haftiium | Háftio | Hafnisen | Hafnisar |
| 73 | Ta | Tantale | Tantalio | Tentaal | Tântalo | Tantal | Terreal |
| 74 | W | Tungstèce | Tungsteno | Wolfrum | Tungstėnio | Wolfram | Volfram |
| 25 | Re | Rhénèum | Renio | Renium | Rénio | Rhenium | Remour |
| 76 | Os | Osmicen | Osmio | Osmium | Osmio | Osmiczn | Osmiczn |
| 77 | tr | Iridium | Iridio | tridium | tridio | Iridium | Iridium |
| 78 | PL | Platine | Platino | Platinum | Platina | Platina | Plating |
| 79 | Au | Or | Ore | Goud | Outo | Guld | Kulta |
| 80 | Hg | Mercure | Mercurio | Kwik | Meccúrio | Kvicksilver | Elohope |
| RI | TI | Thallium | Talkio | Thallisen | Tálio | Tallium | Tillium |
| 82 | Рb | Plomb | Piembo | 1.ood | Chumbo | Bly | Lyijy |
| 83 | Bi | Bismuth | Bismuto | Bismath. | Histricto | Vismut | Visioniti |
| 84 | Po | Poloneum | Polonio | Polonium | Polônio | Polonium | Polonius |
| 85 | Au | Asset | Asimo | Astast | Astato | Astat | Assatin |
| 86 | Rn | Radon | Radon | Radon | Rádon | Radon | Radon |
| 87 | Ft | Francium | Francio | Francium | Frâncio | Francium | Frankiun |
| 88 | Ra | Radium | Radio | Radium | Rádio | Radium | Radium |
| 89 | Ac | Actinium | Attinio | Actinium | Actinio | Aktinium | Aktinium |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| Z | Symb. | DE | DA | EL. | EN | ES |
|-----|-------|--------------|--------------|---------------------------|--------------|-------------|
| 90 | Th | Thorium | Thorium | Өдри | Thorium | Torio |
| 91 | Pa | Protactinium | Protactinium | Πρωτακτίνιο | Protectinium | Protectinio |
| 92 | U | Uran | Uran | Ospánio | Uniciam | Uranio |
| 93 | Np | Neptunium | Neptunium | Νεπτούνιο (Ποσειδώνιο) | Neptunium | Neptunio |
| 94 | Pu | Plutonium | Plutonium | (II) outdwio | Plutonium | Plutonio |
| 95 | Am | Americism | Americium | Αμερίκιο | Americism | Americio |
| 96 | Cm | Curium | Curium | Киогра | Curium | Curio |
| 97 | Hk | Berkelium | Berkelium | Мпераво | Berkelium | Berkelio |
| 98 | cr | Californium | Californium | Καλιφόρνιο | Californium | Californio |
| 90 | Es | Einsteinkum | Einsteinium | ΑΥ?στάνιον | Einsteinium | Einstenio |
| 100 | Fm | Fennium | Fermium | Φέρμιο | Fermium | Fermio |
| 101 | Md | Mendelevium | Mendelevium | Μεντελέβιο | Mendelevium | Mendelevi |
| 102 | No | Nobelium | Nobelium | Nojmikao | Nobelium | Nobelio |
| 103 | Lw | Lawrentism | Lawrentium | Λαυρένσιο | Lawrencium | Laurencio |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| Z | Symb. | FR | IT | NE | PT | sv | Fí |
|-----|-------|--------------|--------------|--------------|-------------|----------------|--------------|
| 90 | Th | Thorium | Torio | Thorium | Tório | Torium | Torium |
| 91 | Pa | Protectinium | Protoettinio | Protectinium | Protactinio | Protaksinisani | Protaktinium |
| 92 | U | Uranium | Uranio | Uranium | Urânio | Uran | Ursani |
| 91 | Np | Neptunium | Nettunio | Neptunium | Neptúnio | Neptuniam | Neptunian |
| 94 | Pu | Plutonium | Plutonio | Plutonium | Plutónio | Plutenium | Plutonium |
| 9.5 | Am | Américium | Americio | Americium | Americio | Americium | Amerikium |
| 96 | Cm | Curium | Curio | Curtism | Cúrio | Curium | Carium |
| 97 | Bk | Berkétium | Herkelio | Berkelium | Berquello | Berkeltum | Berkelium |
| 98. | cı | Californium | Californio | Californium | Califòrnio | Californium | Kalifornium |
| 99 | Es | Einsteinium | Einstenio | Einsteinium | Einsteinlo | Einsteinium | Finsteinium |
| 100 | Fm | Permium | Fermio | Fermium | Férmio | Fermium | Fermium |
| 101 | Md | Mendélévium | Mendelevio | Mendelevium | Mendelévio | Mendelevium | Mendelevium |
| 102 | No | Nobělium | Nobelio | Nobelium | Nobélio | Nobelium | Nobelium |
| 103 | Lw | Lawrentism | Lawrencio | Laurentium | Laurêncio | Lawrentium | Lawrensium |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

TABLA B – TABELL B – TABELLE B – ΠΙΝΑΚΑΣ Β – TABLE B – TABLEAU B – TABELLA B – TABEL B – TABELL B – TABEL B – TA

Clasificación especial para las sustancias orgánicas
Sarlig inddeling af organiske stoffer
Spezielle Anordnung für die organischen Stoffe
Fiduch ταξινόμηση του οργανικών αποιών
Special classification for organic substances
Classification particulière aux substances organiques
Classificazione speciale per le sostanze organiche
Speciale indeling voor de organische stoffen
Classificação especial para as substancias orgânicas
Sărskild indelning av organiska ämmen
Erityisryhmät organisille aineille

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

601 Hidrocurburos
Carbonhydrider (kulbrinter)
Kohlenwasserstoffe
Yöpernwistfpaurg
Hydrocurbons
Hydrocurbons
Hydrocurburi
Koolwaterstoffen
Hidrocurbonetos
Kolväten

Hillivedyt

Hidrocarburos halogenados
Halogensubstinuerede carbonhydrider
Halogen-Kohlenwasserstoffe
Alerponostupfrynym nöperponostipfakun
Halogenated hydrocarbons
Dérivés halogenés des hydrocarbures
Derivati idrocarburi alogenati
Gehalogeneerde koolwaterstoffen
Hidrocarbonetos halogenados
Halogenerade kolväten

Alcoholes y derivados
Alkoholes og deres derivater
Alkohole und ihre Derivate
Αλκούλες και παρέγεργή τους
Alcohols and their derivatives
Alcohis et dérivés
Alcohis et derivati
Alcoholen en derivaten
Álcoholes et derivados

Alkoholer och deras derivat

Alkoholit ja niiden johdannaiset

Halogenoidat hiilivedyt

O4 Femoles y derivados
Phenoler og deres derivater
Phenole und ihre Derivate
Φαινόλες και παράγουγά τους
Phenols and their derivatives
Phénols et dérivati
Fenoler en derivater

Fenôis e derivados Fenoler och deras derivat Fenolit ja niiden johdannaiset

Aldehídos y derivados

Aldehyder og deres derivater Aldehyde und ihre Derivate Aldehydes and their derivatives Aldehydes and their derivatives Aldehydes et derivati Aldehyden en derivaten Aldehyden en derivaten Aldehyder och deras derivat Aldehyder och deras derivat Aldehyder in miden johdannaiset 6 Cetonas y derivados
Ketoner og deres derivater
Ketone und ihre Derivate
Kritôvic kun πορόγρηγά τους
Ketones and their derivatives
Cétones et dérivés
Chetoni e derivate
Ketonen en derivaten
Cetonas e derivaten
Cetonas e derivaten
Ketoner och deras derivat
Ketoner ja niiden johdannaiset

Acidos orgânicos y derivados
Organiske syrer og deres derivater
Organische Säuren und ähre Derivate
Organic ocida son mopérpurph tous,
Organic ocids and their derivatives
Acides organiques et dérivate
Acidos organiques et dérivatel
Organische zuren en derivaten
Acidos orgânicos e derivados
Organiska syrer och deras derivat
Organische zuren en derivaten
Acidos orgânicos e derivados
Organiska syrer och deras derivat
Organisch hapot ja miden johdannaiset

608 Nitrilos Nitriler Nitriles Nitriles Nitriles Nitrilis Nitrillen Nitrillos Nitrilos Nitrilos Nitrilos

609 Derivados nitrados Nitroforbindelser Nitroverbindungen Netposevicese, Nitro compounds Derivés nitrés Nitroderivati Nitroverbindingen Derivados nitrados Kväveföreningar Nitroyludisteet

Operivados eleconitrados
Chlornitroverbindelser
Chlornitroverbindungen
Xanpuvrtpoewioung
Chloronitro compounds
Dérivés chloronitrés
Cloronitro derivati
Chloomitroverbindingen
Derivados eleconitrados
Klornitroffenningar
Kloorinitroyhdisteet

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

611 Derivados azotos y azoxi Azoxy- og azoforbindelser Azoxy- ond Azoverbindungen Agogu- szu fajoe-twinne; Azoxy- and azo compounds Dérivén azoxy et azofques Azoxy- en azoverbindingen Derivados azoxi e azóscos Azoxi- och azoforeningar Atsoksi- ja atsoyhdisteet

612 Derivados aminados Aminer Aminoverbindungen Aproxevieres, Amine compounds Derives amines Aminoderivati Aminoverbindingen Derivados aminados Amaner Aminis/hdistect

6)3 Bases heterociclicas y derivados
Heterocyclische baser og deres derivater
Heterocyclische Basen und ibre Derivate
Expanosokuse phones, som zupdrymyte vinc,
Heterocyclic bases and their derivatives
Bases heterocycliques et dérivés
Basi eterocicliche e derivati
Heterocyclische basen en hun derivaten
Bases heterociclicas e derivados
Heterocyclische baser och deras derivat
Heterocyclische baser och deras derivat
Heterocyclische baser in hiden johdannatiset

614 Glucósidos y alcaloides Glycosider og alkaloider Glycoside und Alkaloide Flyssos/freg km (ikknizosión) Glycosides and alkaloide Glucosides et alcaloides Glucosides et alcaloides Glycosides en alkaloiden Glycosides en alkaloides Glykosider och alkaloides Glykosider och alkaloider Glykosider och alkaloider

Cianatos e isocianatos
Cyanater og isocyanater
Cyanate und Isocyanater
Kvarvide kin stokvarvide evidence
Cyanates and isocyanates
Cyanates and isocyanates
Cianati e isocianati
Cyanaten en isocyanaten
Cianato e isocianati
Cyanaten ech isocyanater
Cyanater och isocyanater
Syanatit ja isocyanatet

616 Amidas y derivados
Amide og deres derivater
Amide und ihre Derivate
Apiden sen supérpeyé tens;
Amides and their derivatives
Amides et dérives
Amides et dérives
Amides et dérivati
Amiden en derivaten
Amiden en derivaten
Amider och deras derivat
Amidir ja miden johdannaiset

617 Peróxidos orgânicos Organische peroxider Organische Peroxide Opyrovich untspoljcióm Organie peroxides Peroxydes organiques Peroxidi organici Organische peroxiden Peróxidos orgânicos Organisch peroxider Organisch peroxider Organisch peroxidir

547 Enzimes
Enzymer
Enzyme
Enzymes
Enzymes
Enzymes
Enzimis
Enzymen
Enzimes
Enzymen
Enzimes

648 Sustancias complejas derivadas del carbón Komplekse kulderivater

Aus Kohle abgeleiste komplexe Stoffe Enjatkasor, ouoire zuparpiturus; carb divilonou Complex substances derived from coal Substances complexes dérived du charbon Sustance complexes dérivate du charbone Complexe stenkoolderivaten Substâncias complexas derivadas do carvão Komplexa kolderivat Montmarkaiset hiilijohdannaiset

649 Sustancias complejas derivadas del petróleo Komplekse obioderivator Aus Erdől abgeleiste komplexe Stoffe Especious, emoise, roperobjevos, omó not pérsuo Complex substances derived from petroleum Substances complexes dérivées du pétrole Sostanze complexes derivate dal petrolio Complexe aardoliederivaten Substâncias complexas derivadas do petróleo Komplexa objederivat Monimutkaiset óljýjohdannaiset

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

50 Suntancias diversas
Diverse stoffer
Verschiedene Stoffe
Aniquops; motis;
Miscellaneous substances
Substances diverses
Sostanze diverse
Diversett
Substancias diversas
Diverse ämmen

Muut nineet

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

PART 3 ANNEX VI TO DIRECTIVE 67/548/EEC

ANNEX VI

GENERAL CLASSIFICATION AND LABELLING REQUIREMENTS FOR DANGEROUS SUBSTANCES AND PREPARATIONS

CONTENTS

- GENERAL INTRODUCTION
- 2. CLASSIFICATION ON THE BASIS OF PHYSICOCHEMICAL PROPERTIES
- 2.1. Introduction
- 2.2. Criteria for classification, choice of symbols, indication of danger, choice of risk phrases
- 2.2.1. Explosive
- 2.2.2. Oxidising
- 2.2.3. Extremely flammable
- 2.2.4. Highly flammable
- 2.2.5. Flammable
- 2.2.6. Other physicochemical properties
- 3. CLASSIFICATION ON THE BASIS OF TOXICOLOGICAL PROPERTIES
- 3.1. Introduction
- 3.2. Criteria for classification, choice of symbols, indication of danger, choice of risk phrases
- 3.2.1. Very toxic
- 3.2.2. Toxic
- 3.2.3. Harmful
- 3.2.4. Comments regarding the use of R48
- 3.2.5. Corrosive
- 3.2.6. Irritant
- 3.2.7. Sensitisation
- 3.2.8. Other toxicological properties
- 4. CLASSIFICATION ON THE BASIS OF SPECIFIC EFFECTS ON HUMAN HEALTH

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- 4.1. Introduction
- 4.2. Criteria for classification, indication of danger, choice of risk phrases
- 4.2.1. Carcinogenic substances
- 4.2.2. Mutagenic substances
- 4.2.3. Substances toxic to reproduction
- 4.2.4. Procedure for the classification of preparations concerning specific effects on health
- 5. CLASSIFICATION ON THE BASIS OF ENVIRONMENTAL EFFECTS
- 5.1. Introduction
- 5.2. Criteria for classification, indication of danger, choice of risk phrases
- 5.2.1. Aquatic environment
- 5.2.2. Non-aquatic environment
- 6. CHOICE OF SAFETY ADVICE PHRASES
- 6.1. Introduction
- 6.2. Safety phrases for substances and preparations
- 7. LABELLING
- 8. SPECIAL CASES: Substances
- 8.1. Mobile gas cylinders
- 8.2. Gas containers intended for propane, butane or liquefied petroleum gas (LPG)
- 8.3. Metals in massive form
- 8.4. Substances classified with R65
- 9. SPECIAL CASES: Preparations
- 9.1. Gaseous preparations (gas mixtures)
- 9.2. Gas containers intended for preparations containing stenched propane, butane or liquefied petroleum gas (LPG)
- 9.3. Alloys, preparations containing polymers, preparations containing elastomers
- 9.4. Preparations classified with R65
- 9.5. Organic peroxides
- 9.6. Additional labelling requirements for certain preparations

COMMISSION STATEMENT

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

1. GENERAL INTRODUCTION

- 1.1. The object of classification is to identify all the physicochemical, toxicological and ecotoxicological properties of substances and preparations which may constitute a risk during normal handling or use. Having identified any hazardous properties the substance or preparation must then be labelled to indicate the hazard(s) in order to protect the user, the general public and the environment.
- 1.2. This Annex sets out the general principles governing the classification and labelling of substances and preparations referred to in Article 4 of this Directive and in Article 4 of Directive 1999/45/EC and other relevant Directives on dangerous preparations.

It is addressed to all those concerned (manufacturers, importers, national authorities) with methods of classifying and labelling dangerous substances and preparations.

1.3. The requirements of this Directive and of Directive 1999/45/EC are intended to provide a primary means by which the general public and persons at work are given essential information about dangerous substances and preparations. The label draws the attention of persons handling or using substances and preparations to the inherent danger of certain such materials.

The label may also serve to draw attention to more comprehensive product information on safety and use available in other forms.

1.4. The label takes account of all potential hazards which are likely to be faced in the normal handling and use of dangerous substances and preparations when in the form in which they are placed on the market, but not necessarily in any different form in which they may finally be used, e.g. diluted. The most severe hazards are highlighted by symbols, such hazards and those arising from other dangerous properties are specified in standard risk phrases, and safety phrases give advice on necessary precautions.

In the case of substances, the information is completed by the name of the substance under an internationally recognised chemical nomenclature, the preferred name being the one used in the European Inventory of Existing Commercial Chemical Substances (Einecs), or in the European List of Notified Chemical Substances (Elincs), the EC number and the name, address and telephone number of the person established in the Community who is responsible for placing the substance on the market.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

In the case of preparations, the information in accordance with Article 10.2. of Directive 1999/45/EC, is completed by:

- the trade name or the designation of the preparation;
- the chemical name of the substance or substances present in the preparation; and
- the name, full address and telephone number of the person established in the Community who is responsible for placing the preparation on the market.
- 1.5. Article 6 requires that manufacturers, distributors and importers of dangerous substances which appear in the Einecs but which have not yet been introduced into Annex I shall be obliged to carry out an investigation to make themselves aware of the relevant and accessible data which exist concerning the properties of such substances. On the basis of this information, they shall package and provisionally label these substances according to the rules laid down in Articles 22 to 25 and the criteria in this Annex.

1.6. Data required for classification and labelling

- 1.6.1. For substances the data required for classification and labelling may be obtained:
 - (a) as regards substances for which the information specified in Annex VII is required, most of the necessary data for classification and labelling appear in the 'base set'. This classification and labelling must be reviewed, if necessary, when further information is available (Annex VIII);
 - (b) as regards other substances (e.g. those referred to in section 1.5 above), the data required for classification and labelling may, if necessary, be obtained from a number of different sources, for example:
 - the results of previous tests;
 - information required by international rules on the transport of dangerous substances;
 - information taken from reference works and the literature; or

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- information derived from practical experience.

The results of validated structure-activity relationships and expert judgement may also be taken into account where appropriate.

- 1.6.2. For preparations, normally the data required for classification and labelling may be obtained:
 - (a) if it concerns physicochemical data, by the application of the methods specified in Annex V. This applies also to preparations covered by Directive 91/414/EEC unless other internationally recognised methods are acceptable in accordance with the provisions of Annexes II and III to Directive 91/414/EEC (Article 5. 5. Of Directive 1999/45/EC). For gaseous preparations a calculation method may be used for flammable and oxidising properties (see 9.1.1.1 and 9.1.1.2). For non-gaseous preparations containing organic peroxides a calculation method may be used for oxidising properties (see 2.2.2.1).
 - (b) if it concerns data on health effects:
 - by the application of the methods specified in Annex V, unless, in the case of plant protection products, other internationally recognised methods are acceptable in accordance with the provisions of Annexes II and III to Directive 91/414/EEC (Article 6. 1. (b) of Directive 1999/45/EC),
 - and/or by the application of a conventional method referred to in Article 6 and Annex II, Parts A. 1. 6. and B. 1. 5. of Directive 1999/45/EC, or,
 - in the case of R65, by the application of the rules under 3.2.3
 - however, if it concerns the evaluation of the carcinogenic, mutagenic and reproductive toxicity properties, by the application of a conventional method referred to in Article 6 and Annex II, Parts A. 7. 9. and B. 6. of Directive 1999/45/EC.
 - (c) if it concerns data on ecotoxicological properties
 - (i) for aquatic toxicity only:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- by the application of the methods specified in Annex V, subject to the conditions referred to in Annex III Part C of Directive 1999/45/EC, unless, in the case of plant protection products, other internationally recognised methods are acceptable in accordance with the provisions of Annexes II and III to Directive 91/414/EEC (Article 7. 1. (b) of Directive 1999/45/EC), or
- by application of a conventional method referred to in Article 7 and Annex III, Parts A and B, of Directive 1999/45/EC.
- (ii) for the evaluation of the potential for (or actual) bioaccumulation through the determination of log Pow (or BCF), or the evaluation of degradability, by application of a conventional method referred to in Article 7 and Annex III, Parts A and B, of Directive 1999/45/EC.
- (iii) for dangers of the ozone layer by application of a conventional method referred to in Article 7 and Annex III, Parts A and B, of Directive 1999/45/EC.

Note concerning the performance of animal tests

The performance of animal tests to establish experimental data is subject to the provisions of Directive 86/609/EEC regarding the protection of animals used for experimental purposes.

Note concerning physicochemical properties

For organic peroxides and organic peroxide preparations data may be derived from the calculation method set out in Chapter 9.5. For gaseous preparations a calculation method may be used for flammable and oxidising properties (see chapter 9).

1.7. Application of the guide criteria

Classification must cover the physicochemical, toxicological and ecotoxicological properties of substances and preparations.

Classification of substances and preparations is made according to Chapter 1.6, on the basis of the criteria in Chapters 2 to 5 (substances) and Chapters

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

2, 3, 4.2.4 and 5 of this Annex. All types of hazard must be considered. For instance, classification under 3.2.1 does not imply that the sections such as 3.2.2 or 3.2.4 can be ignored.

The choice of symbol(s) and risk phrase(s) is made on the basis of the classification in order to ensure that the specific nature of the potential dangers identified in classification is expressed on the label.

Notwithstanding the criteria given under 2.2.3, 2.2.4 and 2.2.5, substances and preparations in the form of aerosols shall be subject to the provisions of Directive 75/324/EEC as amended and adapted to technical progress.

1.7.1. Definitions

'Substances' means chemical elements and their compounds in the natural state or obtained by any production process, including any additive necessary to preserve the stability of the product, and any impurity deriving from the production process used, but excluding any solvent which may be separated without affecting the stability of the substance or changing its composition.

A substance may be chemically very well defined (e.g. acetone) or a complex mixture of constituents of variable composition (e.g. aromatic distillates). For certain complex substances, some individual constituents have been identified.

'Preparations' means mixtures or solutions composed of two or more substances.

1.7.2. Application of the guide criteria for substances

The guidance criteria set out in this Annex are directly applicable when the data in question have been obtained from test methods comparable with those described in Annex V. In other cases, the available data must be evaluated by comparing the test methods employed with those indicated in Annex V and the rules specified in this Annex for determining the appropriate classification and labelling.

In some cases there may be doubt over the application of the relevant criteria, especially where these require the use of expert judgement. In such cases the manufacturer, distributor or importer should provisionally classify and label the substance on the basis of an assessment of the evidence by a competent person.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Without prejudice to Article 6, where the above procedure has been followed and there is concern over possible inconsistencies then a proposal may be submitted for the entry of the provisional classification into Annex I. The proposal should be made to one of the Member States and should be accompanied by appropriate scientific data (see also section 4.1).

A similar procedure may be followed when information is identified which gives cause for concern over the accuracy of an existing entry in Annex I.

1.7.2.1. Classification of substances containing impurities, additives or individual constituents

Where impurities, additives or individual constituents of substances have been identified, they shall be taken into account if their concentration is greater than or equal to the limits specified

- 0,1 % for substances classified as very toxic, toxic, carcinogenic (category 1 or 2), mutagenic (category 1 or 2), toxic to reproduction (category 1 or 2), or dangerous for the environment (assigned the symbol 'N' for the aquatic environment, dangerous for the ozone layer)
- 1% for substances classified as harmful, corrosive, irritant sensitising, carcinogenic (category 3), mutagenic (category 3), toxic to reproduction (category 3), or dangerous for the environment (not assigned the symbol 'N', i.e. harmful to aquatic organisms, may cause long-term adverse effects)

unless lower values have been specified in Annex I.

With the exception of substances listed specifically in Annex I, classification should be carried out according to the requirements of Articles 5, 6 and 7 of Council Directive 1999/45/EC.

In the case of asbestos (650-013-00-6) this general rule does not apply until a concentration limit has been fixed in Annex I. Substances in which asbestos is present must be classified and labelled according to the principles in Article 6 of this Directive.

1.7.3. Application of the guide criteria for preparations

The guidance criteria set out in this Annex are directly applicable when the data in question have been obtained from test methods comparable with those described in Annex V with the exception of the criteria of Chapter 4

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

for which only the conventional method is applicable. A conventional method is also applicable in relation to the criteria of Chapter 5, with the exception of aquatic toxicity, subject to the conditions referred to in Annex III Part C of Directive 1999/45/EC. For preparations covered by Directive 91/414/EEC data for classification and labelling are also acceptable from other internationally recognised methods (see special provisions in Section 1.6 of this Annex). In other cases, the available data must be evaluated by comparing the test methods employed with those indicated in Annex V and the rules specified in this Annex for determining the appropriate classification and labelling.

Where the health and environmental hazards are assessed by applying a conventional method referred to in Articles 6 and 7 and Annexes II and III of Directive 1999/45/EC the individual concentration limits to be used are those set out either:

- in Annex I to this Directive, or
- in Annex II Part B and/or Annex III Part B of Directive 1999/45/EC where the substance or substances do not appear in Annex I to this Directive or appear in it without concentration limits.

In the case of preparations containing mixtures of gases, classification with respect to the health and environmental effects will be established by the calculation method on the basis of the individual concentration limits from Annex I to this Directive or when these limits are not in Annex I on the basis of the criteria of Annexes II and III of Directive 1999/45/EC.

1.7.3.1. Preparations or substances described in Section 1.7.2.1 used as constituents of another preparation

The labelling of such preparations must be in conformity with the provisions of Article 10 according to the principles set out in Articles 3 and 4 of Directive 1999/45/EC. However, in certain cases, the information on the label of the preparation or substance described in Section 1.7.2.1 is insufficient to enable other manufacturers who wish to use it as a constituent of their own preparation(s) to carry out the classification and labelling of their preparation(s) correctly.

In these cases, the person established within the Community responsible for placing the original preparation or substance described in Section 1.7.2.1 on the market, whether it be the manufacturer, the importer or the distributor shall supply upon justified request and as soon as possible all necessary data

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

concerning the dangerous substances present to enable correct classification and labelling of the new preparation. This data is also necessary to enable the person responsible for placing the new preparation on the market to comply with other requirements of Directive 1999/45/EC.

2. CLASSIFICATION ON THE BASIS OF PHYSICOCHEMICAL PROPERTIES

2.1. Introduction

The test methods relating to explosive, oxidising and flammable properties included in Annex V serve to give specific meaning to the general definitions given in Article 2 (2) (a) to (e). Criteria follow directly from the test methods in Annex V as far as they are mentioned.

If adequate information is available to demonstrate in practice that the physicochemical properties of substances and preparations (apart from organic peroxides) are different from those revealed by the test methods given in Annex V, then such substances and preparations should be classified according to the hazard they present, if any, to those handling the substances and preparations or to other persons.

2.2. Criteria for classification, choice of symbols, indication of danger and choice of risk phrases

In the case of preparations, the criteria referred to in Article 5 of Directive 1999/45/EC need to be taken into consideration.

2.2.1. Explosive

Substances and preparations shall be classified as explosive and assigned the symbol 'E' and the indication of danger 'explosive' in accordance with the results of the tests given in Annex V and in so far as the substances and preparations are explosive as placed on the market. One risk phrase is obligatory, it is to be specified on the basis of the following:

- R2 Risk of explosion by shock, friction, fire or other sources of ignition
 - substances and preparations except those set out below.
- R3 Extreme risk of explosion by shock, friction, fire or other source of ignition

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- substances and preparations which are particularly sensitive such as picric acid salts or PETN.

2.2.2. Oxidising

Substances and preparations shall be classified as oxidising and assigned the symbol 'O'and the indication of danger 'oxidising' in accordance with the results of the tests given in Annex V. One risk phrase is obligatory, it is to be specified on the basis of the test results but subject to the following:

R7 May cause fire

- organic peroxides which have flammable properties even when not in contact with other combustible material.

R8 Contact with combustible material may cause fire

- other oxidising substances and preparations, including inorganic peroxides, which may cause fire or enhance the risk of fire when in contact with combustible material.

R9 Explosive when mixed with combustible material

- other substances and preparations, including inorganic peroxides, which become explosive when mixed with combustible materials, e.g. certain chlorates.

2.2.2.1. Remarks concerning peroxides

For the explosive properties, an organic peroxide or preparation thereof in the form in which it is placed on the market is classified according to the criteria in section 2.2.1. on the basis of tests carried out in accordance with the methods given in Annex V.

For the oxidising properties the existing methods in Annex V cannot be applied to organic peroxides.

For substances, organic peroxides not already classified as explosive are classified as dangerous on the basis of their structure (e.g. R-O-O-H; R1-O-O-R2).

Preparations not already classified as explosive shall be classified using the calculation method based on the percentage of active oxygen shown in Section 9.5.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Any organic peroxide or preparation thereof not already classified as explosive is classified as oxidising, if the peroxide or its formulation contains:

- more than 5 % of organic peroxides, or
- more than 0.5 % available oxygen from the organic peroxides, and more than 5 % hydrogen peroxide

2.2.3. Extremely flammable

Substances and preparations shall be classified as extremely flammable and assigned the symbol 'F+' and the indication of danger 'extremely flammable' in accordance with the results of the tests given in Annex V. The risk phrase shall be assigned in accordance with the following criteria:

R12 Extremely flammable

- Liquid substances and preparations which have a flash point lower than 0 °C and a boiling point (or in case of a boiling range the initial boiling point) lower than or equal to 35 °C.
- Gaseous substances and preparations which are flammable in contact with air at ambient temperature and pressure.

2.2.4. Highly flammable

Substances and preparations shall be classified as highly flammable and assigned the symbol 'F' and the indication of danger 'highly flammable' in accordance with the results of the tests given in Annex V. Risk phrases shall be assigned in accordance with the following criteria:

R11 Highly flammable

- Solid substances and preparations which may readily catch fire after brief contact with a source of ignition and which continue to burn or to be consumed after removal of the source of ignition.
- Liquid substances and preparations having a flash point below 21°C but which are not extremely flammable.

R15 Contact with water liberates extremely flammable gases

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

 Substances and preparations which, in contact with water or damp air, evolve extremely flammable gases in dangerous quantities, at a minimum rate of one litre per kilogram per hour.

R17 Spontaneously flammable in air

- Substances and preparations which may become hot and finally catch fire in contact with air at ambient temperature without any input of energy.

2.2.5. Flammable

Substances and preparations shall be classified as flammable in accordance with the results of the tests given in Annex V. The risk phrase shall be assigned in accordance with the criteria mentioned below.

R10 Flammable

- Liquid substances and preparations having a flash point equal to or greater than 21 °C, and less than or equal to 55 °C.

However, in practice it has been shown that a preparation having a flash point equal to or greater than 21 °C and less than or equal to 55 °C need not be classified as flammable if the preparation could not in any way support combustion and only so long as there is no reason to fear risks to those handling these preparations or to other persons.

2.2.6. Other physicochemical properties

Additional risk phrases shall be assigned to substances and preparations which have been classified by virtue of Sections 2.2.1 to 2.2.5 above or by Chapter 3, 4 and 5 below, in accordance with the following criteria (based on experience obtained during compilation of Annex I):

R1 Explosive when dry

For explosive substances and preparations put on the market in solution or in a wetted form, e.g. nitrocellulose with more than 12.6 % nitrogen.

R4 Forms very sensitive explosive metallic compounds

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

For substances and preparations which may form sensitive explosive metallic derivatives, e.g. picric acid, styphnic acid.

R5 Heating may cause an explosion

For thermally unstable substances and preparations not classified as explosive, e.g. perchloric acid > 50 %.

R6 Explosive with or without contact with air

For substances and preparations which are unstable at ambient temperatures, e.g. acetylene.

R7 May cause fire

For reactive substances and preparations, e.g. fluorine, sodium hydrosulphite.

R14 Reacts violently with water

For substances and preparations which react violently with water, e.g. acetyl chloride, alkali metals, titanium tetrachloride.

R16 Explosive when mixed with oxidising substances

For substances and preparations which react explosively with an oxidising agent, e.g. red phosphorus.

R18 In use, may form flammable/explosive vapour-air mixture

For preparations not in themselves classified as flammable, which contain volatile components which are flammable in air.

R19 May form explosive peroxides

For substances and preparations which may form explosive peroxides during storage, e.g. diethyl ether, 1,4- dioxan

R30 Can become highly flammable in use

For preparations not in themselves classified as flammable, which may become flammable due to the loss of non-flammable volatile components.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

R44 Risk of explosion if heated under confinement

For substances and preparations not in themselves classified as explosive in accordance with Section 2.2.1 above but which may nevertheless display explosive properties in practice if heated under sufficient confinement. For example, certain substances which would decompose explosively if heated in a steel drum do not show this effect if heated in less-strong containers.

For other additional risk phrases see Section 3.2.8.

3. CLASSIFICATION ON THE BASIS OF TOXICOLOGICAL PROPERTIES

3.1. Introduction

3.1.1. Classification is concerned with both the acute and long-term effects of substances and preparations, whether resulting from a single instance of exposure or repeated or prolonged exposure.

Where it can be demonstrated by epidemiological studies, by scientifically valid case studies as specified in this Annex or by statistically backed experience, such as the assessment of data from poison information units or concerning occupational diseases, that toxicological effects on man differ from those suggested by the application of the methods outlined in Section 1.6 of this Annex, then the substance or preparation shall be classified according to its effects on man. However, tests on man should be discouraged and should not normally be used to negate positive animal data

.

Directive 86/609/EEC seeks to protect animals used for experimental and other scientific purposes. For several endpoints there are validated in vitro test methods in Annex V of this Directive and these tests should be used where appropriate.

- 3.1.2. The classification of substances must be made on the basis of the experimental data available in accordance with the following criteria which take into account the magnitude of these effects:
 - (a) for acute toxicity (lethal and irreversible effects after a single exposure), the criteria under Sections 3.2.1 to 3.2.3 are to be used,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (b) for sub-acute, sub-chronic or chronic toxicity the criteria under Sections 3.2.2 to 3.2.4 are to be used,
- (c) for corrosive and irritant effects the criteria under Sections 3.2.5 and 3.2.6 are to be used,
- (d) for sensitising effects the criteria under Section 3.2.7 are to be used,
- (e) for specific effects on health (carcinogenicity, mutagenicity and reproductive toxicity), the criteria in Chapter 4 are to be used.
- 3.1.3. For preparations, the classification relating to dangerous for health is carried out:
 - (a) on the basis of a conventional method referred to in Article 6 and Annex II of Directive 1999/45/EC in the absence of experimental data. In this case, the classification is based on the individual concentration limits:
 - either taken from Annex I to this Directive, or
 - from Annex II, Part B of Directive 1999/45/EC where the substance or substances do not appear in Annex I of this Directive or appear in it without concentration limits.
 - (b) or when experimental data are available, according to the criteria described under Sections 3.1.2 excluding the carcinogenic, mutagenic and toxic to reproduction properties referred to under 3.1.2 (e) which must be evaluated by a conventional method referred to in Article 6 and Annex II, Parts A. 7. 9. And B. 6. Of Directive 1999/45/EC.

Note: Without prejudice to requirements of Directive 91/414/EEC, only where it can be scientifically demonstrated by the person responsible for placing the preparation on the market that the toxicological properties of the preparation cannot correctly be determined by the method outlined in paragraph 3.1.3 (a), or on the basis of existing test results on animals, the methods outlined in paragraph 3.1.3 (b) may be used, provided they are justified or specifically authorised under Article 12 of Directive 86/609/EEC.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Whichever method is used for the evaluation of the danger of a preparation, all the dangerous effects on health as defined in Annex II, Part B of Directive 1999/45/EC must be taken into consideration.

- 3.1.4. When the classification is to be established from experimental results obtained in animal tests the results should have validity for man in that the tests reflect, in an appropriate way, the risks to man.
- 3.1.5. The acute oral toxicity of substances or preparations placed on the market may be established either by a method permitting assessment of the LD50 value, or by determining the discriminating dose (the fixed dose method, or by determining the range of exposure where lethality is expected (the acute toxic class method).
- 3.1.5.1. The discriminating dose is the dose which causes evident toxicity but not mortality and must be one of the four dosage levels specified in Annex V (5, 50, 500 or 2 000 mg per kg body weight).

The concept 'evident toxicity' is used to designate toxic effects, after exposure to the substance tested, which are so severe that exposure to the next highest fixed dose would probably lead to mortality.

The results of testing at a particular dose following the fixed dose method may be either:

- less than 100 % survival,
- 100 % survival, but evident toxicity,
- 100 % survival, but no evident toxicity.

In the criteria in sections 3.2.1, 3.2.2 and 3.2.3 only the final test result is shown. The 2 000 mg/kg dose should be used primarily to obtain information on the toxic effects of substances which are of low acute toxicity and which are not classified on the basis of acute toxicity.

The fixed dose method requires in some cases testing at higher or lower doses, if not already tested at the relevant dose level. Refer also to the evaluation table in test method B.1 bis.

3.1.5.2. The range of exposure where lethality is expected is derived from the observed absence or presence of substance related mortality following the acute toxic class method. For initial testing one of three fixed starting doses (25, 200 or 2 000 mg per kg body weight) is used.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

The acute toxic class method requires in some cases testing at higher or lower doses, if not already tested at the relevant dose level. Refer also to the test procedure flow charts in test method B.1 tris of Annex V.

3.2. Criteria for classification, choice of symbols, indication of danger, choice of risk phrases

3.2.1. Very toxic

Substances and preparations shall be classified as very toxic, and assigned the symbol 'T+' and indication of danger 'very toxic' in accordance with the criteria specified below.

Risk phrases shall be assigned in accordance with the following criteria:

R28 Very toxic if swallowed

Acute toxicity results:

- LD50 oral, rat \leq 25 mg/kg,
- less than 100 % survival at 5 mg/kg oral, rat by the fixed dose procedure, or
- high mortality at doses < 25 mg/kg oral, rat, by the acute toxic class method (for test result interpretation see flow charts in Annex 2 of test method B.1 tris of Annex V).

R27 Very toxic in contact with skin

Acute toxicity results:

- LD₅₀ dermal, rat or rabbit: \leq 50 mg/kg.

R26 Very toxic by inhalation

Acute toxicity results:

- LC50 inhalation, rat, for aerosols or particulates: ≤ 0.25 mg/litre/4hr,
- LC50 inhalation, rat, for gases and vapours: < 0.5 mg/litre/4hr.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

R39 Danger of very serious irreversible effects

- Strong evidence that irreversible damage other than the effects referred to in Chapter 4 is likely to be caused by a single exposure by an appropriate route, generally in the abovementioned dose range.

In order to indicate the route of administration/exposure one of the following combinations shall be used: R39/26, R39/27, R39/28, R39/26/27, R39/26/28, R39/27/28, R39/26/27/28.

3.2.2. Toxic

Substances and preparations shall be classified as toxic and assigned the symbol 'T' and the indication of danger 'toxic' in accordance with the criteria specified below. Risk phrases shall be assigned in accordance with the following criteria.

R25 Toxic if swallowed

Acute toxicity results:

- LD50 oral, rat: $25 < LD50 \le 200 \text{mg/kg}$,
- Discriminating dose, oral, rat, 5 mg/kg: 100 % survival but evident toxicity, or
- high mortality in the dose range > 25 to ≤ 200 mg/kg oral, rat, by the acute toxic class method (for test result interpretation see flow charts in Annex 2 of test method B.1 tris of Annex V).

R24 Toxic in contact with skin

Acute toxicity results:

- LD₅₀ dermal, rat or rabbit: 50 < LD50 < 400 mg/kg.

R23 Toxic by inhalation

Acute toxicity results:

- LC₅₀ inhalation, rat, for aerosols or particulates: $0.25 < LC_{50} \le 1$ mg/litre/4hr,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- LC₅₀ inhalation, rat, for gases and vapours: $0.5 < LC_{50} \le 2$ mg/litre/4hr.

R39 Danger of very serious irreversible effects

- strong evidence that irreversible damage other than the effects referred to in Chapter 4 is likely to be caused by a single exposure by an appropriate route, generally in the abovementioned dose range.

In order to indicate the route of administration/exposure one of the following combinations shall be used: R39/23, R39/24, R39/25, R39/24/25, R39/23/24/25.

R48 Danger of serious damage to health by prolonged exposure

 serious damage (clear functional disturbance or morphological change which have toxicological significance) is likely to be caused by repeated or prolonged exposure by an appropriate route.

Substances and preparations are classified at least as toxic when these effects are observed at levels of one order of magnitude lower (i.e. 10-fold) than those set out for R48 in Section 3.2.3.

In order to indicate the route of administration/exposure one of the following combinations shall be used: R48/23, R48/24, R48/25, R48/23/24, R48/23/25, R48/24/25, R48/23/24/25.

3.2.3. Harmful

Substances and preparations shall be classified as harmful and assigned the symbol 'Xn' and the indication of danger 'harmful' in accordance with the criteria specified below. Risk phrases shall be assigned in accordance with the following criteria:

R22 Harmful if swallowed

Acute toxicity results:

- LD₅₀ per oral, rat: $200 < LD_{50} < 2000 \text{ mg/kg}$,
- discriminating dose, oral, rat, 50 mg/kg: 100 % survival but evident toxicity,

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- less than 100 % survival at 500 mg/kg, rat oral by the fixed dose procedure. Refer to the evaluation table in the test method B.1 bis of Annex V, or
- high mortality in the dose range > 200 to $\le 2\,000$ mg/kg oral, rat, by the acute toxic class method (for test result interpretation see flow charts in Annex 2 of test method B.1 tris of Annex V).

R21 Harmful in contact with skin

Acute toxicity results:

LD₅₀ dermal, rat or rabbit: $400 < LD_{50} \le 2000$ mg/kg.

R20 Harmful by inhalation

Acute toxicity results:

- LC₅₀ inhalation, rat, for aerosols or particulates: $1 < LC_{50} \le 5$ mg/litre/4hr,
- LC50 inhalation, rat, for gases or vapours: $2 < LC50 \le 20$ mg/litre/4hr.

R65 Harmful: may cause lung damage if swallowed

Liquid substances and preparations presenting an aspiration hazard in humans because of their low viscosity:

- (a) For substances and preparations containing aliphatic, alicyclic and aromatic hydrocarbons in a total concentration equal to or greater than 10 % and having either
- a flow time of less than 30 sec. in a 3 mm ISO cup according to ISO 2431 (April 1996 / July 1999 edition) relating to 'Paints and varnishes - Determination of flow time by use of flow cups',
- a kinematic viscosity measured by a calibrated glass capillary viscometer in accordance with ISO 3104/3105 of less than 7 x 10-6 m2/sec at 40° C (ISO 3104, 1994 edition, relating to 'Petroleum products Transparent and opaque liquids -

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Determination of kinematic viscosity and calculation of dynamic viscosity'; ISO 3105, 1994 edition, relating to 'Glass capillary kinematic viscometers - Specifications and operating instructions'), or

- a kinematic viscosity derived from measurements of rotational viscometry in accordance with ISO 3219 of less than 7 x 10-6 m2/sec at 40° C (ISO 3219, 1993 edition, relating to 'Plastics - Polymers/resins in the liquid state or as emulsions or dispersions - Determination of viscosity using a rotational viscometer with defined shear rate').

Note that substances and preparations meeting these criteria need not be classified if they have a mean surface tension greater than 33mN/m at 25° C as measured by the du Nouy tensiometer or by the test methods shown in Annex V Part A.5.

(b) For substances and preparations, based on practical experience in humans

R68 Possible risk of irreversible effects

- strong evidence that irreversible damage other than the effects referred to in Chapter 4 is likely to be caused by a single exposure by an appropriate route, generally in the abovementioned dose range.

In order to indicate route of administration/exposure one of the following combinations shall be used: R68/20, R68/21, R68/22, R68/20/21, R68/20/22, R68/21/22, R68/20/21/22.

R48 Danger of serious damage to health by prolonged exposure

- serious damage (clear functional disturbance or morphological change which has toxicological significance) is likely to be caused by repeated or prolonged exposure by an appropriate route.

Substances and preparations are classified at least as harmful when these effects are observed at levels of the order of:

- oral, rat < 50 mg/kg (bodyweight)/day,
- dermal, rat or rabbit ≤ 100 mg/kg (bodyweight)/day,

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- inhalation, rat ≤ 0.25 mg/l, 6h/day.

These guide values can apply directly when severe lesions have been observed in a sub-chronic (90 days) toxicity test. When interpreting the results of a sub-acute (28 days) toxicity test these figures should be increased approximately three fold. If a chronic (two years) toxicity test is available it should be evaluated on a case-by-case basis. If results of studies of more than one duration are available, then those from the study of the longest duration should normally be used.

In order to indicate route of administration/exposure one of the following combinations shall be used: R48/20, R48/21, R48/22, R48/20/21, R48/20/22, R48/21/22, R48/20/21/22.

3.2.3.1. Comments regarding volatile substances

For certain substances with a high saturated vapour concentration evidence may be available to indicate effects that give cause for concern. Such substances may not be classified under the criteria for health effects in this guide (3.2.3) or not covered by Section 3.2.8. However, where there is appropriate evidence that such substances may present a risk in normal handling and use then classification on a case-by-case basis in Annex I may be necessary.

3.2.4. Comments regarding the use of R48

Use of this risk phrase refers to the specific range of biological effects within the terms described below. For application of this risk phrase serious damage to health is to be considered to include death, clear functional disturbance or morphological changes which are toxicologically significant. It is particularly important when these changes are irreversible. It is also important to consider not only specific severe changes in a single organ or biological system but also generalised changes of a less severe nature involving several organs, or severe changes in general health status.

When assessing whether there is evidence for these types of effects reference should be made to the following guidelines:

- 1. Evidence indicating that R48 should be applied:
 - (a) substance-related deaths;
 - (b)

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (i) major functional changes in the central or peripheral nervous systems, including sight, hearing and the sense of smell, assessed by clinical observations or other appropriate methods (e.g. electrophysiology);
- (ii) major functional changes in other organ systems (for example the lung);
- (c) any consistent changes in clinical biochemistry, haematology or urinalysis parameters which indicate severe organ dysfunction. Haematological disturbances are considered to be particularly important if the evidence suggests that they are due to decreased bone marrow production of blood cells;
- (d) severe organ damage noted on microscopic examination following autopsy:
 - (i) widespread or severe necrosis, fibrosis or granuloma formation in vital organs with regenerative capacity (e.g. liver);
 - (ii) severe morphological changes that are potentially reversible but are clear evidence of marked organ dysfunction (e.g. severe fatty change in the liver, severe acute tubular nephrosis in the kidney, ulcerative gastritis); or
 - (iii) evidence of appreciable cell death in vital organs incapable of regeneration (e.g. fibrosis of the myocardium or dying back of a nerve) or in stem cell populations (e.g. aplasia or hypoplasia of the bone marrow).

The above evidence will most usually be obtained from animal experiments. When considering data derived from practical experience special attention should be given to exposure levels.

2. Evidence indicating that R48 should not be applied:

The use of this risk phrase is restricted to 'serious damage to health by prolonged exposure'. A number of substance-related effects may be observed in both humans and animals that would not justify the use of R48.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

These effects are relevant when attempting to determine a no-effect level for a chemical substance.

Examples of well documented changes which would not normally justify classification with R48, irrespective of their statistical significance, include:

- (a) clinical observations or changes in bodyweight gain, food consumption or water intake, which may have some toxicological importance but which do not, by themselves, indicate 'serious damage';
- (b) small changes in clinical biochemistry, haematology or urinalysis parameters which are of doubtful or minimal toxicological importance;
- (c) changes in organ weights with no evidence of organ dysfunction;
- (d) adaptative responses (e.g. macrophage migration in the lung, liver hypertrophy and enzyme induction, hyperplastic responses to irritants). Local effects on the skin produced by repeated dermal application of a substance which are more appropriately classified with R38 'irritating to skin'; or
- (e) where a species-specific mechanism of toxicity (e.g. specific metabolic pathways) has been demonstrated.

3.2.5. Corrosive

The substance or preparation shall be classified as corrosive and assigned the symbol 'C' and the indication of danger 'corrosive' in accordance with the following criteria:

- A substance or a preparation is considered to be corrosive if, when it is applied to healthy intact animal skin, it produces full thickness destruction of skin tissue on at least one animal during the test for skin irritation cited in Annex V or during an equivalent method.
- Classification can be based on the results of a validated in vitro test, such as that cited in Annex V (B.40. Skin corrosion: rat skin transcutaneous electrical resistance assay and human skin model assay.)

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- A substance or a preparation should also be considered corrosive if the result can be predicted, for example from strongly acid or alkaline reactions indicated by a pH of 2 or less or 11.5 or greater. However, where extreme pH is the basis for classification, acid/alkali reserve¹ may also be taken into consideration. If consideration of alkali/acid reserve suggests the substance or preparation may not be corrosive then further testing should be carried out to confirm this, preferably by use of an appropriate validated in vitro test. Consideration of acid/alkali reserve should not be used alone to exonerate substances or preparations from classification as corrosive.

Risk phrases shall be assigned in accordance with the following criteria:

R35 Causes severe burns

- if, when applied to healthy intact animal skin, full thickness destruction of skin tissue occurs as a result of up to three minutes exposure, or if this result can be predicted.

R34 Causes burns

- if, when applied to healthy intact animal skin, full thickness destruction of skin tissue occurs as a result of up to four hours exposure, or if this result can be predicted,
- organic hydroperoxides, except where evidence to the contrary is available.

Notes:

Where classification is based on results of a validated in vitro test R35 or R34 should be applied according to the capacity of the test method to discriminate between these.

Where classification is based upon consideration of extreme pH alone, R35 should be applied.

3.2.6. Irritant

_

¹ J.R. Young, M.J. How, A.P. Walker and W.M.H. Worth (1988) "Classification as corrosive or irritant to skin of preparations containing acidic or alkaline substances, without testing on animals" *Toxic. In Vitro* **2(1)**: 19-26

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Substances and preparations shall be classified as irritant and assigned the symbol 'Xi' and the indication of danger 'irritant' in accordance with the criteria given below.

3.2.6.1. Inflammation of the skin

The following risk phrase shall be assigned in accordance with the criteria given:

R38 Irritating to skin

- Substances and preparations which cause significant inflammation of the skin which persists for at least 24 hours after an exposure period of up to four hours determined on the rabbit according to the cutaneous irritation test method cited in Annex V.

Inflammation of the skin is significant if:

- (a) the mean value of the scores for either erythema and eschar formation or oedema formation, calculated over all the animals tested, is 2 or more; or
- (b) in the case where the Annex V test has been completed using three animals, either erythema and eschar formation or oedema formation equivalent to a mean value of 2 or more calculated for each animal separately has been observed in two or more animals.

In both cases all scores at each of the reading times (24, 48 and 72 hr) for an effect should be used in calculating respective mean values.

Inflammation of the skin is also significant if it persists in at least two animals at the end of the observation time. Particular effects e.g. hyperplasia, scaling, discoloration, fissures, scabs and alopecia should be taken into account.

Relevant data may also be available from non-acute animal studies (see comments on R48, section 2.d). These are considered significant if the effects seen are comparable to those described above.

- Substances and preparations which cause significant inflammation of the skin, based on practical observations in humans on immediate, prolonged or repeated contact.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- Organic peroxides, except where evidence to the contrary is available.

Paresthesia:

Paresthesia caused in humans by skin contact with pyrethroid pesticides is not regarded as an irritant effect justifying classification as Xi; R38. The Sphrase S24 should however be applied for substances seen to cause this effect.

3.2.6.2. Ocular lesions

The following risk phrases shall also be assigned in accordance with the criteria given:

R36 Irritating to eyes

- Substances and preparations which, when applied to the eye of the animal, cause significant ocular lesions which occur within 72 hours after exposure and which persist for at least 24 hours.

Ocular lesions are significant if the mean scores of the eye irritation test cited in Annex V have any of the following values:

- cornea opacity equal to or greater than 2 but less than 3,
- iris lesion equal to or greater than 1 but not greater than 1.5,
- redness of the conjunctivae equal to or greater than 2.5,
- oedema of the conjunctivae (chemosis) equal to or greater than 2,

or, in the case where the Annex V test has been completed using three animals if the lesions, on two or more animals, are equivalent to any of the above values except that for iris lesion the value should be equal to or greater than 1 but less than 2 and for redness of the conjunctivae the value should be equal to or greater than 2.5.

In both cases all scores at each of the reading times (24, 48 and 72 hours) for an effect should be used in calculating the respective mean values.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- Substances or preparations which cause significant ocular lesions, based on practical experience in humans.
- Organic peroxides except where evidence to the contrary is available.

R41 Risk of serious damage to eyes

Substances and preparations which, when applied to the eye of the animal cause severe ocular lesions which occur within 72 hours after exposure and which persist for at least 24 hours.

Ocular lesions are severe if the means of the scores of the eye irritation test in Annex V have any of the values:

- cornea opacity equal to or greater than 3,
- iris lesion greater than 1.5.

The same shall be the case where the test has been completed using three animals if these lesions, on two or more animals, have any of the values:

- cornea opacity equal to or greater than 3,
- iris lesion equal to 2.

In both cases all scores at each of the reading times (24, 48 and 72 hours) for an effect should be used in calculating the respective mean values.

Ocular lesions are also severe when they are still present at the end of the observation time.

Ocular lesions are also severe if the substance or preparation causes irreversible colouration of the eyes.

- Substances and preparations which cause severe ocular lesions, based on practical experience in humans.

Note:

When a substance or preparation is classified as corrosive and assigned R34 or R35, the risk of severe damage to eyes is considered implicit and R41 is not included in the label.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

3.2.6.3. Respiratory system irritation

The following risk phrase shall be assigned in accordance with the criteria given:

R37 Irritating to respiratory system

Substances and preparations which cause serious irritation to the respiratory system based on:

- practical observation in humans
- positive results from appropriate animal tests.

Comments regarding the use of R37

In interpreting practical observations in humans, care should be taken to distinguish between effects which lead to classification with R48 (see section 3.2.4.) from those leading to classification with R37. Conditions normally leading to classification with R37 are reversible and usually limited to the upper airways.

Positive results from appropriate animal tests may include data obtained in a general toxicity test, including histopathological data from the respiratory system. Data from the measurement of experimental bradypnea may also be used to assess airway irritation.

3.2.7. Sensitisation

3.2.7.1. Sensitisation by inhalation

Substances and preparations shall be classified as sensitising and assigned the symbol 'Xn', the indication of danger 'Harmful' and the risk phrase R42 in accordance with the criteria given below.

R42 May cause sensitisation by inhalation

- if there is evidence that the substance or preparation can induce specific respiratory hypersensitivity;
- where there are positive results from appropriate animal tests; or

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- if the substance is an isocyanate, unless there is evidence that the specific isocyanate does not cause respiratory hypersensitivity

Comments regarding the use of R42

Human evidence

Evidence that the substance or preparation can induce specific respiratory hypersensitivity will normally be based on human experience. In this context hypersensitivity is normally seen as asthma, but other hypersensitivity reactions such as rhinitis and alveolitis are also considered. The condition will have the clinical character of an allergic reaction. However, immunological mechanisms do not have to be demonstrated.

When considering the evidence from human exposure, it is necessary for a decision on classification to take into account in addition to the evidence from the cases:

- the size of the population exposed
- the extent of exposure.

The evidence referred to above could be:

- clinical history and data from appropriate lung function tests related to exposure to the substance, confirmed by other supportive evidence which may include:
- a chemical structure related to substances known to cause respiratory hypersensitivity;
- anin vivo immunological test (e.g. skin prick test);
- anin vitro immunological test (e.g. serological analysis);
- studies indicating other specific but non-immunological mechanisms of action, e.g. repeated lowlevel irritation, pharmacologically mediated effects; or
- data from a positive bronchial challenge test with the substance conducted according to accepted guidelines for the determination of a specific hypersensitivity reaction.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Clinical history should include both medical and occupational history to determine a relationship between exposure to a specific substance or preparation and development of respiratory hypersensitivity. Relevant information includes aggravating factors both in the home and workplace, the onset and progress of the disease, family history and medical history of the patient in question. The medical history should also include a note of other allergic or airway disorders from childhood, and smoking history.

The results of positive bronchial challenge tests are considered to provide sufficient evidence for classification on their own. It is however recognised that in practice many of the examinations listed above will already have been carried out.

Substances that elicit symptoms of asthma by irritation only in people with bronchial hyperreactivity should not be assigned R42.

Animal studies

Data from tests which may be indicative of the potential of a substance or preparation to cause sensitisation by inhalation in humans may include:

- IgE measurements (e.g. in mice), or
- specific pulmonary responses in guinea pigs.

3.2.7.2. Sensitisation by skin contact

Substances and preparations shall be classified as sensitising and assigned the symbol 'Xi', the indication of danger 'Irritant' and the risk phrase R43 in accordance with the criteria given below:

R43 May cause sensitisation by skin contact

- If practical experience shows the substance or preparation to be capable of inducing a sensitisation by skin contact in a substantial number of persons, or
- where there are positive results from an appropriate animal test.

Comments regarding the use of R43

Human evidence

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

The following evidence (practical experience) is sufficient to classify a substance or preparation with R43:

- Positive data from appropriate patch testing, normally in more than one dermatological clinic, or
- Epidemiological studies showing allergic contacts dermatitis caused by the substance or preparation. Situations in which a high proportion of those exposed exhibit characteristic symptoms are to be lookedat with special concern, even if the number of cases is small, or
- Positive data from experimental studies in man (see also 3.1.1).

The following is sufficient to classify a substance with R43 when there is supportive evidence:

- Isolated episodes of allergic contact dermatitis, or
- Epidemiological studies where chance, bias or confounders have not been ruled out fully with reasonable confidence.

Supportive evidence may include:

- data from animal tests performed according to existing guidelines, with a result that does not meet the criteria given in the section on animal studies but is sufficiently close to the limit to be considered significant, or
- data from non-standard methods, or
- appropriate structure-activity relationships.

Animal studies

Positive results from appropriate animal tests are:

- in the case of the adjuvant type test method for skin sensitisation detailed in Annex V or in the case of other adjuvant-type test methods, a response of at least 30% of the animals is considered positive;
- for any other test method a response of at least 15% of the animals is considered positive.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

3.2.7.3. Immunological contact urticaria

Some substances or preparations, which meet the criteria for R42 may in addition cause immunological contact urticaria. In these cases, information concerning contact urticaria should be included by the use of appropriate Sphrases, usually S24 and S36/37, and in the Safety Data Sheet.

For substances or preparations, which produce signs of immunological contact urticaria which do not fulfil the criteria for R42, consideration should be given to classification with R43.

There is no recognised animal model available to identify substances which cause immunological contact urticaria. Therefore, classification will normally be based on human evidence which will be similar to that for skin sensitisation (R43).

3.2.8. Other toxicological properties

Additional risk phrases shall be assigned in accordance with the following criteria (based on experience obtained during compilation of Annex I) to substances and preparations classified by virtue of 2.2.1. to 3.2.7. above and/or chapters 4 and 5:

R29 Contact with water liberates toxic gas

For substances and preparations which in contact with water or damp air, evolve very toxic/toxic gases in potentially dangerous amounts, e.g. aluminium phosphide, phosphorus pentasulphide.

R31 Contact with acids liberates toxic gas

For substances and preparations which react with acids to evolve toxic gases in dangerous amounts, e.g. sodium hypochlorite, barium polysulphide. For substances used by members of the general public, the use of S50 (do not mix with ... (to be specified by the manufacturer)) would be more suitable.

R32 Contact with acids liberates very toxic gas

For substances and preparations which react with acids to evolve very toxic gases in dangerous amounts; e.g. salts of hydrogen cyanide, sodium azide. For substances used by members of the general public, the use of S50 (do not mix with ... (to be specified by the manufacturer)) would be more suitable.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

R33 Danger of cumulative effects

For substances and preparations when accumulation in the human body is likely and may cause some concern which, however, is not sufficient to justify the use of R48.

For comments on the use of this R-phrase see Section 4.2.3.3 for substances and Annex V, Part A.3. of Directive 1999/45/EC for preparations

.

R64 May cause harm to breastfed babies

For substances and preparations which are absorbed by women and may interfere with lactation or which may be present (including metabolites) in breast milk in amounts sufficient to cause concern for the health of a breastfed child.

For comments on the use of this R-phrase see Section 4.2.3.3 for substances and Annex V, Part A.4. of Directive 1999/45/EC for preparations.

R66 Repeated exposure may cause skin dryness or cracking

For substances and preparations which may cause concern as a result of skin dryness, flaking or cracking but which do not meet the criteria for R38 based on either:

- practical observation after normal handling and use, or
- relevant evidence concerning their predicted effects on the skin.

See also paragraphs 1.6 and 1.7.

R67 Vapours may cause drowsiness and dizziness

For volatile substances and preparations containing such substances which cause clear symptoms of central nervous system depression by inhalation and which are not already classified with respect to acute inhalation toxicity (R20, R23, R26, R68/20, R39/23 or R39/26).

The following evidence may be used:

(a) Data from animal studies showing clear signs of CNS depression such as narcotic effects, lethargy, lack of coordination (including loss of righting reflex) and ataxia either:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- at concentrations/exposure times not exceeding 20 mg/l/4h or,
- for which the ratio of the effect concentration at ≤ 4 h to the saturated vapour concentration (SVC) at 20?C is $\leq 1/10$.
- (b) Practical experience in humans (e.g. narcosis, drowsiness, reduced alertness, loss of reflexes, lack of co-ordination, vertigo) from well documented reports under comparable exposure conditions to the effects specified above for animals.

See also Paragraphs 1.6 and 1.7.

For other supplementary risk phrases see Section 2.2.6.

4. CLASSIFICATION ON THE BASIS OF SPECIFIC EFFECTS ON HUMAN HEALTH

4.1. Introduction

- 4.1.1. This Chapter sets out the procedure for the classification of substances which may have the effects mentioned below. For preparations see Section 4.2.4.
- 4.1.2. If a manufacturer, distributor or importer has information available which indicates that a substance should be classified and labelled in accordance with the criteria given in Section 4.2.1, 4.2.2 or 4.2.3, he shall provisionally label the substance in accordance with these criteria, on the basis of the assessment of the evidence by a competent person.
- 4.1.3. The manufacturer, distributor or importer shall submit as soon as possible a document summarising all relevant information to one Member State in which the substance is placed on the market. Relevant information in this context comprises in particular all available published and unpublished information required for appropriate classification of the substance in question, on the basis of the intrinsic properties according to the categories laid down in Article 2 (2) and in accordance with the criteria in this Annex. The submitted summary document should include a bibliography containing all relevant references, including any relevant unpublished data.
- 4.1.4. Furthermore, a manufacturer, distributor or importer who has new data which are relevant to the classification and labelling of a substance in accordance with the criteria given in Section 4.2.1., 4.2.2. or 4.2.3., shall

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

submit this data as soon as possible to one Member State in which the substance is placed on the market.

4.1.5. To obtain as quickly as possible a harmonised classification for the Community by the procedure defined in Article 28 of this Directive, Member States which have relevant information available justifying the classification of a substance in one of these categories, whether submitted by the manufacturer or not, should forward such information together with suggestions for classification and labelling, to the Commission as soon as possible.

The Commission will forward to the other Member States the classification and labelling proposal that it receives. Any Member State may ask the Commission for the information it has received.

Any Member State which has good reason to believe that the suggested classification and labelling is inappropriate as far as the carcinogenic, mutagenic or reproductive toxicity effects are concerned shall notify the Commission thereof.

4.2. Criteria for classification, indication of danger, choice of risk phrases

4.2.1. Carcinogenic substances

For the purpose of classification and labelling, and having regard to the current state of knowledge, such substances are divided into three categories:

Category 1

Substances known to be carcinogenic to man. There is sufficient evidence to establish a causal association between human exposure to a substance and the development of cancer.

Category 2

Substances which should be regarded as if they are carcinogenic to man. There is sufficient evidence to provide a strong presumption that human exposure to a substance may result in the development of cancer, generally on the basis of:

appropriate long-term animal studies,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

other relevant information.

Category 3

Substances which cause concern for man owing to possible carcinogenic effects but in respect of which the available information is not adequate for making a satisfactory assessment. There is some evidence from appropriate animal studies, but this is insufficient to place the substance in Category 2. 4.2.1.1. The following symbols and specific risk phrases apply:

Categories 1 and 2:

Substances classified as carcinogenic category 1 or 2 shall be assigned the symbol 'T' and the risk phrase R45 May cause cancer

However, substances and preparations which present a carcinogenic risk only when inhaled, for example, as dust, vapour or fumes, (other routes of exposure e.g. by swallowing or in contact with skin do not present any carcinogenic risk), shall be assigned the symbol 'T' and the risk phrase

R49 May cause cancer by inhalation

Category 3:

Substances classified as carcinogenic category 3 shall be assigned the symbol 'Xn' and the risk phrase R40 Limited evidence of a carcinogenic effect

4.2.1.2. Comments regarding the categorisation of carcinogenic substances

The placing of a substance into Category 1 is done on the basis of epidemiological data; placing into Categories 2 and 3 is based primarily on animal experiments.

For classification as a Category 2 carcinogen either positive results in two animal species should be available or clear positive evidence in one species, together with supporting evidence such as genotoxicity data, metabolic or biochemical studies, induction of benign tumours, structural relationship with other known carcinogens, or data from epidemiological studies suggesting an association.

Category 3 actually comprises 2 sub-categories:

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- (a) substances which are well investigated but for which the evidence of a tumour-inducing effect is insufficient for classification in Category 2. Additional experiments would not be expected to yield further relevant information with respect to classification;
- (b) substances which are insufficiently investigated. The available date are inadequate, but they raise concern for man. This classification is provisional; further experiments are necessary before a final decision can be made.

For a distinction between Categories 2 and 3 the arguments listed below are relevant which reduce the significance of experimental tumour induction in view of possible human exposure. These arguments, especially in combination, would lead in most cases to classification in Category 3, even though tumours have been induced in animals:

- carcinogenic effects only at very high dose levels exceeding the 'maximal tolerated dose'. The maximal tolerated dose is characterised by toxic effects which, although not yet reducing lifespan, go along with physical changes such as about 10 % retardation in weight gain,
- appearance of tumours, especially at high dose levels, only in particular organs of certain species known to be susceptible to a high spontaneous tumour formation,
- appearance of tumours, only at the site of application, in very sensitive test systems (e.g., i.p. or s.c. application of certain locally active compounds), if the particular target is not relevant to man,
- lack of genotoxicity in short-term tests in vivo and in vitro,
- existence of a secondary mechanism of action with the implication of a practical threshold above a certain dose level (e.g., hormonal effects on target organs or on mechanisms of physiological regulation, chronic stimulation of cell proliferation),
- existence of a species specific mechanism of tumour formation (e.g. by specific metabolic pathways) irrelevant for man.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

For a distinction between Category 3 and no classification arguments are relevant which exclude a concern for man:

- a substance should not be classified in any of the categories if the mechanism of experimental tumour formation is clearly identified, with good evidence that this process cannot be extrapolated to man,
- if the only available tumour data are liver tumours in certain sensitive strains of mice, without any other supplementary evidence, the substance may not be classified in any of the categories,
- particular attention should be paid to cases where the only available tumour data are the occurrence of neoplasms at sites and in strains where they are well known to occur spontaneously with a high incidence.

4.2.2. Mutagenic substances

4.2.2.1. For the purposes of classification and labelling, and having regard to the current state of knowledge, such substances are divided into three categories:

Category 1

Substances known to be mutagenic to man.

There is sufficient evidence to establish a causal association between human exposure to a substance and heritable genetic damage.

Category 2

Substances which should be regarded as if they are mutagenic to man.

There is sufficient evidence to provide a strong presumption that human exposure to the substance may result in the development of heritable genetic damage, generally on the basis of:

- appropriate animal studies,
- other relevant information.

Category 3

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Substances which cause concern for man owing to possible mutagenic effects. There is evidence from appropriate mutagenicity studies, but this is insufficient to place the substance in Category 2.

4.2.2.2. The following symbols and specific risk phrases apply:

Categories 1 and 2:

Substances classified as mutagenic category 1 or 2 shall be assigned the symbol 'T' and the risk phrase

R46 May cause heritable genetic damage

Category 3:

Substances classified as mutagenic category 3 shall be assigned the symbol 'Xn' and the risk phrase

R68 Possible risk of irreversible effects

4.2.2.3. Comments regarding the categorisation of mutagenic substances

Definition of terms:

A mutation is a permanent change in the amount or structure of the genetic material in an organism, resulting in a change of the phenotypic characteristics of the organism. The alterations may involve a single gene, a block of genes, or a whole chromosome. Effects involving single genes may be a consequence of effects on single DNA bases (point mutations) or of large changes, including deletions, within the gene. Effects on whole chromosomes

may involve structural or numerical changes. A mutation in the germ cells in sexually reproducing organisms may be transmitted to the offspring. A mutagen is an agent that gives rise to an enhanced occurrence of mutations.

It should be noted that substances are classified as mutagens with specific reference to inherited genetic damage. However, the type of results leading to classification of chemicals in Category 3: 'induction of genetically relevant events in somatic cells', is generally also regarded as an alert for possible carcinogenic activity.

Method development for mutagenicity testing is an ongoing process. For many new tests no standardised protocols and evaluation criteria are

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

presently available. For the evaluation of mutagenicity data the quality of the test performance and the degree of validation of the test method have to be considered.

Category 1

To place a substance in Category 1, positive evidence from human mutation epidemiology studies will be needed. Examples of such substances are not known to date. It is recognised that it is extremely difficult to obtain reliable information from studies on the incidence of mutations in human populations, or on possible increases in their frequencies.

Category 2

To place a substance in Category 2, positive results are needed from assays showing (a) mutagenic effects, or (b) other cellular interactions relevant to mutagenicity, in germ cells of mammals in vivo, or (c) mutagenic effects in somatic cells of mammals in vivo in combination with clear evidence that the substance or a relevant metabolite reaches the germ cells.

With respect to placement in Category 2, at present the following methods are appropriate:

2 (a) in vivo germ cell mutagenicity assays:

- specific locus mutation test,
- heritable translocation test,
- dominant lethal mutation test.

These assays actually demonstrate the appearance of affected progeny or a defect in the developing embryo.

2 (b) in vivo assays showing relevant interaction with germ cells (usually DNA):

- assays for chromosomal abnormalities, as detected by cytogenetic analysis, including aneuploidy, caused by malsegregation of chromosomes,
- test for sister chromatid exchanges (SCEs),
- test for unscheduled DNA synthesis (UDS),

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- assay of (covalent) binding of mutagen to germ cell DNA,
- assaying other kinds of DNA damage.

These assays provide evidence of a more or less indirect nature. Positive results in these assays would normally be supported by positive results from in vivo somatic cell mutagenicity assays, in mammals or in man (see under Category 3, preferably methods as under 3 (a)).

2 (c) in vivo assays showing mutagenic effects in somatic cells of mammals (see under 3 (a)), in combination with toxicokinetic methods, or other methodologies capable of demonstrating that the compound or a relevant metabolite reaches the germ cells.

For 2 (b) and 2 (c), positive results from host-mediated assays or the demonstration of unequivocal effects in in vitro assays can be considered as supporting evidence.

Category 3

To place a substance in Category 3, positive results are needed in assays showing (a) mutagenic effects or (b) other cellular interaction relevant to mutagenicity, in somatic cells in mammals in vivo. The latter especially would normally be supported by positive results from in vitro mutagenicity assays.

For effects in somatic cells in vivo at present the following methods are appropriate:

- 3 (a) in vivo somatic cell mutagenicity assays:
 - bone marrow micronucleus test or metaphase analysis,
 - metaphase analysis of peripheral lymphocytes,
 - mouse coat colour spot test.
- 3 (b) in vivo somatic cell DNA interaction assays:
 - test for SCEs in somatic cells.
 - test for UDS in somatic cells,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- assay for the (covalent) binding of mutagen to somatic cell DNA,
- assay for DNA damage, e.g. by alkaline elution, in somatic cells

Substances showing positive results only in one or more in vitro mutagenicity assays should normally not be classified. Their further investigation using in vivo assays, however, is strongly indicated. In exceptional cases, e.g. for a substance showing pronounced responses in several in vitro assays, for which no relevant in vivo data are available, and which shows resemblance to known mutagens/carcinogens, classification in Category 3 could be considered.

4.2.3. Substances toxic to reproduction

4.2.3.1. For the purposes of classification and labelling and having regard to the present state of knowledge, such substances are divided into 3 categories:

Category 1:

Substances known to impair fertility in humans

There is sufficient evidence to establish a causal relationship between human exposure to the substance and impaired fertility.

Substances known to cause developmental toxicity in humans

There is sufficient evidence to establish a causal relationship between human exposure to the substance and subsequent developmental toxic effects in the progeny.

Category 2

Substances which should be regarded as if they impair fertility in humans

There is sufficient evidence to provide a strong presumption that human exposure to the substance may result in impaired fertility on the basis of:

 clear evidence in animal studies of impaired fertility in the absence of toxic effects, or, evidence of impaired fertility occurring at around the same dose levels as other toxic effects

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

but which is not a secondary nonspecific consequence of the other toxic effects,

- other relevant information.

Substances which should be regarded as if they cause developmental toxicity to humans

There is sufficient evidence to provide a strong presumption that human exposure to the substance may result in developmental toxicity, generally on the basis of:

- clear results in appropriate animal studies where effects have been observed in the absence of signs of marked maternal toxicity, or at around the same dose levels as other toxic effects but which are not a secondary non-specific consequence of the other toxic effects,
- other relevant information.

Category 3

Substances which cause concern for human fertility Generally on the basis of:

- results in appropriate animal studies which provide sufficient evidence to cause a strong suspicion of impaired fertility in the absence of toxic effects, or evidence of impaired fertility occuring at around the same dose levels as other toxic effects, but which is not a secondary non-specific consequence of the other toxic effects, but where the evidence is insufficient to place the substance in Category 2,
- other relevant information.

Substances which cause concern for humans owing to possible developmental toxic effects

Generally on the basis of:

- results in appropriate animal studies which provide sufficient evidence to cause a strong suspicion of developmental toxicity in the absence of signs of marked maternal toxicity, or at around the same dose levels as other toxic effects but which are

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

not a secondary non-specific consequence of the other toxic effects, but where the evidence is insufficient to place the substance in Category 2,

- other relevant information.

4.2.3.2. *The following symbols and specific risk phrases apply:*

Category 1:

for substances that impair fertility in humans:

Substances classified as toxic to reproduction category 1 shall be assigned the symbol 'T' and the risk phrase

R60 May impair fertility

for substances that cause developmental toxicity:

Substances classified as toxic to reproduction category 1 shall be assigned the symbol 'T' and the risk phrase

R61 May cause harm to the unborn child

Category 2:

for substances that should be regarded as if they impair fertility in humans:

Substances classified as toxic to reproduction category 2 shall be assigned the symbol 'T' and the risk phrase

R60 May impair fertility

for substances that should be regarded as if they cause developmental toxicity in humans:

Substances classified as toxic to reproduction category 2 shall be assigned the symbol 'T' and the risk phrase

R61 May cause harm to the unborn child.

Category 3:

for substances which cause concern for human fertility:

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Substances classified as toxic to reproduction category 3 shall be assigned the symbol 'Xn' and the risk phrase

R62 Possible risk of impaired fertility

for substances which cause concern for humans owing to possible developmental toxic effects:

Substances classified as toxic to reproduction category 3 shall be assigned the symbol 'Xn' and the risk phrase

R63 Possible risk of harm to the unborn child.

4.2.3.3. Comments regarding the categorisation of substances toxic to reproduction

Reproductive toxicity includes impairment of male and female reproductive functions or capacity and the induction of non-inheritable harmful effects on the progeny. This may be classified under two main headings of 1. Effects on male or female fertility; 2. Developmental toxicity.

- Effects on male or female fertility, includes adverse effects on libido, sexual behaviour, any aspect of spermatogenesis or oogenesis, or on hormonal activity or physiological response which would interfere with the capacity to fertilise, fertilisation itself or the development of the fertilised ovum up to and including implantation.
- Developmental toxicity, is taken in its widest sense to include any effect interfering with normal development, both before and after birth. It includes effects induced or manifested prenatally as well as those manifested postnatally. This includes embrytoxic/fetotoxic effects such as reduced body weight, growth and developmental retardation, organ toxicity, death, abortion, structural defects (teratogenic effects), functional defects, peri-postnatal defects, and impaired postnatal mental or physical development up to and including normal pubertal development.

Classification of chemicals as toxic to reproduction is intended to be used for chemicals which have an intrinsic or specific property to produce such toxic effects. Chemicals should not be classified as toxic to reproduction where such effects are solely produced as a non-specific secondary

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

consequence of other toxic effects. Chemicals of most concern are those which are toxic to reproduction at exposure levels which do not produce other signs of toxicity.

The placing of a compound in Category 1 for effects on fertility and/or developmental toxicity is done on the basis of epidemiological data. Placing into Categories 2 or 3 is done primarily on the basis of animal data. Data from in vitro studies, or studies on avian eggs, are regarded as 'supportive evidence' and would only exceptionally lead to classification in the absence of in vivo data.

In common with most other types of toxic effect, substances demonstrating reproductive toxicity will be expected to have a threshold below which adverse effects would not be demonstrated. Even when clear effects have been demonstrated in animal studies the relevance for humans may be doubtful because of the doses administered, for example, where effects have been demonstrated only at high doses, or where marked toxicokinetic differences exist, or the route of administration is inappropriate. For these or similar reasons it may be that classification in Category 3, or even no classification, will be warranted.

Annex V to the Directive specifies a limit test in the case of substances of low toxicity. If a dose level of at least 1000 mg/kg orally produces no evidence of effects toxic to reproduction, studies at other dose levels may not be considered necessary. If data are available from studies carried out with doses higher than the above limit dose, this data must be evaluated together with other relevant data. Under normal circumstances it is considered that effects seen only at doses in excess of the limit dose would not necessarily lead to classification as 'Toxic to reproduction'.

EFFECTS ON FERTILITY

For the classification of a substance into Category 2 for impaired fertility, there should normally be clear evidence in one animal species, with supporting evidence on mechanism of action or site of action, or chemical relationship to other known anti-fertility agents or other information from humans which would lead to the conclusion that effects would be likely to be seen in humans. Where there are studies in only one species without other relevant supporting evidence then classification in Category 3 may be appropriate.

Since impaired fertility may occur as a non-specific accompaniment to severe generalised toxicity or where there is severe inanition, classification into Category 2 should only be made where there is evidence that there is

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

some degree of specificity of toxicity for the reproductive system. If it was demonstrated that impaired fertility in animal studies was due to failure to mate, then for classification into Category 2, it would normally be necessary to have evidence on the mechanism of action in order to interpret whether any adverse effect such as alteration in pattern of hormonal release would be likely to occur in humans.

DEVELOPMENTAL TOXICITY

For classification into Category 2 there should be clear evidence of adverse effects in well conducted studies in one or more species. Since adverse effects in pregnancy or postnatally may result as a secondary consequence of maternal toxicity, reduced food or water intake, maternal stress, lack of maternal care, specific dietary deficiencies, poor animal husbandry, intercurrent infections, and so on, it is important that the effects observed should occur in well conducted studies and at dose levels which are not associated with marked maternal toxicity. The route of exposure is also important. In particular, the injection of irritant material intraperitoneally may result in local damage to the uterus and its contents, and the results of such studies must be interpreted with caution and on their own would not normally lead to classification.

Classification into Category 3 is based on similar criteria as for Category 2 but may be used where the experimental design has deficiencies which make the conclusions less convincing, or where the possibility that the effects may have been due to non-specific influences such as generalised toxicity cannot be excluded.

In general, classification in Category 3 or no category would be assigned on an ad hoc basis where the only effects recorded are small changes in the incidences of spontaneous defects, small changes in the proportions of common variants such as are observed in skeletal examinations, or small differences in postnatal developmental assessments.

Effects during lactation

Substances which are classified as toxic to reproduction and which also cause concern due to their effects on lactation should in addition be labelled with R64 (see criteria in Section 3.2.8.).

For the purpose of classification, toxic effects on offspring resulting only from exposure via the breast milk, or toxic effects resulting from direct exposure of children will not be regarded as 'Toxic to reproduction', unless such effects result in impaired development of the offspring.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Substances which are not classified as toxic to reproduction but which cause concern due to toxicity when transferred to the baby during the period of lactation should be labelled with R64 (see criteria in Section 3.2.8.). This R-phrase may also be appropriate for substances which affect the quantity or quality of the milk.

R64 would normally be assigned on the basis of :

- (a) toxicokinetic studies that would indicate the likelihood that the substance would be present in potentially toxic levels in breast milk; and/or
- (b on the basis of results of one or two generation studies in animals which indicate the presence of adverse effects on the offspring due to transfer in the milk; and/or
- (c) on the basis of evidence in humans indicating a risk to babies during the lactational period.

Substances which are known to accumulate in the body and which subsequently may be released into milk during lactation may be labelled with R33 and R64.

4.2.4. Procedure for the classification of preparations concerning specific effects on health

If a preparation contains one or more substances classified with respect to the criteria laid out above, it must be classified according to the criteria referred to in Annex II, Part A. 7. - 9. and Part B. 6. of Directive 1999/45/EC (the concentration limits are either in Annex I of this Directive, or in Annex II, Part B. 6. of Directive 1999/45/EC where the substance or substances under consideration do not appear in Annex I or appear in it without concentration limits).

5. CLASSIFICATION ON THE BASIS OF ENVIRONMENTAL EFFECTS

5.1. Introduction

The primary objective of classifying substances and preparations dangerous for the environment is to alert the user to the hazards these substances and preparations present to ecosystems. Although the present criteria refer to

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

aquatic ecosystems it is recognised that certain substances and preparations may simultaneously or alternatively affect other ecosystems whose constituents may range from soil microflora and microfauna to primates.

The criteria set out below follow directly from the test methods set out in Annex V in so far as they are mentioned. The test methods required for the 'base set' referred to in Annex VII are limited and the information derived from them may be insufficient for an appropriate classification. Classification may require additional data derived from Level 1 (Annex VIII) or other equivalent studies. Furthermore, classified substances may be subject to review in the light of other new data.

For the purposes of classification and labelling and having regard to the current state of knowledge such substances and preparations are divided into two groups according to their acute and/or long-term effects in aquatic systems or their acute and/or long-term effects in non-aquatic systems.

- 5.1.1. The classification of substances is usually made on the basis of experimental data for acute aquatic toxicity, degradation, and log Pow (or BCF if available).
- 5.1.2. The classification of preparations shall normally be carried out on the basis of a conventional method referred to in Article 7 and Annex III, Parts A and B of Directive 1999/45/EC. In this case, the classification is based on the individual concentration limits
 - in Annex I to this Directive
 - or in Annex III Part B to Directive 1999/45/EC where the substance or substances do not appear in Annex I to this Directive or appear in it without concentration limits.
- 5.1.3. Normally, the classification of a preparation is made on the basis of a conventional method. However, for the determination of the acute aquatic toxicity, there may be cases for which it is appropriate to carry out tests on the preparation. The result of these tests on the preparation may only modify the classification concerning acute aquatic toxicity which would have been obtained by the application of a conventional method. If such tests are chosen by the person responsible for the placing on the market, it must be ensured that the quality criteria of the test methods in Part C of Annex V to this Directive have been complied with. Furthermore, the tests are to be carried out on all three groups of species in conformity with the criteria in this Annex (algae, daphnia and fish), unless the highest hazard classification relating to acute aquatic toxicity has been assigned to the preparation after

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

testing on one of the species or a test result was already available before Directive 1999/45/EC entered into force.

5.2. Criteria for classification, indication of danger, choice of risk phrases

The classification criteria for substances in Section 5.2.1. only apply to preparations where they have been tested in accordance with 5.1.3.

- 5.2.1. Aquatic environment
- 5.2.1.1. Substances shall be classified as dangerous for the environment and assigned the symbol 'N' and the appropriate indication of danger, and assigned risk phrases in accordance with the following criteria:
- R50 Very toxic to aquatic organisms

and

R53 May cause long-term adverse effects in the aquatic environment

Acute toxicity: 96 hr LC₅₀ (for fish) ≤ 1 mg/l

or 48 hr EC₅₀ (for Daphnia) $\leq 1 \text{ mg/l}$

or 72 hr IC₅₀ (for algae) ≤ 1 mg/l

and

- the substance is not readily degradable

or

the log Pow (log octanol/water partition coefficient) ≥ 3.0 (unless the experimentally determined BCF ≤ 100).

R50 Very toxic to aquatic organisms

Acute toxicity: $96 \text{ hr LC}_{50} \text{ (for fish)} < 1 \text{ mg/l}$

or $48 \text{ hr EC}_{50}(\text{for Daphnia}) \leq 1 \text{ mg/l}$

or $72 \text{ hr IC}_{50}(\text{for algae}) < 1 \text{ mg/l}$

R51 Toxic to aquatic organisms and

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

R53 May cause long-term adverse effects in the aquatic environment

Acute toxicity: 96 hr LC₅₀ (for fish) $1 \text{ mg/l} < \text{LC}_{50} < 10 \text{mg/l}$

or 48 hr EC₅₀ (for Daphnia) $1 \text{ mg/l} \le \text{EC}_{50} \le 10 \text{mg/l}$

or 72 hr IC₅₀ (for algae) 1 mg/l \leq IC₅₀ \leq 10 mg/l

and

- the substance is not readily degradable or
- the log Pow \geq 3.0 (unless the experimentally determined BCF \leq 100)
- 5.2.1.2. Substances shall be classified as dangerous for the environment in accordance with the criteria set out below. Risk phrases shall also be assigned in accordance with the following criteria

R52 Harmful to aquatic organisms

R53 May cause long-term adverse effects in the aquatic environment

Acute toxicity: 96 hr LC₅₀ (for fish) $10 \text{ mg/l} < \text{LC}_{50} < 100 \text{mg/l}$

or 48 hr EC₅₀ (for Daphnia) $10 \text{ mg/l} < \text{EC}_{50} \le 100 \text{mg/l}$

or 72 hr IC₅₀ (for algae) $10 \text{ mg/l} \le \text{IC}_{50} \le 100 \text{ mg/l}$

and

the substance is not readily degradable.

This criterion applies unless there exists additional scientific evidence concerning degradation and/or toxicity sufficient to provide an adequate assurance that neither the substance nor its degradation products will constitute a potential long-term and/or delayed danger to the aquatic environment. Such additional scientific evidence should normally be based on the studies required at Level 1 (Annex VIII), or studies of equivalent value, and could include:

(i) a proven potential to degrade rapidly in the aquatic environment,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

(ii) an absence of chronic toxicity effects at a concentration of 1.0 mg/litre, e.g. a no-observed effect concentration of greater than 1.0 mg/litre determined in a prolonged toxicity study with fish or Daphnia.

R52 Harmful to aquatic organisms

Substances not falling under the criteria listed above in this chapter, but which on the basis of the available evidence concerning their toxicity may nevertheless present a danger to the structure and/or functioning of aquatic ecosystems.

R53 May cause long-term adverse effects in the aquatic environment

Substances not falling under the criteria listed above in this chapter, but which, on the basis of the available evidence concerning their persistence, potential to accumulate, and predicted or observed environmental fate and behaviour may nevertheless present a long-term and/or delayed danger to the structure and/or functioning of aquatic ecosystems.

For example, poorly water-soluble substances, i.e. substances with a solubility of less than 1 mg/l will be covered by this criterion if:

- (a) they are not readily degradable; and
- (b) the log Pow \geq 3.0 (unless the experimentally determined BCF \leq 100).

This criterion applies to substances unless there exists additional scientific evidence concerning degradation and/or toxicity sufficient to provide an adequate assurance that neither the substance nor its degradation products will constitute a potential long-term and/or delayed danger to the aquatic environment.

Such additional scientific evidence should normally be based on the studies required at Level 1 (Annex VIII), or studies of equivalent value, and could include

- (i) a proven potential to degrade rapidly in the aquatic environment;
- (ii) an absence of chronic toxicity effects at the solubility limit e.g. a no-observed effect concentration of greater than the solubility

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

limit determined in a prolonged toxicity study with fish or Daphnia.

- 5.2.1.3. Comments on the determination of IC50 for algae and of degradability
 - where it can be demonstrated in the case of highly coloured substances that algal growth is inhibited solely as a result of a reduction in light intensity, then the 72h IC50 for algae should not be used as a basis for classification.
 - Substances are considered readily degradable if the following criteria hold true.
 - (a) If in 28-day biodegradation studies the following levels of degradation are achieved
 - in tests based upon dissolved organic carbon: 70%,
 - in tests based upon oxygen depletion or carbon dioxide generation: 60 % of the theoretical maxima.

These levels of biodegradation must be achieved within 10 days of the start of degradation, which point is taken as the time when 10% of the substance has been degraded.

or

(b) if in those cases where only COD and BOD5 data are available when the ratio of BOD5/COD is greater than or equal to 0.5;

or

- (c) if other convincing scientific evidence is available to demonstrate that the substance can be degraded (biotically and/or abiotically) in the aquatic environment to a level of > 70% within a 28-day period.
- 5.2.2. Non aquatic environment
- 5.2.2.1. Substances and preparations shall be classified as dangerous for the environment and assigned the symbol 'N' and the appropriate indication of danger, and assigned risk phrases in accordance with the following criteria:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- R54 Toxic to flora
- R55 Toxic to fauna
- R56 Toxic to soil organisms
- R57 Toxic to bees
- R58 May cause long-term adverse effects in the environment

Substances and preparations which on the basis of the available evidence concerning their toxicity, persistence, potential to accumulate and predicted or observed environmental fate and behaviour may present a danger, immediate or long-term and/or delayed, to the structure and/or functioning of natural ecosystems other than those covered under 5.2.1 above. Detailed criteria will be elaborated later.

5.2.2.2. Substances and preparations shall be classified as dangerous for the environment, and assigned the symbol 'N' and the appropriate indication of danger, where applicable, and assigned risk phrases in accordance with the following criteria:

R59 Dangerous for the ozone layer

Substances which on the basis of the available evidence concerning their properties and their predicted or observed environmental fate and behaviour may present a danger to the structure and/or the functioning of the stratospheric ozone layer. This includes the substances which are listed in Annex I to Council Regulation (EC) No 2037/2000 on substances that deplete the ozone layer (OJ No L 244, 29.9.2000, p.1) and its subsequent amendments.

Preparations shall be classified on the basis of a conventional method referred to in Article 7 and Annex III, Parts A and B of Directive 1999/45/EC.

6. CHOICE OF SAFETY ADVICE PHRASES

6.1. Introduction

Safety advice phrases (S-phrases) shall be assigned to dangerous substances and preparations in accordance withthe following general criteria. In addition, for certain preparations, the safety advice listed in Annex V of Directive 1999/45/EC is mandatory.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

Whenever the manufacturer is mentioned in Chapter 6 it refers to the person responsible for placing the substance or preparation on the market.

6.2. Safety phrases for substances and preparations

S1 Keep locked up

- Applicability:
 - very toxic, toxic and corrosive substances and preparations.
- Criteria for use:
 - *obligatory* for those substances and preparations mentioned above if sold to the general public.
- S2 Keep out of the reach of children
- Applicability:
 - all dangerous substances and preparations.
 - Criteria for use:
 - *obligatory* for all dangerous substances and preparations sold to the general public, except for those only classified as dangerous for the environment.
- S3 Keep in a cool place
- Applicability:
 - organic peroxides,
 - other dangerous substances and preparations having a boiling point $< 40^{\circ}$ C.
- Criteria for use:
 - obligatory for organic peroxides unless S47 is used,
 - recommended for other dangerous substances and preparations having a boiling point ≤ 40 °C.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- S4 Keep away from living quarters
- Applicability:
 - very toxic and toxic substances and preparations.
 - Criteria for use:
 - normally limited to very toxic and toxic substances and preparations when desirable to supplement S13; for example when there is an inhalation risk and the substance or preparation should be stored away from living quarters. The advice is not intended to preclude proper use of the substance or preparation in living quarters.
- S5 Keep contents under ... (appropriate liquid to be specified by the manufacturer)
- Applicability:
 - spontaneously flammable solid substances and preparations.
- Criteria for use:
 - normally limited to special cases, e.g. sodium, potassium or white phosphorous.
- S6 Keep under ... (inert gas to be specified by the manufacturer)
- Applicability:
 - dangerous substances and preparations which must be kept under an inert atmosphere.
 - Criteria for use:
 - normally limited to special cases, e.g. certain organometallic compounds.
- S7 Keep container tightly closed

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- Applicability:
 - organic peroxides,
 - substances and preparations which can give off very toxic, toxic, harmful or extremely flammable gases,
 - substances and preparations which in contact with moisture give off extremely flammable gases,
 - highly flammable solids.
- Criteria for use:
 - *obligatory* for organic peroxides,
 - recommended for the other fields of application mentioned above.
- S8 *Keep container dry*
- Applicability:
 - substances and preparations which may react violently with water,
 - substances and preparations which on contact with water liberate extremely flammable gases,
 - substances and preparations which on contact with water liberate very toxic or toxic gases.
- Criteria for use:
 - normally limited to the fields of application mentioned above when necessary to reinforce warnings given by R14, R15 in particular, and R29.
- S9 Keep container in a well-ventilated place
- Applicability:
 - volatile substances and preparations which may give off very toxic, toxic or harmful vapours,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- extremely flammable or highly flammable liquids and extremely flammable gases.
- Criteria for use:
 - recommended for volatile substances and preparations which may give off very toxic, toxic or harmful vapours,
 - recommended for extremely flammable or highly flammable liquids or extremely flammable gases.
- S12 Do not keep the container sealed
- Applicability:
 - substances and preparations which will by giving off gases or vapours be liable to burst the container.
- Criteria for use:
 - normally limited to the special cases mentioned above.
- S13 Keep away from food, drink and animal feedingstuffs
- Applicability:
 - very toxic, toxic and harmful substances and preparations.
- Criteria for use:
 - recommended when such substances and preparations are likely to be used by the general public.
- S14 Keep away from ... (incompatible materials to be indicated by the manufacturer)
- Applicability:
 - organic peroxides.
- Criteria for use:

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- *obligatory* for and normally limited to organic peroxides. However, may be useful in exceptional caseswhen incompatibility is likely to product a particular risk
- S15 Keep away from heat
- Applicability:
 - substances and preparations which may decompose or which may react spontaneously under the effect of heat.
- Criteria for use:
 - normally limited to special cases, e.g. monomers, but not assigned if risk phrases R2, R3 and/or R5 have already been applied.
- S16 Keep away from sources of ignition No smoking
- Applicability:
 - extremely flammable or highly flammable liquids and extremely flammable gases.
- Criteria for use:
 - recommended for the substances and preparations mentioned above but not assigned if risk phrases R2, R3 and/or R5 have already been applied.
- S17 *Keep away from combustible material*
- Applicability:
 - substances and preparations which may form explosive or spontaneously flammable mixtures with combustible material.
- Criteria for use:
 - available for use in special cases, e.g. to emphasise R8 and R9.
- S18 *Handle and open container with care*

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- Applicability:
 - substances and preparations liable to produce an overpressure in the container,
 - substances and preparations which may form explosive peroxides.
- Criteria for use:
 - normally limited to the above-mentioned cases when there is risk of damage to the eyes and/or when the substances and preparations are likely to be used by the general public.
- S20 When using do not eat or drink
- Applicability:
 - very toxic, toxic and corrosive substances and preparations.
- Criteria for use:
 - normally limited to special cases (e.g. arsenic and arsenic compounds; fluoracetates) in particular when any of these are likely to be used by the general public.
- S21 When using do not smoke
- Applicability:
 - substances and preparations which produce toxic products on combustion.
- Criteria for use
 - normally limited to special cases (e.g. halogenated compounds).
- S22 Do not breathe dust
- Applicability:

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- all solid substances and preparations dangerous for health.
- Criteria for use:
 - *obligatory* for those substances and preparations mentioned above to which R42 is assigned,
 - recommended for those substances and preparations mentioned above which are supplied in the form of an inhalable dust and for which the health hazards following inhalation are not known.
- S23 Do not breathe gas/fumes/vapour/spray (appropriate wording to be specified by the manufacturer)
- Applicability:
 - all liquid or gaseous substances and preparations dangerous to health.
- Criteria for use:
 - *obligatory* for those substances and preparations mentioned above to which R42 is assigned,
 - *obligatory* for substances and preparations intended for use by spraying. Either S38 or S51 must be ascribed in addition,
 - recommended when it is necessary to draw the attention of the user to inhalation risks not mentioned in the risk phrases which have to be ascribed.
- S24 Avoid contact with skin
- Applicability:
 - all substances and preparations dangerous for health.
- Criteria for use:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- *obligatory* for those substances and preparations to which R43 has been ascribed, unless S36 has also been ascribed.
- recommended when it is necessary to draw the attention of the user to skin contact risks not mentioned in the risk phrases (e.g. paresthesia) which have to be ascribed. However, may be used to emphasise such risk phrases.
- S25 Avoid contact with eyes
- Applicability:
 - all substances and preparations dangerous to health.
- Criteria for use:
 - recommended when it is necessary to draw the attention of the user to eye contact risks not mentioned in the risk phrases which have to be applied. However, may be used to emphasise such risk phrases.
 - recommended for substances ascribed R34, R35, R36 or R41 which are likely to be used by the general public.
- S26 In case of contact with eyes, rinse immediately with plenty of water and seek medical advice
- Applicability:
 - corrosive or irritant substances and preparations.
- Criteria for use:
 - *obligatory* for corrosive substances and preparations and those to which R41 has already been ascribed,
 - recommended for irritant substances and preparations to which the risk phrase R36 has already been ascribed.
- S27 *Take off immediately all contaminated clothing.*
- Applicability:

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- very toxic, toxic or corrosive substances and preparations.

Criteria for use:

- obligatory for very toxic substances and preparations to which R27 has been ascribed and which are likely to be used by the general public.
- recommended for very toxic substances and preparations to which R27 has been ascribed used in industry. However, this safety phrase should not be used if S36 has been ascribed.
- recommended for toxic substances and preparations to which R24 has been ascribed as well as corrosive substances and preparations which are likely to be used by the general public.
- S28 After contact with skin, wash immediately with plenty of ... (to be specified by the manufacturer).
- Applicability:
 - very toxic, toxic or corrosive substances and preparations.
- Criteria for use:
 - *obligatory* for very toxic substances and preparations.
 - recommended for the other substances and preparations mentioned above, in particular when water is not the most appropriate rinsing fluid.
 - recommended for corrosive substances and preparations which are likely to be used by the general public.
- S29 Do not empty into drains
- Applicability:
 - extremely or highly flammable liquids immiscible with water,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- very toxic and toxic substances and preparations,
- substances and preparations dangerous for the environment.

- Criteria for use:

- *obligatory* for substances and preparations dangerous for the environment and assigned the symbol 'N', which are likely to be used by the general public, unless this is the intended use.
- recommended for other substances and preparations mentioned above which are likely to be used by the general public, unless this is the intended use.

S30 Never add water to this product

- Applicability:
 - substances and preparations which react violently with water.

- Criteria for use:

- normally limited to special cases (e.g. sulphuric acid) and may be used, as appropriate, to give the clearest possible information, either to emphasise R14 or as an alternative to R14.
- S33 Take precautionary measures against static discharges
- Applicability:
 - extremely or highly flammable substances and preparations.
- Criteria for use:
 - recommended for substances and preparations used in industry which do not absorb moisture. Virtually never used for substances and preparations as placed on the market for use by the general public.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- S35 This material and its container must be disposed of in a safe way.
- Applicability:
 - all dangerous substances and preparations
- Criteria for use:
 - recommended for substances and preparations where special guidance is needed to ensure proper disposal.
- S36 Wear suitable protective clothing
- Applicability:
 - organic peroxides,
 - very toxic, toxic or harmful substances and preparations,
 - corrosive substances and preparations.
- Criteria for use:
 - *obligatory* for very toxic and corrosive substances and preparations,
 - obligatory for those substances and preparations to which either R21 or R24 has been ascribed,
 - *obligatory* for category 3 carcinogens, mutagens and substances toxic to reproduction unless the effects are produced solely by inhalation of the substance or preparation,
 - obligatory for organic peroxides,
 - recommended for toxic substances and preparations if the LD50 dermal value is unknown but the substance or preparation is likely to be toxic through skin contact,
 - recommended for substances and preparations used in industry which are liable to damage health by prolonged exposure.
- S37 *Wear suitable gloves*

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Applicability

- very toxic, toxic, harmful or corrosive substances and preparations,
- organic peroxides,
- substances and preparations irritating to the skin or causing sensitisation by skin contact.

Criteria for use

- *obligatory* for very toxic and corrosive substances and preparations,
- *obligatory* for those substances and preparations to which either R21, R24 or R43 has been ascribed,
- *obligatory* for Category 3 carcinogens, mutagens and substances toxic to reproduction unless the effects are produced solely by inhalation of the substances and preparations,
- *obligatory* for organic peroxides, recommended for toxic substances and preparations if the LD50 dermal value is unknown but the substance or preparation is likely to be harmful by skin contact,
- recommended for substances and preparations irritating to the skin.
- S38 In case of insufficient ventilation, wear suitable respiratory equipment
- Applicability:
 - very toxic or toxic substances and preparations.

- Criteria for use:

- normally limited to special cases involving the use of very toxic or toxic substances and preparations in industry or in agriculture.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- S39 Wear eye/face protection
- Applicability:
 - organic peroxides,
 - corrosive substances and preparations, including irritants which give rise to risk of serious damage to the eyes.
 - very toxic and toxic substances and preparations.
- Criteria for use:
 - *obligatory* for those substances and preparations to which R34, R35 or R41 have been ascribed.
 - *obligatory* for organic peroxides,
 - recommended when it is necessary to draw the attention of the user to eye contact risks not mentioned in the risk phrases which have to be ascribed,
 - normally limited to exceptional cases for very toxic and toxic substances and preparations, where there is a risk of splashing and they are likely to be easily absorbed by the skin.
- S40 To clean the floor and all objects contaminated by this material use ... (to be specified by the manufacturer)
- Applicability:
 - all dangerous substances and preparations.
- Criteria for use:
 - normally limited to those dangerous substances and preparations for which water is not considered to be a suitable cleansing agent (e.g. where absorption by powdered material, dissolution by solvent etc. Is necessary) and where it is important for health and/or safety reasons to provide a warning on the label.
- S41 In case of fire and/or explosion do not breathe fumes

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- Applicability:
 - dangerous substances and preparations which on combustion give off very toxic or toxic gases.
- Criteria for use:
 - normally limited to special cases.
- S42 During fumigation/spraying wear suitable respiratory equipment (appropriate wording to be specified by the manufacturer)
- Applicability:
 - substances and preparations intended for such use but which may endanger the health and safety of the user unless proper precautions are taken.
- Criteria for use:
 - normally limited to special cases.
- S43 In case of fire use ... (indicate in the space the precise type of fire-fighting equipment. If water increases the risk add: Never use water)
- Applicability:
 - extremely flammable, highly flammable and flammable substances and preparations.
- Criteria for use:
 - *obligatory* for substances and preparations which, in contact with water or damp air, evolve extremely flammable gases,
 - recommended for extremely flammable, highly flammable and flammable substances and preparations, particularly when they are immiscible with water.
- S45 In case of accident or if you feel unwell seek medical advice immediately (show the label where possible).
- Applicability:
 - very toxic substances and preparations,

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- toxic and corrosive substances and preparations,
- substances and preparations causing sensitisation by inhalation.

Criteria for use:

- *obligatory* for the substances and preparations mentioned above.
- S46 If swallowed, seek medical advice immediately and show this container or label
- Applicability
 - all dangerous substances and preparations other than those which are very toxic, toxic, corrosive or dangerous to the environment.
- Criteria for use:
 - obligatory for all dangerous substances and preparations mentioned above which are likely to be used by the general public, unless there is no reason to fear any danger from swallowing, particularly by children.
- S47 Keep at temperature not exceeding ... ° C (to be specified by the manufacturer)
- Applicability:
 - substances and preparations which become unstable at a certain temperature.
- Criteria for use:
 - normally limited to special cases (e.g. certain organic peroxides).
- S48 Keep wetted with (appropriate material to be specified by the manufacturer)
- Applicability:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- substances and preparations which may become very sensitive to sparks, friction or impact if allowed to dry out.
- Criteria for use:
 - normally limited to special cases, e.g. nitrocelluloses.
- S49 Keep only in the original container
- Applicability:
 - substances and preparations sensitive to catalytic decomposition.
- Criteria for use:
 - substances and preparations sensitive to catalytic decomposition e.g. certain organic peroxides.
- S50 Do not mix with ... (to be specified by the manufacturer)
- Applicability:
 - substances and preparations which may react with the specified product to evolve very toxic or toxic gases,
 - organic peroxides.
- Criteria for use
 - recommended for substances and preparations mentioned above which are likely to be used by the general public, when it is a better alternative to R31 or R32,
 - obligatory with certain peroxides which may give violent reaction with accelerators or promoters.
- S51 *Use only in well-ventilated areas*
- Applicability:
 - substances and preparations likely to or intended to produce vapours, dusts, sprays, fumes, mists, etc. which give rise to inhalation risks or to a fire or explosion risk

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- Criteria for use:
 - recommended when use of S38 would not be appropriate. Thus important when such substances and preparations are likely to be used by the general public.
- S52 Not recommended for interior use on large surface areas
- Applicability:
 - volatile, very toxic, toxic and harmful substances and preparations containing them.
- Criteria for use:
 - recommended when damage to health is likely to be caused by prolonged exposure to these substances and preparations by reason of their volatilisation from large treated surfaces in the home or other enclosed places where persons congregate.
- S53 Avoid exposure Obtain special instructions before use
- Applicability:
 - substances and preparations that are carcinogenic, mutagenic and/or toxic to reproduction.
- Criteria for use:
 - obligatory for the above-mentioned substances and preparations to which at least one of the following R-phrases has been assigned: R45, R46, R49, R60 or R61.
- S56 Dispose of this material and its container to hazardous or special waste collection point.
- Applicability:
 - all dangerous substances and preparations.
- Criteria for use:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- recommended for all dangerous substances and preparations likely to be used by the general public for which special disposal is required.
- S57 Use appropriate containment to avoid environmental contamination
- Applicability:
 - substances and preparations which have been assigned the symbol 'N'.
- Criteria for use:
 - normally limited to substances and preparations not likely to be used by the general public.
- S59 Refer to manufacturer for information on recovery/recycling
- Applicability:
 - all dangerous substances and preparations.
- Criteria for use:
 - *obligatory* for substances and preparations dangerous for the ozone layer,
 - recommended for other substances and preparations for which recovery/recycling is recommended.
- S60 This material and its container must be disposed of as hazardous waste
- Applicability:
 - all dangerous substances and preparations.
- Criteria for use:
 - recommended for substances and preparations not likely to be used by the general public and where S35 is not assigned.
- S61 Avoid release to the environment. Refer to special instructions/Safety data sheet

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- Applicability:
 - substances and preparations dangerous for the environment.
- Criteria for use:
 - normally used for substances and preparations which have been assigned the symbol 'N',
 - recommended for all substances and preparations classified dangerous for the environment not covered above.
- S62 If swallowed, do not induce vomiting: seek medical advice immediately and show this container or label.
- Applicability:
 - substances and preparations classified as harmful with R65 in accordance with the criteria in section 3.2.3,
 - not applicable to substances and preparations which are placed on the market in aerosol containers (or in containers fitted with a sealed spray attachment), see sections 8 and 9.
- Criteria for use:
 - obligatory for substances and preparations mentioned above, if sold to, or likely to be used by the general public, except when S45 or S46 are obligatory.
 - recommended for the substances and preparations mentioned above when used in industry, except where S45 or S46 are obligatory.
- S63 In case of accident by inhalation: remove casualty to fresh air and keep at rest.
- Applicability:
 - very toxic and toxic substances and preparations (gases, vapours, particulates, volatile liquids),
 - substances and preparations causing respiratory sensitisation.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- Criteria for use:
 - *obligatory* for substances and preparations to which R26, R23 or R42 has been assigned which are likely to be used by the general public in a way which could result in inhalation.

S64 *If swallowed, rinse mouth with water, (only if the person is conscious).*

- Applicability:
 - corrosive or irritant substances and preparations.
- Criteria for use:
 - recommended for the above substances and preparations which are likely to be used by the general public and where the above treatment is suitable.

7. LABELLING

7.1. When a substance or preparation has been classified the appropriate label is determined with reference to the requirements of Article 23 of this Directive and Article 10 of Directive 1999/45/EC for substances and preparations respectively. This section explains how the label is determined and, in particular, gives guidance on how to choose the appropriate risk and safety phrases.

The label contains the following information:

- (a) for preparations the trade name or designation;
- (b) for substances the name of the substance and for preparations the names of the substances present in the preparations in accordance with the rules set out in Article 10.2.3. of Directive 1999/45/EC;
- (c) the name, full address and telephone number of the person responsible for placing the substance or preparation on the market, whether manufacturer, importer or distributor;
- (d) the symbol(s) and indication(s) of danger;
- (e) phrases indicating particular hazards (R-phrases);

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- (f) phrases indicating safety advice (S-phrases);
- (g) for substances, the EC number, and in addition for substances appearing in Annex I, the word 'EC label';
- (h) for preparations offered or sold to the general public the nominal quantity of the contents unless specified elsewhere on the package.

Note

For certain preparations there are additional labelling requirements set out in Article 10.1.2. and Annex V of Directive 1999/45/EC and in Article 20 of Directive 98/8/EC.

7.1.1. Final choice of risk and safety phrases

Although the final choice of the most appropriate risk and safety phrases is primarily governed by the need to give all necessary information, consideration should also be given in the clarity and impact of the label. With clarity in mind, the necessary information should be expressed in a minimum number of phrases.

In the case of substances which are irritant, highly flammable, flammable and oxidising, an indication of Rphrases and S-phrases need not be given where the package does not contain more than 125ml. This shall also apply in the case of the same volume of harmful substances not retailed to the general public.

For preparations, if the contents of the package do not exceed 125 ml:

- if classified as highly flammable, oxidising, irritant, with the exception of those assigned R41, or dangerous for the environment and assigned the 'N' symbol it shall not be necessary to indicate the R-phrases or the Sphrases,
- if classified as flammable or dangerous to the environment and not assigned the 'N' symbol it shall be necessary to indicate the R-phrases but it shall not be necessary to indicate the S-phrases.
- 7.1.2. Without prejudice to Article 16.4. of Directive 91/414/EEC and to Directive 98/8/EC, indications such as 'nontoxic', 'non-harmfuli' 'non-polluting', 'ecological' or any other statement indicating that the substance

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

or preparation is not dangerous or likely to lead to underestimation of the dangers of the substance or preparation in question shall not appear on the label or packaging of substances or preparations subject to this Directive or to Directive 1999/45/EC.

7.2. Chemical name(s) to be displayed on the label

7.2.1. For substances listed in Annex I the label shall show the name of the substances under one of the designations given in Annex I.

For substances not listed in Annex I, the name is established according to an internationally recognised chemical nomenclature as defined in Section 1.4 above.

7.2.2. For preparations, the choice of names to be displayed on the label follows the rules of Article 10.2.3. of Directive 1999/45/EC.

Note:

Subject to Annex V, B. 9. of Directive 1999/45/EC,

- the name of the sensitising substance must be chosen in accordance with Section 7.2.1 of this Annex,
- in the case of concentrate preparations which are intended for the perfume industry:
 - the person responsible for placing them on the market may identify merely the one sensitising substance judged by him to be primarily responsible for the sensitisation hazard.
 - in the case of a natural substance, the chemical name may be of the type: 'essential oil of ...' 'extract of ...', rather than the name of the constituents of that essential oil or extract.

7.3. Choice of danger symbols

The design of the danger symbols and the wording of the indications of danger shall comply with those laid down in Annex II. The symbol shall be printed in black on an orange-yellow background.

7.3.1. For substances appearing in Annex I the danger symbols and indications of danger shall be those shown in the Annex.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

7.3.2. For dangerous substances not yet appearing in Annex I and for preparations, the danger symbols and indications of danger shall be assigned according to the rules laid down in this Annex.

Where more than one danger symbol is assigned to a substance or preparation:

- the obligation to indicate the symbol 'E' makes the symbols 'F+', 'F' and 'O' optional,
- the obligation to indicate the symbol 'T+' or 'T' makes the symbols 'Xn', 'Xi' and 'C' optional,
- the obligation to indicate the symbol 'C' makes the symbols 'Xn' and 'Xi' optional,
- if the symbol 'Xn' is assigned, the symbol 'Xi' is optional.

7.4. Choice of Risk-phrases

The wording of the R-phrases shall comply with that laid down in Annex III.

The combined R-phrases in Annex III shall be used where applicable.

- 7.4.1. For substances appearing in Annex I, the R-phrases shall be those shown in the Annex.
- 7.4.2. For substances not appearing in Annex I, R-phrases will be selected according to the following criteria and priorities:
 - (a) in the case of dangers which give rise to health effects:
 - (i) R-phrases corresponding to the category of danger illustrated by a symbol must appear on the label;
 - (ii) R-phrases corresponding to other categories of danger which are not illustrated by a symbol by virtue of Article 23
 - (b) in the case of dangers arising from physicochemical properties:
 - R-phrases corresponding to the category of danger illustrated by a symbol must appear on the label;

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (c) in the case of dangers for the environment
 - the R-phrase(s) corresponding to the classification category 'dangerous for the environment' must appear on the label.
- 7.4.3. For preparations, R-phrases will be selected according to the following criteria and priorities:
 - (a) in the case of dangers which give rise to health effects:
 - (i) R-phrases which correspond to the category of danger illustrated by a symbol. In certain cases the Rphrases must be adopted according to the tables of Annex II, Part B of Directive 1999/45/EC. More specifically, the R-phrases of the constituent(s) which are responsible for the assignment of the preparation to a danger category must appear on the label.
 - (ii) R-phrases which correspond to other categories of danger which have been attributed to the constituents but which are not illustrated by a symbol by virtue of Article 10.2.4. of Directive 1999/45/EC;
 - (b) in the case of dangers arising from physicochemical properties:
 - the criteria described under 7.4.3 (a) are applicable, except that the risk phrases 'extremely flammable' or 'highly flammable' need not be indicated where they repeat the wording of the indication of danger used with a symbol.
 - (c) in the case of dangers for the environment
 - (i) the R-phrase(s) corresponding to the classification category ëdangerous for the environmentí must appear on the label;
 - (ii) where the R-phrase R50 has been assigned in addition to a combined R-phrase R51/53 or R52/53 or to the R-phrase 53 alone, the combined R-phrase R50/53 shall be used.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

As a general rule, for preparations a maximum of six R-phrases shall suffice to describe the risk; for this purpose the combined phrases listed in Annex III shall be regarded as single phrases. However, if the preparation falls within more than one danger category, those standard phrases shall cover all the principal hazards associated with the preparation. In some cases, more than six R-phrases may be necessary.

7.5. Safety phrases

The wording of S-phrases shall comply with that laid down in Annex IV.

The combined S-phrases in Annex IV shall be used where applicable.

7.5.1. For substances appearing in Annex I, the S-phrases shall be those shown in the Annex. Where no S-phrases are shown, the manufacturer/importer may include any appropriate S-phrase(s). For substances not in Annex I and for preparations, the manufacturer shall include S-phrases in accordance with the criteria given in Chapter 6 of this Annex.

7.5.2. Choice of safety phrases

The final choice of safety phrases must have regard to the risk phrases indicated on the label and to the intended use of the substance or preparation:

- as a general rule, a maximum of six S-phrases shall suffice to formulate the most appropriate safety advice; for this purpose the combined phrases listed in Annex IV shall be regarded as single phrases,
- in the case of S-phrases concerning disposal, one S-phrase shall be used, unless it is clear that disposal of the material and its container does not present a danger for human health or the environment. In particular, advice on safe disposal is important for substances and preparations sold to the general public,
- some R-phrases become superfluous if a careful selection is made of S-phrases and vice versa; S-phrases which obviously correspond to R phrases will appear on the label only if it is intended to emphasise a specific warning,
- particular attention must be given, in the choice of safety phrases, to the foreseen conditions of use of certain substances

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

and preparations, e.g. spraying or other aerosol effects. Phrases should be chosen with the intended use in view,

- the safety phrases S1, S2 and S45 are obligatory for all very toxic, toxic and corrosive substances and preparations sold to the general public,
- the safety phrases S2 and S46 are obligatory for all other dangerous substances and preparations (except those only classified as dangerous for the environment) sold to the general public.

Where the phrases selected according to the strict criteria in 6.2 result in redundancy or ambiguity or are clearly unnecessary given the specific product/package then some phrases may be deleted.

7.6. The EC number

If a substance named on the label is listed in the European Inventory of Existing Commercial Chemical Substances (Einecs) or in the European List of Notified Substances (Elincs), the Einecs or Elincs number of the substances shall be shown on the Label. This requirement does not apply to preparations.

7.7. Dimensions of the label for preparations

The dimensions of the label shall be as follows:

Capacity of the package Dimensions (in millimetres

- not exceeding 3 litres: if possible, at least 52 x 74

- greater than 3 litres

but not exceeding 50 litres: at least 74 x 105

- greater than 50 litres

but not exceeding 500 litres: at least 105 x 148

- greater than 500 litres: at least 148 x 210

Each symbol shall cover at least one-tenth of the surface area of the label but shall not be less than 1cm2. The label shall be firmly affixed to one or more surfaces of the packaging immediately containing the preparation.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

The information required on the label shall stand out clearly from its background and shall be of such size and spacing as to be easily read.

8. SPECIAL CASES: SUBSTANCES

8.1. Mobile gas cylinders

For mobile gas cylinders the requirements concerning labelling are considered to be satisfied when they are in agreement with Article 23 or Article 24 (6) b.

However, by way of derogation from Article 24 (1) and (2), one of the following alternatives can be used for gas cylinders with a water capacity of less than or equal to 150 litres:

- the format and dimensions of the label can follow the prescriptions of the ISO Standard ISO/DP 7225 (1994 edition) relating to 'Gas cylinders Precautionary labels',
- the information specified in Article 23 (2) may be provided on a durable information disc or label held captive on the cylinder.

8.2. Gas containers intended for propane, butane or liquefied petroleum gas (LPG)

These substances are classified in Annex I. Although classified in accordance with Article 2, they do not present a danger to human health when they are placed on the market in closed refillable cylinders or in non-refillable cartridges within the scope of EN 417 as fuel gases which are only released for combustion (EN 417, September 1992 edition, relating to 'Non-refillable metallic gas cartridges for liquefied petroleum gases, with or without a valve, for use with portable appliances; construction, inspection, testing and marking').

These cylinders or cartridges must be labelled with the appropriate symbol and the R- and S-phrases concerning flammability. No information concerning the effects on human health is required on the label. However, the information concerning effects on human health which should have appeared on the label shall be transmitted to the professional user by the person responsible for placing the substance on the market in the format foreseen in Article 27 of the Directive. For the consumer, sufficient information shall be transmitted to enable them to take all necessary measures for health and safety as foreseen in Article 1 (3) of Directive 91/155/EEC, as modified by Directive 93/112/EEC.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

8.3. Metals in massive form

These substances are classified in Annex I or shall be classified in accordance with Article 6. However, some of these substances, although classified in accordance with Article 2 do not present a danger to human health by inhalation, ingestion or contact with skin or to the aquatic environment in the form in which they are placed on the market. Such substances do not require a label according to Article 23. However, all the information which should have appeared on the label shall be transmitted to the user by the person responsible for placing the metal on the market, in a format foreseen in Article 27.

8.4. Substances classified with R65

Substances classified as harmful on the basis of an aspiration hazard need not be labelled as harmful with R65 when placed on the market in aerosol containers or in containers fitted with a sealed spray attachment.

SPECIAL CASES: PREPARATIONS

9.1. Gaseous preparations (gas mixtures)

For gaseous preparations, consideration must be given to:

- the evaluation of the physicochemical properties,
- the evaluation of health hazards,
- the evaluation of the environmental hazards.

9.1.1. Evaluation of physicochemical properties

9.1.1.1. *Flammability*

The flammable properties of these preparations are determined in accordance with Article 5 of Directive 1999/45/EC according to the methods specified in Part A of Annex V to this Directive.

These preparations will be classified according to the results of the tests carried out and with respect to the criteria of Annex V and to the criteria of the labelling guide.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

However, by derogation, in the case where gaseous preparations are produced to order in small amounts, the flammability of these gaseous mixtures can be evaluated by the following calculation method:

the expression of the gaseous mixture

$$A_1F_1 + \ldots + A_iF_i + \ldots A_nF_n + B_1I_1 + \ldots + B_iI_i + \ldots B_pI_p$$

where: A_i and B_i are the molar fractions

Fi flammable gas

Ii inert gas

n number of flammable gases

p number of inert gases

can be transformed in a form where all the II (inert gases) are expressed by a nitrogen equivalent using a coefficient K_I and where the equivalent content of inflammable gas A'_I is expressed as follows:

$$A'_i = A_i \times (100 / (A_i + K_i B_i))$$

By using the value of the maximum content of flammmable gas which, in a mixture with nitrogen, gives a composition which is not flammmable in air (Tci), the following expression can be obtained:

$$\sum_{i} A'_{i} / Tci \leq 1$$

The gas mixture is flammable if the value of the above expression is greater than one. The preparation is classified extremely flammable and, the phrase R12 is assigned.

Coefficients of equivalency (Ki)

The values of the coefficients of equivalency Ki, between the inert gases and nitrogen and the values of the maximum contents of flammable gas (Tci) may be found in tables 1 and 2 of the ISO Standard ISO 10156 edition 15. 12. 1990 (new: 1996 edition) relating to 'Gases and gas mixtures - Determination of fire potential and oxidising ability for the selection of cylinder valve outlets'.

Maximum content of flammable gas (Tci)

The value of the maximum content of flammable gas (Tci) may be found in table 2 of the ISO Standard ISO 10156 edition 15. 12. 1990 (new: 1996)

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

edition) relating to 'Gases and gas mixtures - Determination of fire potential and oxidising ability for the selection of cylinder valve outlets'.

When a Tci value for a flammable gas does not appear in the above standard, the corresponding lower explosivity limit (LEL) will be used. If no LEL value exists, the value of Tci will be set at 1 % by volume.

Remarks

- The expression above can be used to allow an appropriate labelling of gaseous preparations, however, it should not be regarded as a method for replacing experimentation for the determination of technical safety parameters.
- Furthermore, this expression gives no information as to whether a mixture containing oxidising gases can be prepared safely. When estimating flammability these oxidising gases are not taken into account.
- The expression above will give reliable results only if the flammable gases do not influence each other as far as their flammability is concerned. This has to be considered, e.g. with halogenated hydrocarbons.

9.1.1.2. Oxidising properties

Given the fact that Annex V to this Directive does not contain a method to determine the oxidising properties of gaseous mixtures, the evaluation of these properties must be realised according to the following estimation method.

The principle of the method is comparison of the oxidising potential of gases in a mixture with that of the oxidising potential of oxygen in air. The concentrations of gases in the mixture are expressed in % vol.

It is considered that the gas mixture is as oxidant as or more oxidant than air, if the following condition is verified:

$$\sum_{i} x_i C_i \ge 21$$

Where: x_i is the concentration of gas i in % vol,

Ci is the coefficient of oxygen equivalency.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

In this case, the preparation is classified as oxidising and the phrase R8 will be assigned.

Coefficients of equivalency between gases and oxygen

The coefficients used in the calculation to determine the oxidising capacity of certain gases in a mixture with respect to the oxidising capacity of oxygen in air, listed under 5.2. in the ISO Standard ISO 10156 edition 15. 12. 1990 (new: 1996 edition) relating to 'Gases and gas mixtures - Determination of fire potential and oxidising ability for the selection of cylinder valve outlets', are the following.

 O_2 1 N_2O 0,6

When no value for the Ci coefficient exists for a gas in the cited standard a value of 40 is attributed to this coefficient.

9.1.2. Labelling

For mobile gas containers the requirements concerning labelling are considered to be satisfied when they are in agreement with Article 11. 6. (b) of Directive 1999/45/EC.

However, by way of derogation from Articles 11. 1. and 11. 2., for gas containers with a water capacity of less than or equal to 150 litres, the format and dimensions of the label can follow the prescriptions of the ISO Standard 7225 (1994 edition) relating to 'Gas cylinders - Precautionary labels'. In this case, the label can bear the generic name or industrial/commercial name of the preparation provided that the dangerous component substances of the preparation are shown on the body of the gas cylinder in a clear and indelible way.

The information specified in Article 10 may be provided on a durable information disc or label held captive on the containers.

9.2. Gas containers intended for preparations containing stenched propane, butane or liquefied petroleum gas (LPG)

Propane, butane and liquefied petroleum gas are classified in Annex I. Although preparations containing these substances are classified in accordance with Articles 5, 6 and 7 of Directive 1999/45/EC, they do not present a danger to human health when they are placed on the market in closed refillable cylinders or non-refillable cartridges within the scope on EN 417 as fuel gases which are only released for combustion (EN 417,

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

September 1992 edition, relating to 'Non-refillable metallic gas cartridges for liquefied petroleum gases, with or without a valve, for use with portable appliances; construction, inspection, testing and marking').

These cylinders and cartridges must be labelled with the appropriate symbol and the R- and S-phrases concerning flammability. No information concerning the effects on human health is required on the label. However, the information concerning effects on human health which should have appeared on the label shall be transmitted to the professional user by the person responsible for placing the substance on the market in the format foreseen in Article 14 of Directive 1999/45/EC. For the consumer, sufficient information shall be transmitted to enable them to take all necessary measures for health and safety as foreseen in Article 1 (3) of Directive 91/155/EEC.

9.3. Alloys, preparations containing polymers, preparations containing elastomers

These preparations shall be classified according to the requirements of Articles 5, 6 and 7 and labelled according to the requirements of Article 10 of Directive 1999/45/EC.

However some of these preparations although classified in accordance with Articles 6 and 7 do not present a danger to human health by inhalation, ingestion or contact with the skin or to the aquatic environment in the form in which they are placed on the market. Such preparations do not require a label according to Article 10 or according to Annex V B. 9.. However, all the information which would have appeared on the label shall be transmitted to the professional user by means of an information system in a format foreseen in Article 14 of the above-mentioned Directive.

9.4. Preparations classified with R65

Preparations classified as harmful on the basis of an aspiration hazard need not be labelled as harmful with R65 when placed on the market in aerosol containers or in containers fitted with a sealed spray attachment.

9.5. Organic peroxides

Organic peroxides combine the properties of an oxidiser and a combustible substance in one molecule: when an organic peroxide decomposes, the oxidising part of the molecule reacts exothermically with the combustible (oxidisable) part. For the oxidising properties the existing methods in Annex V cannot be applied to the organic peroxides.

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

The following calculation method based on the presence of active oxygen must be used.

The available oxygen content (%) of an organic peroxide preparation is given by the formula:

$$16 \text{ x} \sum (n_i \text{x } c_i / m_i)$$

where:

n_i= number of peroxygen groups per molecule of organic peroxide i,

 c_i = concentration (mass %) of organic proxide i,

m_i= molecular mass of organic proxide i.

9.6. Additional labelling requirements for certain preparations

For certain preparations there are additional labelling requirements set out in Article 10.1.2. and Annex V of Directive 1999/45/EC and Article 20 of Directive 98/8/EC.

COMMISSION STATEMENT

With regard to Section 4.1.5. and in particular to the last paragraph of Section 4.1.5., the Commission states that, should it envisage making use of the procedure of Article 28, it is prepared to consult in advance appropriate experts designated by Members States and having special qualifications with respect to either carcinogenicity, mutagenicity or reproductive toxicity.

This consultation will take place in the framework of the normal consultation procedure with national experts and/or in the framework of existing committees. The same will be the case when substances already included in Annex I must be reclassified in respect of their carcinogenic, mutagenic effects, or effects toxic to reproduction.

PART 4 ANNEX II TO DIRECTIVE 99/45/EEC

ANNEX II

METHODS FOR THE EVALUATION OF HEALTH HAZARDS OF PREPARATIONS IN ACCORDANCE WITH ARTICLE 6

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Introduction

An assessment must be made for all the health effects corresponding to the health effects of substances contained in a preparation. This conventional method described in Parts A and B of this Annex is a calculation method which is applicable to all preparations and which takes into consideration all the health hazards of substances contained in the preparation. For that purpose the dangerous health effects have been subdivided into:

- 1. acute lethal effects;
- 2. non-lethal irreversible effects after a single exposure;
- 3. severe effects after repeated or prolonged exposure;
- 4. corrosive effects, irritant effects;
- 5. sensitising effects;
- 6. carcinogenic effects, mutagenic effects, toxic effects for reproduction.

The health effects of a preparation are to be assessed in accordance with Article 6(1)(a) by the conventional method described in parts A and B of this Annex using individual concentration limits.

- (a) where the dangerous substances listed in Annex I to Directive 67/548/EEC are assigned concentration limits necessary for the application of the method of assessment described in part A of this Annex, these concentration limits must be used;
- (b) where the dangerous substances do not appear in Annex I to Directive 67/548/EEC or appear there without the concentration limits necessary for the application of the method of evaluation described in part A of this Annex, the concentration limits must be assigned in accordance with the specifications in part B of this Annex.

The procedure for classification is set out in Part A of this Annex.

The classification of the substance(s) and the resulting classification of the preparation are expressed:

- either by a symbol and one or more risk phrases, or

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- by categories (category 1, category 2 or category 3) also assigned risk phrases when substances and preparations are shown to be carcinogenic, mutagenic or toxic for reproduction. Therefore it is important to consider, in addition to the symbol, all the phrases denoting specific risks which are assigned to each substance under consideration.

The systematic assessment of all the dangerous health effects is expressed by means of concentration limits, expressed as a weight/weight percentage except for gaseous preparations where they are expressed as a volume/volume percentage and in conjunction with the classification of the substance.

Where they are not given in Annex I to Directive 67/548/EEC, the concentration limits to be taken into account for the application of this conventional method are those set out in Part B of this Annex.

PART A

Procedure for evaluation of health hazards

The evaluation proceeds stepwise as follows:

- 1. The following preparations are to be classified as very toxic:
- 1.1. owing to their acute lethal effects and assigned the symbol 'T', the indication of danger 'very toxic' and the risk phrases R26, R27 or R28;
- 1.1.1. preparations containing one or more substances classified as very toxic that produce such effects, in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
 - (b) the concentration specified at point 1 in Part B of this Annex (Table I and I A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

1.1.2. preparations containing more than one substance classified as very toxic in lower individual concentrations than the limits specified under 1.1.1(a) or (b) if:

$$\sum \left(\frac{P_{T1}}{L_{T1}}\right) \ge 1$$

where:

 P_{T^+} = is the percentage by weight or by volume of each very toxic substance in the preparation,

 L_{T^+} = is the very toxic limit specified for each very toxic substance, expressed as a percentage by weight or by volume;

1.2. owing to their non-lethal irreversible effects after a single exposure and assigned the symbol 'T⁺', the indication of danger 'very toxic' and the risk phrase R39/route of exposure.

Preparations containing at least one dangerous substance that produces such effects in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 2 in Part B of this Annex (Table II and II A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 2. The following preparations shall be classified as toxic:
- 2.1. owing to their acute lethal effects and assigned the symbol 'T', the indication of danger 'toxic' and the risk phrases R23, R24 or R25;
- 2.1.1. preparations containing one or more substances classified as very toxic or toxic that produce such effects in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

Subsidiary 2003/082

This version is out of date

- (b) the concentration specified at point 1 in Part B of this Annex (Table I and I A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 2.1.2. preparations containing more than one substance classified as very toxic or toxic in lower individual concentrations than the limits specified under 2.1.1(a) or (b) if:

$$\sum \left(\frac{P_{T1}}{L_T} + \frac{P_T}{L_T} \right) \ge 1$$

where:

 P_{T^+} = is the percentage by weight or by volume of each very toxic substance in the preparation,

 P_T = is the percentage by weight or by volume of each toxic substance in the preparation,

 L_T = is the respective toxic limit specified for each very toxic or toxic substance, expressed as a percentage by weight or by volume;

2.2. owing to their non-lethal irreversible effects after a single exposure and assigned the symbol 'T' the indication of danger 'toxic' and the risk phrase R39/route of exposure.

Preparations containing at least one dangerous substance classified as very toxic or toxic that produce such effects in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 2 in Part B of this Annex (Table II and II A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 2.3. owing to their long-term effects and assigned the symbol 'Tl' the indication of danger Îtoxicl and the risk phrase R48/route of exposure.

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Preparations containing at least one dangerous substance that produces such effects in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 3 in Part B of this Annex (Table III and III A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 3. The following preparations shall be classified as harmful:
- 3.1. owing to their acute lethal effects and assigned the symbol ' X_n ' and the indication of danger 'harmful' and the risk phrases R20, R21 or R22;
- 3.1.1. preparations containing one or more substances classified as very toxic, toxic or harmful and that produce such effects in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
 - (b) the concentration specified at point 1 in Part B of this Annex (Table I and I A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 3.1.2. preparations containing more than one substance classified as very toxic, toxic or harmful in lower individual concentrations than the limits specified under 3.1.1(a) or (b) if:

$$\sum \left(\frac{P_{T1}}{L_{xn}} + \frac{P_{T}}{L_{xn}} + \frac{P_{Xn}}{L_{xn}} \right) \ge 1$$

where:

 P_{T^+} = is the percentage by weight or by volume of each very toxic substance in the preparation,

 P_T = is the percentage by weight or by volume of each toxic substance in the preparation,

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

 P_{Xn} = is the percentage by weight or by volume of each harmful substance in the preparation,

 L_{Xn} = is the respective harmful limit specified for each very toxic, toxic or harmful substance, expressed as percentage by weight or by volume;

3.2. owing to their acute effects to the lungs if swallowed and assigned the symbol ' X_n ', and the indication of danger Îharmfull and the risk phrase R65.

Preparations classified as harmful according to the criteria specified in paragraph 3.2.3 of Annex VI to Directive 67/548/EEC. In applying the conventional method according to the above paragraph 3.1 no account shall be taken of the classification of a substance as R65;

3.3. owing to their non-lethal irreversible effects after a single exposure and assigned the symbol 'Xn', the indication of danger Îharmfull and the risk phrase R40/route of exposure.

Preparations containing at least one dangerous substance classified as very toxic, toxic or harmful that produces such effects in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 2 in Part B of this Annex (Table II and II A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 3.4. owing to their long-term effects and assigned the symbol 'Xn', the indication of danger 'harmful' and the risk phrase R48/route of exposure.

Preparations containing at least one dangerous substance classified as toxic or harmful that produces such effects in individual concentrations equal to or greater than:

(a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (b) the concentration specified at point 3 in Part B of this Annex (Table III and III A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 4. The following preparations are to be classified as corrosive
- 4.1. and assigned the symbol 'C', the indication of danger 'corrosive' and the risk phrase R35;
- 4.1.1. preparations containing one or more substances classified as corrosive to which is assigned the phrase R35 in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
 - (b) the concentration specified at point 4 in Part B of this Annex (Table IV and IV A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 4.1.2. preparations containing more than one substance classified as corrosive to which is assigned phrase R35 in lower individual concentrations than the limits specified under 4.1.1(a) or (b) if:

$$\sum \left(\frac{P_{C,R35}}{L_{C,R35}} \right) \ge 1$$

where:

 P_{C} , R_{35} = is the percentage by weight or by volume of each corrosive substance which is assigned phrase R35 in the preparation,

L_C, R₃₅ = is the corrosive limit R35 specified for each corrosive substance to which is assigned phrase R35, expressed as a percentage by weight or by volume;

4.2. and assigned the symbol 'C', the indication of danger 'corrosive' and the risk phrase R34;

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- 4.2.1. preparations containing one or more substances classified as corrosive to which is assigned the phrase R35 or R34 in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
 - (b) the concentration specified at point 4 in Part B of this Annex (Table IV and IV A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 4.2.2. preparations containing more than one of the substances classified as corrosive to which is assigned the phrase R35 or R34 in lower individual concentrations than the limits specified under 4.2.1(a) or (b) if:

$$\sum \left(\frac{P_{C,R35}}{L_{C,R35}} + \frac{P_{C,R34}}{L_{C,R34}} \right) \ge 1$$

where:

 $P_{C, R35}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R35 in the preparation,

 $P_{C, R34}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R34 in the preparation,

 $L_{C, R34}$ = is the respective corrosive limit R34 specified for each corrosive substance to which is assigned phrase R35 or R34, expressed as a percentage by weight or by volume.

- 5. The following preparations are to be classified as irritants:
- 5.1. liable to cause serious eye damage and assigned the symbol 'X_i', the indication of danger 'irritant' and the risk phrase R41;
- 5.1.1. preparations containing one or more substances classified as irritant to which is assigned phrase R41 in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

- (b) the concentration specified at point 4 in Part B of this Annex (Table IV and IV A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 5.1.2. preparations containing more than one of the substances classified as irritant and to which is assigned phrase R41, or classified as corrosive and to which is assigned phrase R35 or R34, in lower individual concentrations than the limits specified under 5.1.1(a) or (b) if:

$$\sum \left(\frac{P_{C,R35}}{L_{Xi,R41}} + \frac{P_{C,R34}}{L_{Xi,R41}} + \frac{P_{Xi,R41}}{L_{Xi,R41}} \right) \ge 1$$

where:

 $P_{C, R35}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R35 in the preparation,

 $P_{C, R34}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R34 in the preparation,

 P_{Xi} , R_{41} = is the percentage by weight or by volume of each irritant substance to which is assigned phrase R41 in the preparation,

 L_{Xi} , $_{R41}$ = is the respective irritant limit R41 specified for each corrosive substance to which is assigned phrase R35 or R34 or irritant substance to which is assigned phrase R41, expressed as percentage by weight or by volume:

- 5.2. irritant to eyes and assigned the symbol 'X_i', the indication of danger 'irritant' and the risk phrase R36;
- 5.2.1. preparations containing one or more substances classified as corrosive to which is assigned phrase R35 or R34 or as irritant and to which is assigned phrase R41 or R36 in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
 - (b) the concentration specified at point 4 in Part B of this Annex (Table IV and IV A) where the substance or substances do not

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;

5.2.2. preparations containing more than one substance classified as irritant to which is assigned phrase R41 or R36, or as corrosive and to which is assigned phrase R35 or R34, in lower individual concentrations than the limits specified under 5.2.1(a) or (b) if:

$$\sum \left(\frac{P_{C,R35}}{L_{Xi,R36}} + \frac{P_{C,R34}}{L_{Xi,R36}} + \frac{P_{Xi,R41}}{L_{Xi,R36}} + \frac{P_{Xi,R36}}{L_{Xi,R36}} \right) \ge 1$$

where:

 $P_{C, R35}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R35 in the preparation,

 $P_{C, R34}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R34 in the preparation,

 $P_{Xi, R41}$ = is the percentage by weight or by volume of each irritant substance to which is assigned phrase R41 in the preparation,

 $P_{Xi, R36}$ = is the percentage by weight or by volume of each irritant substance to which is assigned phrase R36 in the preparation,

L _{Xi, R36} = is the respective irritant limit R36 specified for each corrosive substance to which is assigned phrase R35 or R34 or irritant substance to which is assigned phrase R41, or R36 expressed as percentage by weight or by volume;

- 5.3. irritant to skin and assigned the symbol 'X_i', the indication of danger 'irritant' and the risk phrase R38;
- 5.3.1. preparations containing one or more substances classified as irritant and to which is assigned phrase R38 or as corrosive and to which is assigned phrase R35 or R34 in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
 - (b) the concentration specified at point 4 in Part B of this Annex (Table IV and IV A) where the substance or substances do not

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits:

5.3.2. preparations containing more than one of the substances classified as irritant and to which is assigned phrase R38, or as corrosive and to which is assigned phrase R35 or R34 in lower individual concentrations than the limits specified under 5.3.1(a) or (b) if:

$$\sum \left(\frac{P_{C,R35}}{L_{Xi,R38}} + \frac{P_{C,R34}}{L_{Xi,R38}} + \frac{P_{Xi,R38}}{L_{Xi,R38}} \right) \ge 1$$

where:

 $P_{C, R35}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R35 in the preparation,

 $P_{C, R35}$ = is the percentage by weight or by volume of each corrosive substance to which is assigned phrase R34 in the preparation,

 $P_{Xi, R38}$ = is the percentage by weight or by volume of each irritant substance to which is assigned phrase R38 in the preparation,

L _{Xi, R38} = is the respective irritant limit R38 specified for each corrosive substance to which is assigned phrase R35 or R34 or irritant substance to which is assigned phrase R38, expressed as percentage by weight or by volume;

- 5.4. irritant to respiratory system and assigned the symbol 'X_i', the indication of danger 'irritant' and the risk phrase R37;
- 5.4.1. preparations containing one or more substances classified as irritant and to which is assigned phrase R37 in individual concentrations equal to or greater than:
 - (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
 - (b) the concentration specified at point 4 in Part B of this Annex (Table IV and IV A) where the substance or the substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits:

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

5.4.2. preparations containing more than one substance classified as irritant and to which is assigned phrase R37 in lower individual concentrations than the limits specified under 5.4.1(a) or (b) if:

$$\sum \left(\frac{P_{Xi,R37}}{L_{Xi,R37}}\right) \ge 1$$

where:

P_{Xi, R37} = is the percentage by weight or by volume of each irritant substance to which is assigned phrase R37 in the preparation,

L _{Xi, R37} = is the irritant limit R37 specified for each irritant substance to which is assigned phrase R37, expressed as percentage by weight or by volume;

5.4.3. gaseous preparations containing more than one of the substances classified as irritant to which is assigned phrase R37 or as corrosive and to which is assigned phrase R35 or R34 in lower individual concentrations than the limits specified under 5.4.1(a) or (b) if:

$$\sum \left(\frac{P_{C,R35}}{L_{Xi,R37}} + \frac{P_{C,R34}}{L_{Xi,R37}} + \frac{P_{Xi,R37}}{L_{Xi,R37}} \right) \ge 1$$

where:

 $P_{C, R35}$ = is the percentage by volume of each corrosive substance to which is assigned phrase R35 in the preparation,

 $P_{C, R34}$ = is the percentage by volume of each corrosive substance to which is assigned phrase R34 in the preparation,

 $P_{Xi, R37}$ = is the percentage by volume of each irritant substance to which is assigned phrase R37 in the preparation,

L _{Xi, R37} = is the respective irritant limit R37 specified for each gaseous corrosive substance to which is assigned phrase R35 or R34 or gaseous irritant substance to which is assigned phrase R37, expressed as percentage by weight or by volume.

6. The following preparations are to be classified as sensitising:

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

6.1. by skin contact and assigned the symbol 'X_i', the indication of danger 'irritant' and the risk phrase R43.

Preparations containing at least one substance classified as sensitising and to which is assigned phrase R43 that produces such effects in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 5 in Part B of this Annex (Table V and V A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 6.2. by inhalation and assigned the symbol ' X_n ', the indication of danger 'harmful' and the risk phrase R42.

Preparations containing at least one substance classified as sensitising to which is assigned phrase R42 that produces such effects in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 5 in Part B of this Annex (Table V and V A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 7. The following preparations are to be classified as carcinogenic:
- 7.1. those of category 1 or 2 which are assigned the symbol 'T' and the phrase R45 or R49.

Preparations containing at least one substance producing such effects, classified as carcinogenic and to which is assigned phrase R45 or R49 which denotes carcinogenic substances in category 1 and category 2, in individual concentrations equal to or greater than:

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 7.2. those of category 3 which are assigned the symbol ' X_n ' and the phrase R40.

Preparations containing at least one substance producing such effects classified as carcinogenic and to which is assigned phrase R40 which denotes carcinogenic substances in category 3, in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 8. The following preparations are to be classified as mutagenic:
- 8.1. those of category 1 or 2 which are assigned the symbol 'T' and the phrase R46.

Preparations containing at least one substance producing such effects, classified as mutagenic and to which is assigned phrase R46 which denotes mutagenic substances in category 1 and category 2, in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;

8.2. those of category 3 which are assigned the symbol ' X_n ' and the phrase R40.

Preparations containing at least one substance, producing such effects, classified as mutagenic and to which is assigned phrase R40 which denotes mutagenic substances in category 3, in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.
- 9. The following preparations are to be classified as toxic for reproduction:
- 9.1. those of category 1 or 2 which are assigned the symbol 'T' and the phrase R60 (fertility).

Preparations containing at least one substance producing such effects, classified as toxic for reproduction and to which is assigned phrase R60 which denotes substances toxic for reproduction of category 1 and category 2, in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 9.2. those of category 3 which are assigned the symbol ' X_n ' and the phrase R62 (fertility).

Preparations containing at least one substance producing such effects, classified as toxic for reproduction and to which is assigned phrase R62

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

which denotes substances toxic for reproduction of category 3, in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 9.3. those of category 1 or 2 which are assigned the symbol 'T' and the phrase R61 (development).

Preparations containing at least one substance producing such effects, classified as toxic for reproduction and to which is assigned phrase R61 which denotes substances toxic for reproduction of category 1 and category 2, in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits;
- 9.4. those of category 3 which are assigned the symbol ' X_n ' and the phrase R63 (development).

Preparations containing at least one substance producing such effects, classified as toxic for reproduction and to which is assigned phrase R63 which denotes substances toxic for reproduction of category 3, in individual concentrations equal to or greater than:

- (a) either the concentration specified in Annex I to Directive 67/548/EEC for the substance or substances under consideration, or
- (b) the concentration specified at point 6 in Part B of this Annex (Table VI and VI A) where the substance or substances do not

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

appear in Annex I to Directive 67/548/EEC or appear in it without concentration limits.

PART B

Concentration limits to be used in evaluation of health hazards

For each health effect, the first table (Tables I to VI) sets out the concentration limits (expressed as a weight/weight percentage) to be used for non-gaseous preparations and the second table (Tables I A to VI A) sets out the concentration limits (expressed as a volume/volume percentage) to be used for gaseous preparations. These concentration limits are used in the absence of specific concentration limits for the substance under consideration in Annex I to Directive 67/548/EEC.

1. Acute lethal effects

1.1. Non - gaseous preparations

The concentration limits fixed in Table I, expressed as a weight/weight percentage, determine the classification of the preparation in relation to the individual concentration of the substance(s) present whose classification is also shown.

Table I

| Classification of the substance | Classification of the preparation | | | | |
|---------------------------------|-----------------------------------|----------------------|----------------------|--|--|
| | T^+ | T | X _n | | |
| T ⁺ with R26, R27, | concentration | 1% | 0,1% | | |
| R28 | <u>≥</u> 7% | \leq concentration | \leq concentration | | |
| | | <7% | <1% | | |
| T with R23, R24, | | concentration | 3% | | |
| R25 | | <u>≥</u> 25% | \leq concentration | | |
| | | | <25% | | |
| X _n with R20, R21, | | | concentration | | |
| R22 | | | ≥25% | | |

The R phrases denoting risk are to be assigned to the preparation in accordance with the following criteria:

- the label shall include one or more of the abovementioned R phrases according to the classification used,

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- in general, the R phrases selected should be those applicable to the substance(s) present in the concentration which gives rise to the most severe classification.

1.2. Gaseous preparations

The concentration limits expressed as a volume/volume percentage in Table I A below determine the classification of the gaseous preparations in relation to the individual concentration of the gas(es) present whose classification is also shown.

Table I A

| Classification of the substance (gas) | Classification of the gaseous preparation | | | | |
|---------------------------------------|---|-----------------|-------------------------------|--|--|
| | T^{+} | T | X _n | | |
| T ⁺ with R26, | concentration | 0,2% | 0,02% | | |
| R27, R28 | <u>≥</u> 1% | ≤ concentration | <pre>< concentration</pre> | | |
| | | <1% | <0,2% | | |
| T with R23, | | concentration | 0,5% | | |
| R24, R25 | | ≥5% | ≤ concentration | | |
| | | | <5% | | |
| X_n with R20, | | | concentration | | |
| R21, R22 | | | <u>≥</u> 5% | | |

The R phrases denoting risk shall be assigned to the preparation in accordance with the following criteria:

- the label shall include one or more of the abovementioned R phrases according to the classification used,
- in general, the R phrases selected should be those applicable to the substance(s) present in the concentration which gives rise to the most severe classification.
- 2. Non-lethal irreversible effects after a single exposure

2.1.Non-gaseous preparations

For substances that produce non-lethal irreversible effects after a single exposure (R39/route of exposure, R40/ route of exposure), the individual concentration limits specified in Table II, expressed as a weight/weight

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

percentage, determine, when appropriate, the classification of the preparation.

Table II

| Classification of the substance | Classification of the preparation | | | |
|---------------------------------|-----------------------------------|-----------------|-------------------------------|--|
| | T^+ | T | X _n | |
| T ⁺ with | concentration | 1% | 0,1% | |
| R39/route of | ≥10% R39 (*) | ≤ concentration | ≤ concentration | |
| exposure | obligatory | <10% R39 (*) | <1% R40 (*) | |
| | | obligatory | obligatory | |
| T with | | concentration | 1% | |
| R39/route of | | ≥10% R39 (*) | <pre>< concentration</pre> | |
| exposure | | obligatory | <10% R40 (*) | |
| | | | obligatory | |
| X _n with | | | concentration | |
| R40/route of | | | ≥10% R40 (*) | |
| exposure | | | obligatory | |

^(*) In order to indicate the route of administration/exposure (route of exposure) the combined R phrases listed under points 3.2.1, 3.2.2 and 3.2.3 of the labelling guide (Annex VI to Directive 67/548/EEC) are to be used.

2.2. Gaseous Preparations

For gases that produce non-lethal irreversible effects after a single exposure (R39/route of exposure, R40/route of exposure), the individual concentration limits specified in Table II A, expressed as a volume/volume percentage, determine, when appropriate, the classification of the preparation.

Table II A

| Classification of the substance (gas) | Classification of the gaseous preparation | | | | |
|---------------------------------------|--|-------------|----------------|--|--|
| | T^+ | T | X _n | | |
| T ⁺ with | concentration | 0,2% | 0,02% | | |
| R39/route of | $\geq 1\% R39 (*)$ \leq concentration \leq concentration | | | | |
| exposure | obligatory | <1% R39 (*) | <0,2% R40 (*) | | |
| | | obligatory | obligatory | | |
| T with | concentration $0.5\% \le$ | | | | |
| R39/route of | | ≥5% R39 (*) | concentration | | |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| exposure | obligatory | <5% R40 (*) |
|----------------------------------|--------------------|--------------------------|
| | | obligatory |
| X _n with R40/route of | | concentration $\geq 5\%$ |
| exposure | | R40 (*) obligatory |
| (4) T 1 1 1 1 | (, , ,) , (1 | 1: 10 1 1: 1 |

^(*) In order to indicate the route of administration/exposure (route of exposure) the combined R phrases listed under points 3.2.1, 3.2.2 and 3.2.3 of the labelling guide (Annex VI to Directive 67/548/EEC) are to be used.

3. Severe effects after repeated or prolonged exposure

3.1. Non-gaseous preparations

For substances that produce severe effects after repeated or prolonged exposure (R 48/route of exposure), the individual concentration limits specified in Table III, expressed as a weight/weight percentage, determine, when appropriate, the classification of the preparation

Table III

| Classification of the substance | Classification of the preparation | | |
|---------------------------------|-----------------------------------|---|--|
| | T | X _n | |
| T with R48/route of | concentration ≥10% | $1\% \le \text{concentration} \le 10\%$ | |
| exposure | R48 (*) obligatory | R48 (*) obligatory | |
| X _n with R48/route | | concentration ≥10% | |
| of exposure | | R48 (*) obligatory | |

^(*) In order to indicate the route of administration/exposure (route of exposure) the combined R phrases listed under points 3.2.1, 3.2.2 and 3.2.3 of the labelling guide (Annex VI to Directive 67/548/EEC) are to be used.

3.2. Gaseous preparations

For gases that produce severe effects after repeated or prolonged exposure (R48/route of exposure), the individual concentration limits specified in Table III A below, expressed as a volume/volume percentage, determine, when appropriate, the classification of the preparation.

Table III A

| Classification of the substance (gas) | Classification of the gaseous preparation | |
|---------------------------------------|---|--|
| | T X _n | |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| T with R48/route of | concentration | $0.5\% \le \text{concentration} \le 5\%$ |
|----------------------------------|---------------|--|
| exposure | <u>≥</u> 5% | R48 (*) obligatory |
| | R48 (*) | |
| | obligatory | |
| X _n with R48/route of | | concentration >5% |
| exposure | | R48 (*) obligatory |

^(*) In order to indicate the route of administration/exposure (route of exposure) the combined R phrases listed under points 3.2.1, 3.2.2 and 3.2.3 of the labelling guide (Annex VI to Directive 67/548/EEC) are to be used

- 4. Corrosive and irritant effects including serious damage to the eye
- 4.1. Non gaseous preparations

For substances that produce corrosive effects (R34, R35) or irritant effects (R36, R37, R38, R41), the individual concentration limits specified in Table IV, expressed as a weight/weight percentage, determine, when appropriate, the classification of the preparation.

Table IV

| Classification | Classification of the preparation | | | |
|--------------------------|-----------------------------------|---------------|-------------------------|--------------------------|
| of the | | | | |
| substance | | | | |
| | C with R35 | C with R34 | X _i with R41 | X _i with R36, |
| | | | | R37, R38 |
| C with R35 | concentration | 5% ≤ | 5% (*) | 1% ≤ |
| | ≥10% R35 | concentration | | concentration |
| | obligatory | <10% R34 | | <5% R36/38 |
| | | obligatory | | obligatory |
| C with R34 | | concentration | 10% (*) | 5% <u><</u> |
| | | ≥10% R34 | | concentration |
| | | obligatory | | <10% |
| | | | | R36/38 |
| | | | | obligatory |
| X _i with R41 | | | concentration | 5% ≤ |
| | | | ≥10% R41 | concentration |
| | | | obligatory | <10% R36 |
| | | | | obligatory |
| X _i with R36, | | | | concentration |
| R37, R38 | | | | ≥20% R36, |
| | | | | R37, R38 are |
| | | | | obligatory in |
| | | | | the light of |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| | | the |
|--|--|---------------|
| | | concentration |
| | | present if |
| | | they apply to |
| | | the |
| | | substances |
| | | under |
| | | consideration |

^(*) According to the labelling guide (Annex VI to Directive 67/548/EEC), corrosive substances assigned risk phrases R35 or R34 must also be considered as being assigned phrase R41. Consequently, if the preparation contains corrosive substances with R35 or R34 below the concentration limits for a classification of the preparation as corrosive, such substances can contribute to a classification of the preparation as irritant with R41 or irritant with R36.

4.2. Gaseous preparations

For gases that produce such effects (R34, R35 or R36, R37, R38, R41), the individual concentration limits specified in Table IV A below, expressed as a volume/volume percentage determine, when appropriate, the classification of the preparation.

Table IV A

| Classification of the substance (gas) | Classification of the gaseous preparation | | | | |
|--|---|---|--|---|--|
| | C with R35 | C with R34 | X _i with R41 | X _i with R36, R37, R38 | |
| C with R35 | concentration ≥1% R35 obligatory | 0,2% ≤ concentration <1% R34 obligatory | 0,2% (*) 5% (*) | 0,02% ≤ concentration <0,2% R36/37/38 obligatory 0,5% < | |
| C WIIII R34 | | ≥5% R34 obligatory | 370() | concentration <5% R36/37/38 obligatory | |
| X _i with R41 | | | concentration ≥5% R41 obligatory | 0,5% ≤ concentration <5% R36 obligatory | |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| X _i with R36, | | concentration |
|--------------------------|--|---------------|
| R37, R38 | | ≥5% R36, |
| | | R37, R38 |
| | | obligatory as |
| | | appropriate |

^(*) According to the labelling guide (Annex VI to Directive 67/548/EEC), corrosive substances assigned risk phrases R35 or R34 must also be considered as being assigned phrase R41. Consequently, if the preparation contains corrosive substances with R35 or R34 below the concentration limits for a classification of the preparation as corrosive, such substances can contribute to a classification of the preparation as irritant with R41 or irritant with R36.

- 5. Sensitising effects
- 5.1. Non- gaseous preparations

Preparations that produce such effects are classified as sensitising and assigned:

- the symbol X_n and phrase R42 if this effect can be produced by inhalation,
- the symbol X_i and phrase R43 if this effect can be produced through contact with the skin.

The individual concentration limits specified in Table V, expressed as a weight/weight percentage, determine, when appropriate, the classification of the preparation.

Table V

| Classification of the substance | Classification of the preparation | | |
|---------------------------------|-----------------------------------|-----------------------------------|--|
| | Sensitising with R42 | Sensitising with R43 | |
| Sensitising with R42 | concentration ≥1% R42 obligatory | | |
| Sensitising with R43 | | concentration > 1% R43 obligatory | |

5.2. Gaseous preparations

Gaseous preparations that produce such effects are classified as sensitising and assigned:

- the symbol X_n and phrase R42 if this effect can be produced by inhalation,

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

- the symbol X_i and phrase R43 if this effect can be produced through contact with the skin.

The individual concentration limits specified in Table V A below, expressed as a volume/volume percentage, determine, when appropriate, the classification of the preparation.

Table V A

| Classification of the substance (gas) | Classification of the gaseous preparation | | |
|---------------------------------------|---|------------------|--|
| (8) | Sensitising with R42 | Sensitising with | |
| | K42 | R43 | |
| Sensitising with R42 | concentration | | |
| | ≥0,2% | | |
| | R42 obligatory | | |
| Sensitising with R43 | | concentration | |
| | | ≥0,2% | |
| | | R43 obligatory | |

6. Carcinogenic/mutagenic/toxic effects for reproduction

6.1. Non-gaseous preparations

For substances which produce such effects, the concentration limits laid down in Table VI, expressed as a weight/weight percentage, shall determine, where appropriate, the classification of the preparation. The following symbol and risk phrases are assigned:

| Carcinogenic categories 1 and 2: | T; R45 or R49 |
|--|----------------------|
| Carcinogenic category 3: | X _n ; R40 |
| Mutagenic categories 1 and 2: | T; R46 |
| Mutagenic category 3: | X _n ; R40 |
| Toxic for reproduction fertility categories 1 and 2: | T; R60 |
| Toxic for reproduction development categories 1 and 2: | T; R61 |
| Toxic for reproduction fertility category 3: | X _n ; R62 |
| Toxic for reproduction development category 3: | X _n ; R63 |

Table VI

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

| Classification of the substance | Classification of the preparation | | |
|--|---|--|--|
| | Categories 1 and 2 | | |
| carcinogenic substances of category 1 or 2 with R45 or R49 | concentration ≥ 0,1% carcinogenic R45, R49 obligatory as appropriate | | |
| carcinogenic substances of category 3 with R40 | | concentration ≥ 1% carcinogenic R40 obligatory | |
| mutagenic substances of category 1 or 2 with R46 | concentration ≥ 0,1% mutagenic R46 obligatory | | |
| mutagenic substances of category 3 with R40 | | concentration ≥ 1% mutagenic R40 obligatory | |
| substances 'toxic for reproduction' of category 1 or 2 with R60 (fertility) | concentration ≥ 0,5% toxic for reproduction (fertility) R60 obligatory | | |
| substances'toxic for reproduction' of category 3 with R62 (fertility) | | concentration ≥5% toxic for reproduction (fertility) R62 obligatory | |
| substances 'toxic for reproduction' of category 1 or 2 with R61 (development) | concentration ≥0,5% toxic for reproduction (development) R61 obligatory | | |
| substances 'toxic for reproduction' of category 3 with R63 (Development) | | concentration ≥5% toxic for reproduction (development) R63 obligatory | |

6.2. Gaseous preparations

For gases which produce such effects, the concentration limits laid down in Table VI A, expressed as a volume/ volume percentage, shall determine, where appropriate, the classification of the preparation. The following symbol and risk phrases are assigned:

Carcinogenic categories 1 and 2:

T; R45 or R49

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

| Carcinogenic category 3: | X _n ; R40 |
|--|----------------------|
| Mutagenic categories 1 and 2: | T; R46 |
| Mutagenic category 3: | X _n ; R40 |
| Toxic for reproduction fertility categories 1 and 2: | T; R60 |
| Toxic for reproduction development categories 1 and 2: | T; R61 |
| Toxic for reproduction fertility category 3: | X _n ; R62 |
| Toxic for reproduction development category 3: | X _n ; R63 |

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Table VI A

| Classification of the substance (gas) | Classification of the gaseous preparation | | |
|--|---|--|--|
| | Categories 1 and 2 | Category 3 | |
| carcinogenic substances of category 1 or 2 with R45 or R49 | concentration ≥ 0,1% carcinogenic R45, R49 obligatory as appropriate | | |
| carcinogenic substances of category 3 with R40 | | concentration ≥1% carcinogenic R40 obligatory | |
| mutagenic substances of category 1 or 2 with R46 | concentration ≥ 0,1% mutagenic R46 obligatory | | |
| mutagenic substances of category 3 with R40) | | concentration ≥ 1% mutagenic R40 obligatory | |
| substances 'toxic for reproduction' of category 1 or 2 with R60 (fertility | concentration ≥ 0,2% toxic for reproduction (fertility) R60 obligatory | | |
| substances 'toxic for reproduction' of category 3 with R62 (fertility) | | concentration ≥ 1% toxic for reproduction (fertility) R62 obligatory | |
| substances 'toxic for reproduction' of category 1 or 2 with R61 (development) | concentration ≥ 0,2% toxic for reproduction (development) R61 obligatory | | |
| substances 'toxic for reproduction' of category 3 with R63 (development) | | concentration ≥ 1% toxic for reproduction (development) R63 obligatory | |

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

Subsidiary 2003/082

SCHEDULE 2

regulation 2

List of substances, mixtures and processes which also constitute carcinogens

- 1. Manufacture of auramine.
- 2. Work involving exposure to polycyclic aromatic hydrocarbons present in coal soot, coal tar or coal pitch.
- 3. Work involving exposure to dusts, fumes and sprays produced during the roasting and electro-refining of cupro-nickel mattes.
- 4. Strong acid process in the manufacture of isopropyl alcohol.
- 5. Work involving exposure to hardwood dusts¹

¹ A list of some hardwoods is to be found in Volume 62 of the Monographs on the Evaluation of Carcinogenic Risks to Humans 'Wood Dust and Formaldehyde', published by the International Agency for Research on Cancer, Lyon 1995

Subsidiary 2003/082

FACTORIES (CONTROL OF CARCINOGENS AND MUTAGENS AT WORK) REGULATIONS 2003.

This version is out of date

SCHEDULE 3

regulation 2

Limit values for occupational exposure

| | | | Limit values | | Notation |
|------------------------|------------|----------------|-----------------|----------------|-----------------|
| | | | mg/cubic | | |
| Name of agent | Einecs (2) | CAS <u>(3)</u> | meter (4) | ppm <u>(5)</u> | |
| Benzene | 200-753-7 | 71-43-2 | 3,25 | 1 <u>(6)</u> | Skin <u>(7)</u> |
| Vinyl chloride monomer | 200-831 | 75-01-4 | 7,77 <u>(5)</u> | 3 <u>(5)</u> | |
| Hardwood dusts | | | 5,0 (5) (8) | | |

- (2) ELINCS: European Inventory of Existing Chemical Substances.
- (3) CAS: Chemical Abstract Service Number.
- (4) mg/cubic metre = milligrams per cubic metre of air at 20(C and 101,3 KPa (760 mm mercury pressure).
- (5) ppm = parts per million by volume in air (ml/ cubic metre).
- (6) Measured or calculated in relation to a reference period of eight hours.
- (7) Substantial contribution to the total body burden via dermal exposure possible.
- (8) Inhalable fraction: if hardwood dusts are mixed with other wood dusts, the limit value shall apply to all wood dusts present in that mixture.